UNITED STATES SECURITIES AND EXCHANGE COMMISSION WASHINGTON, D.C. 20549 FORM 10-K

[X] ANNUAL REPORT PURSUANT TO SECTION 13 OR 1 ACT OF 1934 For The For the Fiscal Year E	
[] TRANSITION REPORT PURSUANT TO SECTION 13 EXCHANGE ACT OF 1934 For The Transiton Period	OR 15(d) OF THE SECURITIES d FromTo
	COMMISSION FILE NUMBER 0-19271 IDEXX LABORATORIES, INC. Exact name of registrant as specified in its charter)
Delaware (State or other jurisdiction of incorporation or organization)	01-0393723 (IRS Employer Identification No.)
One IDEXX Drive, Westbrook, Maine (Address of principal executive offices)	04092 (ZIP Code)
(R	(207) 856-0300 Registrant's telephone number, including area code)
SECURITIES R	EGISTERED PURSUANT TO SECTION 12(b) OF THE ACT: NONE
SECURITIES R	EGISTERED PURSUANT TO SECTION 12(g) OF THE ACT: Common Stock, \$0.10 par value per share Preferred StockPurchase Rights (Title of Class)
Indicate by check mark if the registrant is a well-known seaso	oned issuer, as defined in Rule 405 of the Securities Act. Yes [X] No [
Indicate by check mark if the registrant is not required to file	reports pursuant to Section 13 or Section 15(d) of the Act. Yes [] No [X]
	l reports required to be filed by Section 13 or 15(d) of the Securities Exchange Act of 1934 during the preceding red to file such reports), and (2) has been subject to such filing requirements for the past 90 days. Yes [X] No [
	uant to Item 405 of Regulation S-K is not contained herein, and will not be contained, to the best of registrant's rated by reference in Part III of this Form 10-K or any amendment to this Form 10-K. []
Indicate by check mark whether the registrant is a large accel	erated filer, an accelerated filer, or a non-accelerated filer (as defined in Rule 12b-2 of the Exchange Act).
Large accelerated filer [X] \qquad Accelerated filer []	Non-accelerated filer []
Indicate by check mark whether the registrant is a shell company ((as defined in Rule 12b-2 of the Act). Yes [] No [X]
Based on the closing sale price on June 30, 2005, the aggregathe registrant considers its Directors and executive officers to be it	ate market value of the voting stock held by non-affiliates of the registrant was \$2,022,715,832. For these purpose to only affiliates.
The number of shares outstanding of the registrant's Commo	n Stock was 31,852,202 on February 22, 2006.
DOG	CUMENTS INCORPORATED BY REFERENCE
LOCATION IN FORM 10-K Part III	INCORPORATED DOCUMENT Specifically identified portions of the Company's definitive proxy statement to be filed in connection with the Company's Annual Meeting to be held on May 10, 2006 are incorporated herein by reference.

CAUTIONARY STATEMENT REGARDING FORWARD-LOOKING INFORMATION

Forward-looking statements, within the meaning of Section 27A of the Securities Act of 1933, as amended, and Section 21E of the Securities Exchange Act of 1934, as amended, are made throughout this Annual Report on Form 10-K, including statements relating to, among other things, supply commitments, product launches, our competitive position in the industry, future growth rates and gross margins, realization of inventory, product sales, integration of acquisitions and operating expenses. For this purpose, any statements contained herein that are not statements of historical fact may be deemed to be forward-looking statements. Without limiting the foregoing, the words "believes," "anticipates," "plans," "expects," "seeks," "estimates," and similar expressions are intended to identify forward-looking statements. There are a number of important factors that could cause our results to differ materially from those indicated by such forward-looking statements, including those detailed under the caption "Management's Discussion and Analysis of Financial Condition and Results of Operations—Critical Accounting Policies and Estimates" and "Part I, Item 1A. Risk Factors."

In addition, any forward-looking statements represent our views only as of the day this Annual Report on Form 10-K was filed with the Securities and Exchange Commission and should not be relied upon as representing our views as of any subsequent date. While we may elect to update forward-looking statements at some point in the future, we

specifically disclaim any obligation to do so, even if our views change.

PART I.

ITEM 1. BUSINESS

We develop, manufacture and distribute products and provide services for the veterinary and the food and water testing markets. Our products and services include:

- Point-of-care veterinary diagnostic products, comprising rapid assays and instruments and consumables;
- Laboratory and consulting services used by veterinarians;
- Veterinary pharmaceutical products;
- Information products and services and digital radiography systems for veterinarians;
- Diagnostic and health-monitoring products for production animals;
- Products that test water for certain microbiological contaminants; and
- Products that test milk for antibiotic residues.

We are a Delaware corporation and were incorporated in 1983. Our principal executive offices are located at One IDEXX Drive, Westbrook, Maine 04092, our telephone number is 207-856-0300, and our Internet address is idexx.com. References herein to "we," "us;" the "Company," or "IDEXX" include our wholly-owned subsidiaries unless the otherwise requires. References to our website are inactive textual references only and the content of our website should not be deemed incorporated by reference into this Form 10-K for any purpose.

We make available free of charge on our Web site our Annual Report on Form 10-K, Quarterly Reports on Form 10-Q, Current Reports on Form 8-K, and amendments to those reports as soon as reasonably practicable after we file such information with, or furnish it to, the Securities and Exchange Commission ("SEC"). In addition, copies of our reports filed electronically with the SEC may be accessed on the SEC's Web site at www.sec.gov. The public may also read and copy any materials filed with the SEC at the SEC's Public Reference Room at 100 F Street, NE, Washington, DC 20549. Information on the operation of the Public Reference Room may be obtained by calling the SEC at 1-800-SEC-0330.

2

PRODUCTS AND SERVICES

We operate primarily through three business segments: products and services for the veterinary market, which we refer to as our Companion Animal Group ("CAG"), water quality products ("Water") and products for production animal health and dairy quality, which we refer to as the Food Diagnostics Group ("FDG"). See Note 18 to the consolidated financial statements for the year ended December 31, 2005 included in this Form 10-K for financial information about our business segments, including geographic information, and about our product and service categories.

COMPANION ANIMAL GROUP

Instruments and Consumables

We currently market several instrument systems, as well as associated consumable products, for use in veterinary clinics. These instruments, which we refer to collectively as the IDEXX VetLab®, are described below.

Blood and Urine Chemistry. Our VetTest[®] Chemistry Analyzer is used to measure levels of certain enzymes and other substances in blood or urine in order to assist the veterinarian in diagnosing physiologic conditions. Twenty-four separate tests can be performed on the VetTest[®] Chemistry Analyzer and additional parameters can be calculated. Blood tests commonly run include glucose, alkaline phosphatase, ALT (alanine aminotransferase), creatinine, BUN (blood urea nitrogen) and total protein. The VetTest[®] Chemistry Analyzer also runs tests for urine protein/urine creatinine ratio, which assists in the detection of early renal disease. Tests are sold individually and in prepackaged panels, such as the Preanesthetic Panel, the General Health Profile, the Equine Panel, the NSAID ("nonsteroidal anti-inflammatory drug") Monitoring Panel and the Quality Control Panel.

We purchase all of the reagents used in the VetTest[®] Chemistry Analyzer ("dry chemistry slides" or "VetTest[®] slides") from Ortho-Clinical Diagnostics, Inc. ("Ortho"), a subsidiary of Johnson & Johnson. See "Business-Production and Supply." In October 2003, we entered into an agreement with Ortho under which we are developing a next-generation chemistry analyzer for the veterinary market based on Ortho's dry-slide technology, and Ortho will supply the slide consumables used in both the VetTest[®] Chemistry Analyzer and the new analyzer through 2020. We do not expect this new instrument to be commercially available before the latter part of 2007.

Hematology. We sell two hematology analyzers: the LaserCyte[®] Hematology Analyzer, which uses laser-flow cytometry technology to analyze components of blood, including red blood cells, white blood cells, and platelets; and the VetAutoreadTM Hematology Analyzer.

<u>Electrolytes and Blood Gases</u>. Our VetLyte[®] analyzer measures three electrolytes—sodium, potassium and chloride—to aid in evaluating acid-base and electrolyte balances and assessing plasma hydration. We purchase our VetLyte[®] Electrolyte Analyzers and consumables from Roche Diagnostics Corporation.

Our VetStatTM analyzer measures electrolytes, blood gases, acid-base balance, glucose and ionized calcium, and calculates other parameters, such as bicarbonate and anion gap. These measurements aid veterinarians in evaluating fluid therapy choices and measuring respiratory function. The VetStatTM Electrolyte and Blood Gas Analyzer runs single-use disposable cassettes that contain various configurations of analytes. We purchase all of our VetStatTM Electrolyte and Blood Gas Analyzers and consumables from Osmetech, Inc.

Quantitative Immunoassay Testing. The IDEXX SNAP® Reader allows the veterinarian to obtain quantitative measurements of total thyroxine (T₄), cortisol and bile acids. These measurements assist in diagnosing and monitoring the treatment of certain endocrine diseases, such as hyper- and hypothyroidism, Cushing's syndrome and Addison's disease. Samples and reagents are introduced to the analyzer using our SNAP® platform.

We provide a broad range of single-use, handheld test kits that allow quick, accurate and convenient test results for a variety of companion animal diseases and health conditions. These products enable veterinarians to provide improved service to animal owners by delivering test results and a diagnosis at the time of the patient visit, allowing the veterinarian to initiate therapy or prevention, if required.

3

Our principal single-use tests are sold under the SNAP® name, and include a feline combination test, the SNAP® Combo FIV antibody/FeLV antigen test, which enables veterinarians to test simultaneously for feline immunodeficiency virus ("FIV") (which is similar to the human AIDS virus) and feline leukemia virus ("FeLV"); a canine combination test, the SNAP® 3Dx®, which tests simultaneously for Lyme disease, Ehrlichia canis and heartworm; a canine heartworm-only test; a canine test for parvovirus; a feline test for FeLV only; and canine and feline tests for Giardia, a parasitic disease. Sales of heartworm tests are greater in the first half of our fiscal year due to seasonality of the disease.

In addition to our single-use tests, we sell a line of microwell-based test kits, under the PetChek® name, that are used by larger clinics and laboratories to test multiple samples. PetChek® tests offer accuracy, ease of use and cost advantages to high-volume customers. We currently sell PetChek® tests for canine heartworm disease, FIV, and FeLV.

Veterinary Reference Laboratory and Consulting Services

We offer commercial veterinary reference laboratory and consulting services to veterinarians in the U.S., Europe, Australia and Japan. Veterinarians use our services by submitting samples by courier or overnight delivery to one of our facilities. Our laboratories offer a large selection of tests and diagnostic panels to detect a number of disease states and other conditions in production and companion animals.

Additionally, we provide specialized veterinary consultation, telemedicine and advisory services, including cardiology, radiology, internal medicine and ultrasound consulting. These services permit veterinarians to obtain readings and interpretations of test results transmitted by telephone and over the Internet from the veterinarians' offices.

Practice Information Management and Digital Radiography Systems

Practice Information Management Systems and Services. We develop, market and sell practice information management systems ("PIMS") including hardware and software that run key functions of veterinary clinics, including patient electronic health records management, scheduling (including boarding and grooming), billing and inventory management. Our principal system is the Cornerstone® system. We believe we are one of the leading providers of veterinary practice information management systems in North America, with an installed base of more than 7,200 of the approximately 28,000 veterinary hospitals in North America. We also provide software and hardware support to our PIMS customers, and related supplies and services to veterinarian PIMS users in general, and we derive a significant portion of our revenues for this product line from ongoing service contracts

Digital Radiography Systems and Services. Our digital radiography systems capture radiograph images in digital form, replacing traditional x-ray film. Use of digital radiography systems eliminates the need for the film and processor, hazardous chemicals and darkroom required for the production of film images, and provides for image manipulation and enhancement through contrast management. We market and sell the IDEXX Digital Radiography System, which is appropriate for use in the small animal veterinary clinic, and two systems that are primarily used as portable units in ambulatory veterinary practices, such as equine practices: the IDEXX EquiviewTM Digital Radiography System and the IDEXX Digital Radiography Compact System. Our digital radiography systems use IDEXX-PACSTM picture archiving and communication system ("PACS") software for the viewing, manipulation, management, storage and retrieval of the digital images generated by the digital capture plate. The IDEXX-PACSTM software also permits images from our digital radiography systems to be integrated into patients' medical records in the Cornerstone[®] system, as well as transferred to other practice information management systems.

Pharmaceutical Products

We develop, market and sell therapeutics for the veterinary market. We currently market and sell four pharmaceutical products: PZI VET[®], an insulin product for the treatment of diabetic cats; ACAREXX[®] (.01% ivermectin) otic suspension for the treatment of ear mites in cats; SURPASS[®] (1% diclofenac sodium), a topical, nonsteroidal anti-inflammatory drug for equine use; and Navigator[®] (32% nitazoxanide) Antiprotozoal Oral Paste, a treatment for equine protozoal myeloencephalitis (EPM). We are developing a long-acting, injectable form of the antibiotic tilmicosin for cats.

4

WATER

We offer a range of products used in the detection of various microbiological analytes in water.

Our Colilert[®], Colilert[®]-18 and Colisure[®] tests simultaneously detect total coliforms and E. coli in water. These organisms are broadly used as indicators of microbial contamination in water. These products utilize indicator-nutrients that produce a change in color or fluorescence when metabolized by target microbes in the sample. Our water tests are used by government laboratories, water utilities and private certified laboratories to test drinking water in compliance with U.S. Environmental Protection Agency ("EPA") standards. The tests also are used in evaluating water used in production processes (for example, in beverage and pharmaceutical applications) and in evaluating bottled water, recreational water, waste water and water from private wells.

Our Enterolert[™] product detects enterococci in drinking and recreational waters. Our Quanti-Tray[®] products, when used in conjunction with our Colilert[®], Colilert[®]-18, Colisure[®] or Enterolert[™] products, provide users quantitative measurements of microbial contamination, rather than a presence/absence indication. The Colilert[®], Colilert-18[®], Colisure[®] and Quanti-Tray[®] products have been approved by the EPA and by regulatory agencies in certain other countries.

Our Filta-Max[®] product is used in the detection of *Cryptosporidia* in water. *Cryptosporidia* are parasites that can cause potentially fatal gastrointestinal illness if ingested. Testing of water supplies for *Cryptosporidia* is mandated by regulation in the United Kingdom, but is not regulated in other countries at this time.

FOOD DIAGNOSTICS GROUP

Production Animal Services

We sell diagnostic tests and related instrumentation and software that are used to detect a wide range of diseases and to monitor health status in production animals. Our production animal products are purchased primarily by government laboratories and poultry and swine producers. Significant products include diagnostic tests for porcine

reproductive and respiratory syndrome and pseudorabies virus in pigs; Newcastle disease in poultry; and Johne's disease, bovine viral diarrhea virus, and brucellosis in cattle. In December 2004, we completed the acquisition of Dr. Bommeli AG, a Swiss manufacturer of production animal tests, for cash consideration of approximately \$15.8 million, net of cash acquired.

We have developed a postmortem test for bovine spongiform encephalopathy ("BSE" or "mad cow disease"). This test was approved for use in the U.S. by the United States Department of Agriculture ("USDA") in 2004 and for use in the European Union ("EU") by the European Commission in February 2005. We also offer a related kit for the detection of a similar disease, scrapie, in small ruminants, including sheep. Testing for BSE in the U.S. is limited and we do not know when or if the USDA will expand its testing program, which would increase the domestic market for these tests.

Dairy Testing

Our principal product for use in testing for antibiotic residue in milk is the SNAP® beta-lactam test. Dairy producers and processors use our tests for quality assurance of raw milk, and government and food-quality managers use them for ongoing surveillance.

In March 2003, we entered into an agreement with the FDA under which we agreed, among other things, to perform specified lot release and stability testing of our SNAP® beta-lactam products and to provide related data to the FDA. If the FDA were to determine that one or more lots of product failed to meet applicable criteria for product performance or stability, the FDA could take various actions, including requiring us to recall products or restricting our ability to sell these products. Sales of dairy antibiotic residue testing products were \$15.6 million in 2005.

MARKETING AND DISTRIBUTION

We market, sell and service our products worldwide through our marketing, sales and technical service groups, as well as through independent distributors and other resellers. We maintain sales offices outside the U.S. in Australia, China, France, Germany, Italy, Japan, the Netherlands, Spain, Switzerland, Taiwan and the United Kingdom. In 2005, 2004 and 2003, we spent \$102.0 million, \$85.7 million, and \$71.8 million or 16%, 16%, and 15% of sales, respectively, on sales and marketing.

5

Generally, we select the appropriate distribution channel for our products based on the type of product, technical service requirements, number and concentration of customers, regulatory requirements and other factors. We market our veterinary diagnostic and pharmaceutical products to veterinarians both directly and through independent veterinary distributors in the U.S., with most instruments sold directly by IDEXX sales personnel, and test kits, pharmaceutical products and instrument consumables supplied primarily by the distribution channel. Outside the U.S., we sell our veterinary diagnostic products through our direct sales force and, in certain countries, through distributors and other resellers. We sell our reference laboratory services worldwide through our direct sales force. We market our software products through our direct sales force primarily in the U.S. We market our water and food diagnostics products primarily through our direct sales force in the U.S. and Canada. Outside the U.S. and Canada, we market these products through selected independent distributors and, in certain countries, through our direct sales force.

In 2005, two of our CAG distributors, The Butler Company and Burns Veterinary Supply, Inc., merged and, as a result, they collectively accounted for 10% of our 2005 revenue and 4% of our net accounts receivable at December 31, 2005. In 2004 and 2003, no customer accounted for greater than 10% of our revenue. Our largest customers are our U.S. distributors of our products in the CAG segment. The largest consumer of our products and services accounts for approximately 1% of our sales.

RESEARCH AND DEVELOPMENT

Our business includes the development and introduction of new products and may involve entry into new business areas. Our research and development activity is focused primarily on development of new diagnostic instrument platforms and information systems, new immunoassay devices, new diagnostic tests, new animal drugs, enhanced practice information systems, and improvements in the performance, connectivity and interoperability of our products and services. Our research and development expenses, which consist of salaries, employee benefits, materials and consulting costs, were approximately \$40.9 million, \$35.4 million and \$32.3 million, or 6%, 6% and 7% of sales, in 2005, 2004 and 2003, respectively.

PATENTS AND LICENSES

We actively seek to obtain patent protection in the U.S. and other countries for inventions covering our products and technologies. We also license patents and technologies from third parties. These licenses include an exclusive royalty-bearing license of certain patents relating to diagnostic products for FIV that expire in 2009, from The Regents of the University of California; and exclusive licenses from Tulane University and the University of Texas to certain patents and patent applications expiring beginning in 2019 that relate to the detection of Lyme disease. We also have an exclusive royalty-bearing license of certain patents expiring in 2007 relating to defined substrate technology ("DST") that is utilized in the Colilert[®], Colilert-18[®], Colisure[®] and EnterolertTM water-testing products, although we do not believe the expiration of the DST patents in 2007 will have a material effect on our water business. In addition, we hold a U.S. patent expiring in 2014 that specifically covers the Colilert®-18 product and another patent expiring in 2014 that relates to certain methods and kits for simultaneously detecting antigens and antibodies, and which covers certain of our SNAP[®] products, including our SNAP[®] Combo FIV/FeLV and Canine SNAP[®] 3Dx[®] combination tests.

To the extent some of our products may now, or in the future, embody technologies protected by patents, copyrights or trade secrets of others, we may be required to obtain licenses to such technologies in order to continue to sell our products. These licenses may not be available on commercially reasonable terms or at all. Our failure to obtain any such licenses may delay or prevent the sale of certain new or existing products. See "Part I, Item 1A. Risk Factors."

6

PRODUCTION AND SUPPLY

VetTest[®] Chemistry Analyzers are manufactured for us by Tokyo Parts Industrial Company, Ltd. under an agreement that renews annually unless either party notifies the other of its decision not to renew. VetTest[®] slides are supplied exclusively by Ortho under supply agreements with Ortho (the "Ortho Agreements"). We are required to purchase all of our requirements for our current menu of VetTest[®] slides from Ortho to the extent Ortho is able to supply those requirements. In addition, we have committed to minimum annual purchase volumes of certain VetTest[®] slides through 2010. The Ortho Agreements expire on December 31, 2020.

The VetAutoreadTM Hematology Analyzer is manufactured for us by QBC Diagnostics, Inc. ("QBCD") under a supply agreement that expires on December 31, 2020. The VetLyte[®] Electrolyte Analyzer is manufactured for us by Roche Diagnostics Corporation under an agreement that requires Roche Diagnostics to supply analyzers through December 31, 2006, and consumables and spare parts through December 31, 2013. The VetStatTM Electrolyte and Blood Gas Analyzer is manufactured for us by Osmetech, Inc.

under an agreement that requires Osmetech to supply analyzers and consumables through 2015 and we have an option to extend this agreement for an additional four years. We have certain minimum purchase obligations under these agreements.

We purchase certain other products, raw materials and components from a single supplier. These include active ingredients for our pharmaceutical products, certain digital radiography systems, instrument consumables, and certain components used in our SNAP® rapid assay devices, water testing products and LaserCyte® Hematology Analyzers. We have in the past been successful in ensuring an uninterrupted supply of products purchased from single source suppliers. However, there can be no assurance that uninterrupted supply can be maintained if these agreements terminate for any reason or our suppliers otherwise are unable to satisfy our requirements for products.

We do not generally maintain significant backlog and believe that our backlog at any particular date historically has not been indicative of future sales.

COMPETITION

We face intense competition within the markets in which we sell our products and services. We expect that future competition will become even more intense, and that we will have to compete with changing technologies, which could affect the marketability of our products and services. Our competitive position also will depend on our ability to develop proprietary products, attract and retain qualified scientific and other personnel, develop and implement production and marketing plans, obtain or license patent rights, and obtain adequate capital resources.

We compete with many companies ranging from small businesses focused on animal health to large pharmaceutical companies. Our competitors vary in our different markets. Academic institutions, governmental agencies and other public and private research organizations also conduct research activities and may commercialize products, which could compete with our products, on their own or through joint ventures. Some of our competitors have substantially greater capital, manufacturing, marketing, and research and development resources than we do.

Competitive factors in our different business areas are detailed below:

- Veterinary diagnostic products and food and water testing products. We compete primarily on the basis of the ease of use, speed, accuracy and other performance characteristics of our products and services (including unique tests), the breadth of our product line and services, the effectiveness of our sales and distribution channels, the quality of our technical and customer service, and our pricing relative to the value of our products.
- <u>Veterinary laboratory and consulting services</u>. In this market, we compete primarily on the basis of quality, service, technology, and our pricing relative to the value of our services. We compete in certain geographic locations in the U.S. with Antech Diagnostics, a unit of VCA Antech, Inc.
- Veterinary pharmaceuticals. We compete primarily on the basis of the performance characteristics of our products.
- <u>Practice Information Management and Digital Radiography Systems</u>. We compete primarily on the basis of ease of use, connectivity to equipment and other systems, performance characteristics, effectiveness of our customer service, information handling capabilities, advances in technologies, and our pricing relative to the value of our products and services.

7

GOVERNMENT REGULATION

Many of our products are subject to regulation by U.S. and foreign regulatory agencies. The following is a description of the principal regulations affecting our businesses.

Veterinary diagnostic products. Most diagnostic tests for animal health applications are veterinary biological products that are regulated in the U.S. by the Center for Veterinary Biologics within the USDA Animal and Plant Health Inspection Service ("APHIS"). The APHIS regulatory approval process involves the submission of product performance data and manufacturing documentation. Following regulatory approval to market a product, APHIS requires that each lot of product be submitted for review before release to customers. In addition, APHIS requires special approval to market products where test results are used in part for government-mandated disease management programs. A number of foreign governments accept APHIS approval as part of their separate regulatory approvals. However, compliance with an extensive regulatory process is required in connection with marketing diagnostic products in Japan, Germany, the Netherlands and many other countries. We also are required to have a facility license from APHIS to manufacture USDA-licensed products. We have obtained such a license for our manufacturing facility in Westbrook, Maine and Memphis, Tennessee.

Our instrument systems are medical devices regulated by the U.S. Food and Drug Administration ("FDA") under the Food, Drug and Cosmetics Act (the "FDC Act"). While the sale of these products does not require premarket approval by FDA and does not subject us to the FDA's Good Manufacturing Practices regulations ("GMPs"), these products must not be adulterated or misbranded under the FDC Act.

<u>Veterinary pharmaceuticals</u>. The manufacture and sale of veterinary pharmaceuticals are regulated by the Center for Veterinary Medicine ("CVM") of the FDA. A new animal drug may not be commercially marketed in the U.S. unless it has been approved as safe and effective by CVM. Approval may be requested by filing an NADA with CVM containing substantial evidence as to the safety and effectiveness of the drug. Data regarding manufacturing methods and controls also are required to be submitted with the NADA. Manufacturers of animal drugs must also comply with GMPs and Good Laboratory Practices ("GLPs"). Sales of animal drugs in countries outside the U.S. require compliance with the laws of those countries, which may be extensive.

Water testing products. Our water tests are not subject to formal premarket regulatory approval. However, before a test can be used as part of a water-quality monitoring program in the U.S. that is required by the EPA, the test must first be approved by the EPA. The EPA approval process involves submission of extensive product performance data in accordance with an EPA-approved protocol, evaluation of the data by the EPA and publication for public comment of any proposed approval in the Federal Register before final approval. Our Colilert®, Colilert®-18, Colisure®, Quanti-Tray®, Filta-Max® and SimPlate® for heterotropic plate counts ("HPC") products have been approved by the EPA. The sale of water-testing products also is subject to extensive and lengthy regulatory processes in many other countries around the world.

<u>Dairy testing products</u>. The sale of dairy testing products in the U.S. is regulated by the FDA in conjunction with the AOAC Research Institute ("AOAC RI"). Before a product can be sold, extensive product performance data must be submitted in accordance with a protocol that is approved by the FDA and the AOAC RI. Following approval of a product by the FDA, the product must also be approved by the National Conference on Interstate Milk Shipments ("NCIMS"), an oversight body that includes state, federal and industry representatives. Our dairy antibiotic residue testing products have been approved by the FDA and NCIMS. While some foreign countries accept AOAC RI approval as part of their regulatory approval process, many countries have separate regulatory processes.

Any acquisitions of new products and technologies may subject us to additional areas of government regulation. These may involve food, drug and water-quality regulations of the FDA, the EPA and the USDA, as well as state, local and foreign governments. See "Part I, Item 1A. Risk Factors."

EMPLOYEES

ITEM 1A. RISK FACTORS

Our future operating results involve a number of risks and uncertainties. Actual events or results may differ materially from those discussed in this report. Factors that could cause or contribute to such differences include, but are not limited to, the factors discussed below, as well as those discussed elsewhere in this report.

IDEXX's Future Growth and Profitability Depend on Several Factors

The future success of our business depends upon our ability to successfully implement various strategies, including:

- Developing, manufacturing and marketing new products with new features and capabilities, including pharmaceutical products; a new clinical chemistry
 instrument; and rapid assay, water testing and production animal diagnostic products, as well as improving and enhancing existing products;
- Developing and implementing new technology and licensing strategies; and identifying, completing and integrating acquisitions that enhance our existing businesses or create new business areas for us;
- Increasing the value to our customers of our companion animal products and services by enhancing the connectivity of these products, including the connectivity among the IDEXX VetLab[®] instrument suite, Cornerstone[®] practice information management system, the IDEXX-PACSTM software and IDEXX Reference Laboratories;
- Expanding our market by expanding the installed base of our instrumentation through customer acquisition and retention and increasing use of our products by our customers;
- Strengthening our sales and marketing activities both within the U.S. and in geographies outside of the U.S.; and
- Reducing the costs of manufacturing our products and providing services through operating efficiencies and increased focus on quality.

However, we may not be able to successfully implement some or all of these strategies and increase or sustain our rate of growth or profitability.

Our Products and Services Are Subject to Various Government Regulations

In the U.S., the manufacture and sale of our products are regulated by agencies such as the U.S. Department of Agriculture ("USDA"), U.S. Food and Drug Administration ("FDA") and the U.S. Environmental Protection Agency ("EPA"). Most diagnostic tests for animal health applications, including our canine, feline, poultry and livestock tests, must be approved by the USDA prior to sale. Our water testing products must be approved by the EPA before they can be used by customers in the U.S. as a part of a water quality monitoring program required by the EPA. Our pharmaceutical and dairy testing products require approval by the FDA. The manufacture and sale of our products are subject to similar laws in many foreign countries. Any failure to comply with legal and regulatory requirements relating to the manufacture and sale of our products in the U.S. or in other countries could result in fines and sanctions against us or removals of our products from the market, which could have a material adverse effect on our results of operations.

We have entered into an agreement with the FDA under which we have agreed, among other things, to perform specified lot release and stability testing of our SNAP® betalactam dairy-testing products and to provide related data to the FDA. If the FDA were to determine that one or more lots of product failed to meet applicable criteria for product performance or stability, the FDA could take various actions, including requiring us to recall products or restricting our ability to sell these products. Sales of dairy antibiotic residue-testing products were \$15.6 million for the year ended December 31, 2005.

Commercialization of animal health pharmaceuticals in the U.S. requires prior approval by the FDA. To obtain such approvals, we are required to submit substantial clinical, manufacturing and other data to the FDA. Regulatory approval for products submitted to the FDA may take several years and, following approval, the FDA continues to regulate all aspects of the manufacture, labeling, storage, record keeping and promotion of pharmaceutical products. Failure to obtain, or delays in obtaining, FDA approval for new pharmaceutical products would have a negative impact on our future growth.

9

We Purchase Materials for Our Products from a Limited Number of Sources

We currently purchase many products and materials from single sources or a limited number of sources. Some of the products that we purchase from these sources are proprietary, and, therefore, cannot be readily replaced by alternative sources. These products include our VetTest[®] Chemistry, VetAutoreadTM Hematology, VetLyte[®] Electrolyte, and VetStatTM Electrolyte and Blood Gas Analyzers and related consumables; certain digital radiography system components, specifically image capture plates and readers; active ingredients for pharmaceutical products; and certain components of our SNAP[®] rapid assay devices, water testing products and LaserCyte[®] Hematology Analyzers. If we are unable to obtain adequate quantities of these products in the future, we could face cost increases or reductions or delays in product shipments, which could have a material adverse effect on our results of operations.

The slides sold for use in our VetTest[®] Chemistry Analyzers are purchased under an agreement with Ortho that, as of December 31, 2005, required us to purchase a minimum of \$92.7 million of slides through 2010. We purchase our electrolyte instruments, components and consumables under an agreement with Roche Diagnostics, under which we are required to purchase a minimum of \$4.1 million of these products through 2006. We purchase our VetAutoreadTM Hematology Analyzers, components and consumables under an agreement with QBCD, under which we are required to make aggregate minimum purchases of \$18.0 million through 2020. If demand for any of the products purchased under these agreements is insufficient to support our minimum purchase obligations for those products, we could incur losses related to those obligations. In addition, because we purchase the products at predetermined prices, our profits on sales of these products could decline if we are unable to maintain current pricing levels for such products.

Our Biologic Products Are Complex and Difficult to Manufacture

Many of our rapid assay and production animal diagnostic products are biologics, which are products that are comprised of materials from living organisms, such as antibodies, cells and sera. Manufacturing biologic products is highly complex. Unlike products that rely on chemicals for efficacy (such as most pharmaceuticals), biologics are difficult to

characterize due to the inherent variability of biological input materials. Difficulty in characterizing biological materials or their interactions creates greater risk in the manufacturing process. We attempt to mitigate risk associated with the manufacture of biologics by continuing to improve the characterization of all of our input materials, utilizing multiple vendors, manufacturing some of these materials ourselves and maintaining substantial inventories of materials that have demonstrated the appropriate characteristics. However, there can be no assurance that we will be able to maintain adequate sources of biological materials or that biological materials that we maintain in inventory will yield finished products that satisfy applicable product release criteria. Our inability to obtain necessary biological materials or to successfully manufacture biologic products that incorporate such materials could have a material adverse effect on our results of operations.

Our Success Is Heavily Dependent Upon Our Proprietary Technologies

We rely on a combination of patent, trade secret, trademark and copyright laws to protect our proprietary rights. If we do not have adequate protection of our proprietary rights, our business may be affected by competitors who develop substantially equivalent technologies that compete with us.

We cannot ensure that we will obtain issued patents, that any patents issued or licensed to us will remain valid, or that any patents owned or licensed by us will provide protection against competitors with similar technologies. Even if our patents cover products sold by our competitors, the time and expense of litigating to enforce our patent rights could be substantial, and could have a material adverse effect on our results of operations. In addition, expiration of patent rights could result in substantial new competition in the markets for products previously covered by those patent rights.

In the past, we have received notices claiming that our products infringe third-party patents and we may receive such notices in the future. Patent litigation is complex and expensive, and the outcome of patent litigation can be difficult to predict. We cannot ensure that we will win a patent litigation case or negotiate an acceptable resolution of such a case. If we lose, we may be stopped from selling certain products and/or we may be required to pay damages and/or ongoing royalties as a result of the lawsuit. Any such adverse result could have a material adverse effect on our results of operations.

10

Our Sales Are Dependent on Distributor Purchasing Patterns

We sell many of our products, including substantially all of the rapid assays and instrument consumables sold in the U.S., through distributors. Because significant product sales are made to a limited number of customers, unanticipated changes in the timing and size of distributor purchases can have a negative effect on quarterly results. Our financial performance, therefore, is subject to an unexpected downturn in product demand and may be unpredictable.

Distributors of veterinary products have entered into business combinations resulting in fewer distribution companies. Consolidation within distribution channels would increase our customer concentration level, which could increase the risks described in the preceding paragraph.

Our Markets Are Competitive and Subject to Rapid and Substantial Technological Change

We face intense competition within the markets in which we sell our products and services. We expect that future competition will become even more intense, and that we will have to compete with changing and improving technologies. Some of our competitors and potential competitors, including large pharmaceutical and diagnostic companies, have substantially greater capital, manufacturing, marketing, and research and development resources than we do.

Changes in Diagnostic Testing Could Negatively Affect Our Operating Results

The market for diagnostic tests could be negatively impacted by the introduction or broad market acceptance of vaccines or preventatives for the diseases and conditions for which we sell diagnostic tests and services. Eradication or substantial declines in the prevalence of certain diseases also could lead to a decline in diagnostic testing for such diseases. Our production animal services business in particular is subject to fluctuations resulting from changes in disease prevalence and government-mandated testing programs. Such declines in diagnostic testing could have a material adverse effect on our results of operations.

International Revenue Accounts for a Significant Portion of Our Total Revenue

For the year ended December 31, 2005, 34% of our revenue was attributable to sales of products and services to customers outside the U.S. Various risks associated with foreign operations may impact our international sales. Possible risks include fluctuations in the value of foreign currencies, disruptions in transportation of our products, the differing product and service needs of foreign customers, difficulties in building and managing foreign operations, import/export duties and quotas, and unexpected regulatory, economic or political changes in foreign markets. Prices that we charge to foreign customers may be different than the prices we charge for the same products in the U.S. due to competitive, market or other factors. As a result, the mix of domestic and international sales in a particular period could have a material impact on our results for that period. In addition, many of the products for which our selling price may be denominated in foreign currencies are manufactured, sourced, or both, in the U.S. and our costs are incurred in U.S. dollars. We utilize nonspeculative forward currency exchange contracts to mitigate foreign currency exposure, however, an appreciation of the U.S. dollar relative to the foreign currencies in which we sell these products would reduce our operating margins.

The Loss of Our President, Chief Executive Officer and Chairman Could Adversely Affect Our Business

We rely on the management and leadership of Jonathan W. Ayers, our President, Chief Executive Officer and Chairman. We do not maintain key man life insurance coverage for Mr. Ayers. The loss of Mr. Ayers could have a material impact on our business.

11

We Could Be Subject to Class Action Litigation Due to Stock Price Volatility, which, if Occurs, Could Result in Substantial Costs or Large Judgments Against Us

The market for our common stock may experience extreme price and volume fluctuations, which may be unrelated or disproportionate to our operating performance or prospects. In the past, securities class action litigation has often been brought against companies following periods of volatility in the market prices of their securities. We may be the target of similar litigation in the future. Securities litigation could result in substantial costs and divert our management's attention and resources, which could have a negative effect on our business, operating results and financial condition.

Our prior operating results have fluctuated due to a number of factors, including seasonality of certain product lines; changes in our accounting estimates; the impact of acquisitions; timing of distributor purchases, product launches, research and development expenditures, litigation and claim-related expenditures; changes in competitors' product offerings; and other matters. Similarly, our future operating results may vary significantly from quarter to quarter due to these and other factors, many of which are beyond our control. If our operating results or projections of future operating results do not meet the expectations of market analysts or investors in future periods, our stock price may fall.

Future Operating Results Could Be Materially Affected By the Resolution of Various Uncertain Tax Positions and Adversely Affected by Potential Changes to Tax Incentives

In the ordinary course of our business, there are many transactions and calculations where the ultimate tax determination is uncertain. Significant judgment is required in determining our worldwide provision for income taxes and our income tax filings are regularly under audit by tax authorities. We believe that we have adequately accrued for all potential tax liabilities and, although we believe our tax estimates are reasonable, the final determination of tax audits could be materially different than that which is reflected in historical income tax provisions and accruals. Additionally, we benefit from certain tax incentives offered by various jurisdictions. If we are unable to meet the requirements of such incentives, our inability to use these benefits could have a material negative effect on future earnings.

ITEM 1B. UNRESOLVED STAFF COMMENTS

Not applicable.

ITEM 2. PROPERTIES

We lease approximately 329,000 square feet of office and manufacturing space in Westbrook, Maine, under leases expiring in 2013 and 2018. On January 17, 2006, we entered into an agreement with the owner to purchase the building in which our headquarters facility is located, which includes an additional 130,000 square feet of vacant space, for \$18.0 million less the face value of the existing mortgage of approximately \$6.5 million. Our acquisition of this facility is subject to certain conditions to closing, including completion of our due diligence review and receipt of certain state and local incentives. We lease approximately 97,500 square feet of industrial space in Memphis, Tennessee, for use as a distribution facility, under a lease expiring in 2013; approximately 40,000 square feet of office and manufacturing space in Eau Claire, Wisconsin, for our veterinary practice information management software business, under a lease expiring in 2009; and approximately 48,000 square feet of warehouse and office space in the Netherlands for use as our headquarters for European operations, under a lease expiring in 2008.

We also lease a total of approximately 43,000 square feet of smaller office, manufacturing and warehouse space in the U.S. and elsewhere in the world. In addition, we own or lease approximately 235,000 square feet of space in the U.S., Australia, Germany, Switzerland and the United Kingdom for use as veterinary reference laboratories and office space for our veterinary consulting services. Of this space, 63,000 square feet is owned by us and the remaining amount is leased, under leases having expiration dates up to the year 2012.

12

We consider that the properties are generally in good condition, are well-maintained, and are generally suitable and adequate to carry on our business.

ITEM 3. LEGAL PROCEEDINGS

We are not a party to any material legal proceedings.

ITEM 4. SUBMISSION OF MATTERS TO A VOTE OF SECURITY HOLDERS

No matters were submitted to a vote of security holders during the fourth quarter of the fiscal year covered by this report.

EXECUTIVE OFFICERS OF THE COMPANY

Our executive officers as of February 28, 2006 were as follows:

Name	Age	Title
Jonathan W. Ayers	49	President, Chief Executive Officer and Chairman of the Board of Directors
William C. Wallen, PhD	62	Senior Vice President and Chief Scientific Officer
Conan R. Deady	44	Vice President, General Counsel and Secretary
S. Sam Fratoni, PhD	58	Vice President and Chief Information Officer
Robert S. Hulsy	61	Vice President Laboratory Services
Jennifer A. Joiner	50	Vice President CAG North American Commercial Operations
Laurel E. LaBauve	47	Vice President Worldwide Operations
Ali Naqui, PhD	52	Vice President Water, Dairy, Asia Pacific and Latin America Operations
Merilee Raines	50	Vice President, Chief Financial Officer and Treasurer
Quentin J. Tonelli, PhD	57	Vice President Rapid Assay and Production Animal Services

Mr. Ayers has been Chairman of the Board, Chief Executive Officer and President of IDEXX since January 2002. Prior to joining IDEXX, from 1999 to 2001, Mr. Ayers was President of Carrier Corporation, the then-largest business unit of United Technologies Corporation, and from 1997 to 1999, he was President of Carrier Asia Pacific Operations. From 1995 to 1997, Mr. Ayers was Vice President, Strategic Planning at United Technologies. Before joining United Technologies, from 1986 to 1995, Mr. Ayers held various positions at Morgan Stanley & Co. in mergers and acquisitions and corporate finance. Prior to Morgan Stanley, Mr. Ayers was a strategy consultant for Bain & Company from 1983 to 1986 and was in the field sales organization of IBM's Data Processing Division from 1978 to 1981. Mr. Ayers holds an undergraduate degree in molecular biophysics and biochemistry from Yale University and graduated from Harvard Business School in 1983.

Dr. Wallen has been Senior Vice President and Chief Scientific Officer of IDEXX since September 2003. Prior to joining IDEXX, Dr. Wallen held various positions with Bayer Corporation, most recently as Senior Vice President, Research and Development, and Head, Office of Technology for the Diagnostics Division of Bayer Healthcare. From 2001 to 2003, Dr. Wallen served as Senior Vice President and Head of Research, Nucleic Acid Diagnostics Segment; from 1999 to 2001, as Senior Vice President of Research and Development Laboratory Testing Segment; and from 1993 to 1999, as Vice President of Research and Development, Immunodiagnostic and Clinical Chemistry Business Units. Before joining Bayer Corporation, from 1990 to 1993, Dr. Wallen was Vice President Research and Development at Becton Dickinson Advanced Diagnostics.

Mr. Deady has been a Vice President and General Counsel of the Company since 1999 and was Deputy General Counsel of the Company from 1997 to 1999. Before joining the Company in 1997, Mr. Deady was Deputy General Counsel of Thermo Electron Corporation, a manufacturer of technology-based instruments. Previously, Mr. Deady was a partner at Hale and Dorr LLP (now WilmerHale).

Dr. Fratoni has been a Vice President of the Company since May 1997 and Chief Information Officer since November 2000. He was President of the Company's Food and Environmental Group from July 1999 to December 2000. From May 1997 to July 1999, Dr. Fratoni was Vice President of Human Resources of the Company, and from October 1996 to May 1997, he was Director of Business Development for the Food and Environmental Group. Before joining the Company in October 1996, Dr. Fratoni held various positions with Hewlett-Packard Company.

Mr. Hulsy has been Vice President of the Company since February 1999 and President of the Company's IDEXX Reference Laboratories business since August 1998. Before joining the Company in August 1998, Mr. Hulsy was President of American Environmental Network, Inc., a network of environmental laboratories, from 1992 to 1998.

Ms. Joiner joined IDEXX as Vice President CAG North American Commercial Operations in August 2004. Prior to joining the Company, Ms. Joiner was Vice President, Marketing and Strategic Planning, of Molecular Staging, Inc., an emerging technology firm from 2000 to August 2004. From 1998 to 2000, Ms. Joiner was Vice President, Commercial Operations for the Diagnostics Division of Bayer Healthcare, and from 1996 to 1998, she was Managing Director, Australia and New Zealand, for GE Medical Systems.

Ms. LaBauve joined IDEXX as Vice President, Worldwide Operations in February 2004. From 1999 until 2004, Ms. LaBauve held various senior positions with the Ortho-Clincial Diagnostics subsidiary of Johnson & Johnson, including General Manager and Vice President, Clinical Laboratory Franchise, from 2002 to 2004; Vice President, Worldwide Systems R&D, from 2000 to 2002; and Vice President Design Excellence, from 1999 to 2000. Prior to joining Ortho, Ms. LaBauve held various positions with AlliedSignal Corporation, most recently serving as Vice President, Six Sigma Quality.

Dr. Naqui became a Vice President of the Company in January 2006 and oversees the Company's Water and Dairy testing businesses, as well as the Company's Asia Pacific and Latin American operations. Dr. Naqui served as Division Vice President, Water and Dairy from January 2000 to December 2005, General Manager, Water from September 1997 to January 2000, and Director of Research and Development from February 1993 to September 1997. Dr. Naqui joined the Company in 1993 as a result of the acquisition of Environetics, where he was the Director of Research and Development. Prior to joining Environetics, he was a Research and Development manager with Becton, Dickinson and Company.

Ms. Raines has been Chief Financial Officer since October 2003 and Vice President, Finance of the Company since May 1995. Ms. Raines served as Division Vice President, Finance from March 1995 to May 1995, Director of Finance from 1988 to March 1995 and Controller from 1985 to 1988.

Dr. Tonelli became a Vice President of the Company in June 2001 and currently oversees the Company's production animal services and rapid assay lines of business. Previously he held various positions with the Company, including Division Vice President for Research and Development and Division Vice President, Business Development. Before joining the Company in 1984, he was a Group Leader of Research and Development for the Hepatitis and AIDS Business Unit within the diagnostic division of Abbott Laboratories, Inc.

PART II.

ITEM 5. MARKET FOR THE REGISTRANT'S COMMON EQUITY, RELATED STOCKHOLDER MATTERS AND ISSUER PURCHASES OF SECURITIES

Our common stock is quoted on the NASDAQ Stock Market under the symbol IDXX. The table below shows the high and low sale prices per share of our common stock as reported on the NASDAQ Stock Market for the years 2005 and 2004.

CALENDAR YEAR	HIGH	LOW
2005		
First Quarter	\$ 58.23	\$ 52.18
Second Quarter	63.00	
Third Quarter	67.95	52.94 60.16
Fourth Quarter	75.14	61.11
2004		
First Quarter	\$ 57.57	\$ 45.30
Second Quarter	68.82	56.75
Third Quarter	64.50	45.43 46.80
Fourth Quarter	55.02	46.80

As of February 27, 2006, there were 949 holders of record of our common stock.

14

We have never paid any cash dividends on our common stock. From time to time our Board of Directors may consider the declaration of a dividend. However, we have no present intention to pay a dividend.

For the three months ended December 31, 2005, we repurchased our shares as described below:

Period	Total Number of Shares Purchased (a)	 Average Price Paid per Share (b)	Total Number of Shares Purchased as Part of Publicly Announced Plans or Programs (c)	Maximum Number of Shares that May Yet Be Purchased Under the Plans or Programs (d)	
October 1, 2005 to October 31, 2005	132,300	\$ 64.97	132,300	2,419,730	
November 1, 2005 to November 30, 2005	175,900	70.41	175,900	2,243,830	
December 1, 2005 to December 31, 2005	191,500	73.05	191,500	2,052,330	
Total	499,700	\$ 69.98	499,700	2,052,330	

Our Board of Directors has approved the repurchase of up to 16,000,000 shares of the Company's common stock in the open market or in negotiated transactions, under which 2,052,330 shares remained to be repurchased as of December 31, 2005. The plan was approved and announced on August 13, 1999, and subsequently amended on October 4, 1999, July 21, 2000, October 20, 2003, October 12, 2004, and October 12, 2005, and does not have a specified expiration date. During the year ended December 31, 2005, we repurchased

ITEM 6. SELECTED FINANCIAL DATA

The following table sets forth selected consolidated financial data of the Company for each of the five years ending with December 31, 2005. The selected consolidated financial data presented below have been derived from the Company's consolidated financial statements. These financial data should be read in conjunction with the consolidated financial statements, related notes and other financial information appearing elsewhere in this Form 10-K.

		For the	e Years E	nded December	31,		
	 2005	2004		2003		2002	2001*
(in thousands, except per share data)	 	 					
STATEMENT OF OPERATIONS DATA:							
Revenue	\$ 638,095	\$ 549,181	\$	475,992	\$	412,670	\$ 386,081
Cost of revenue	315,195	270,164		245,688		219,945	202,750
Gross profit	 322,900	 279,017		230,304		192,725	 183,331
Expenses:	322,700	277,017		250,501		1,2,720	103,331
Sales and marketing	101,990	85,710		71,846		56,794	57,087
General and administrative	64,631	49,870		45,752		40,787	41,266
Research and development	40,948	35,402		32,319		29,329	28,426
Income from operations	 115,331	 108,035		80,387		65,815	 56,552
Interest income	3,141	3,068		2,867		2,955	2,229
Income before provision for income							
taxes and partner's interest	118,472	111,103		83,254		68,770	58,781
Provision for income taxes	40,670	33,165		26,278		23,381	21,161
Partner's interest in loss of subsidiary	(452)	(394)		(114)			
Net income	\$ 78,254	\$ 78,332	\$	57,090	\$	45,389	\$ 37,620
Earnings per share:							
Basic	\$ 2.41	\$ 2.29	\$	1.67	\$	1.35	\$ 1.13
Diluted	\$ 2.30	\$ 2.19	\$	1.59	\$	1.30	\$ 1.09
Weighted average shares outstanding:							
Basic	32,521	34,214		34,271		33,622	33,293
Diluted	34,055	35,800		35,931		35,043	34,640
Dividends paid	\$ 	\$ 	\$		\$		\$
BALANCE SHEET DATA:							
Cash and investments	\$ 132,731	\$ 156,959	\$	255,787	\$	162,763	\$ 100,575
Working capital	192,679	201,640		270,244		217,740	164,199
Total assets	490,676	514,237		521,875		417,426	373,107
Total debt	551	1,810		494		973	8,380
Stockholders' equity	369,010	397,660		413,292		340,973	301,730

^{*} As a result of the adoption of Statement of Financial Accounting Standards No. 142, "Accounting for Goodwill and Other Intangible Assets", goodwill is no longer amortized commencing January 1, 2002. Goodwill amortization expense, net of tax, was \$4.5 million for the year ended December 31, 2001.

ITEM 7. MANAGEMENT'S DISCUSSION AND ANALYSIS OF FINANCIAL CONDITION AND RESULTS OF OPERATIONS

BUSINESS OVERVIEW

We operate primarily through three business segments: the Companion Animal Group ("CAG"), Water testing business ("Water") and the Food Diagnostics Group ("FDG"). CAG is comprised of the following product and service categories: instruments and consumables, rapid assays, reference laboratory and consulting services, practice information management and digital radiography systems, and pharmaceuticals. Water develops, designs, manufactures and distributes products to detect contaminants in water. FDG develops, designs, manufactures and distributes products to detect disease and contaminants in production animals and dairy products. Other items that are not included in our reportable segments are comprised primarily of corporate research and development and interest income.

16

For the three years ended December 31, 2005, revenues by product and service categories were as follows (in thousands):

December 31,						
 2005	2004			2003		
\$ 217,537	\$	197,939	\$	177,374		
100,255		93,506		82,978		
156,425		118,596		94,650		
				22,463		
14,024		10,483		6,954		
520,830		448,687		384,419		
 56,760		53,098		46,936		
44,945		31,690		28,580		
15,560		15,706		16,057		
 60,505		47,396		44,637		
\$ 638,095	\$	549,181	\$	475,992		
	\$ 217,537 100,255 156,425 32,589 14,024 520,830 56,760 44,945 15,560 60,505	\$ 217,537 \$ 100,255 156,425 32,589 14,024 520,830 56,760 44,945 15,560 60,505	2005 2004 \$ 217,537 \$ 197,939 100,255 93,506 156,425 118,596 \$ 32,589 28,163 14,024 10,483 \$ 520,830 448,687 \$ 56,760 53,098 \$ 44,945 31,690 15,560 15,706 \$ 60,505 47,396	2005 2004 \$ 217,537 \$ 197,939 \$ 100,255 93,506 \$ 156,425 \$ 118,596 118,596 \$ 32,589 \$ 28,163 14,024 10,483 \$ 520,830 \$ 448,687 448,687 44,945 53,098 \$ 44,945 \$ 31,690 15,706 60,505 47,396		

The following is a discussion of the strategic and operating factors that we believe have the most significant effect on the performance of our business.

Companion Animal Group

In the CAG segment, we believe we have developed a strategic advantage over companies with more narrow product or service offerings. The breadth of our products and services gives us scale in sales and distribution, permits us to offer integrated disease-management solutions that leverage the advantages of both point-of-care and outside laboratory testing, and facilitates the flow of medical and business information in the veterinary practice by connecting practice information software systems, including connecting the electronic health record with laboratory test data, in-clinic test data from our IDEXX VetLab® suite of analyzers, and radiographic data in the IDEXX-PACSTM software taken by our digital radiography systems.

In the U.S., we sell instrument consumables, rapid assays and pharmaceuticals primarily through distributors, and, therefore, our reported sales of these products are sales made to distributors, rather than sales to veterinarians, the end-users. Because distributor inventory levels and purchasing patterns may fluctuate, sales of a particular product line in a particular period may not always be representative of the underlying customer demand for the product. Therefore, we closely track sales of these products by our U.S. distributors to the veterinarians ("practice-level sales"), which we think provides a more accurate picture of the real growth rate for these products. In the discussion of results below, we note certain instances where we believe reported sales have been influenced, positively or negatively, by changes in distributor inventories.

Instruments and Consumables. Our instrument strategy is to provide veterinarians with an integrated set of instruments that, individually and together, provide superior diagnostic information in the clinic, enabling veterinarians to practice better medicine and, in doing so, achieve their practice objectives, including growth and economic success. We derive substantial revenues from the sale of consumables that are used in these instruments. During the early stage of an instrument's life cycle, we derive relatively greater revenues from instrument placements, while consumable sales become relatively more significant in later stages as the installed base of instruments increases and instrument placement revenues begin to decline.

We have a large installed base of VetTest[®] Chemistry Analyzers, and substantially all of our revenues from that product line are now derived from consumables sales, although in recent years we have grown the annual number of unit instrument placements through sales, lease, rental and other programs. Our long-term success in this area of our business is dependent upon new customer acquisition, customer retention and increased customer utilization of existing and new assays introduced on these instruments. To increase utilization, we seek to educate veterinarians about best medical practices that emphasize the importance of blood and urine chemistry testing for a variety of diagnostic purposes.

17

We purchase the consumables used in VetTest[®] Chemistry Analyzers from Ortho under a supply agreement that continues through 2020. This supply agreement provides us with a long-term source of slides at costs that improve annually through 2010, and also improve over the term of the agreement as a result of increasing volume. Under this agreement, we are developing and expect to introduce a next-generation chemistry analyzer for the veterinary market based on the Ortho dry-slide technology, and Ortho will supply us with slide consumables used in both the new instrument and the VetTest[®] Chemistry Analyzer. We do not expect this next-generation analyzer to be commercially available before the latter part of 2007.

In the fourth quarter of 2002, we introduced the LaserCyte[®] Hematology Analyzer, which provides more extensive hematological diagnostic information than our original platform, the VetAutoreadTM Hematology Analyzer. A substantial portion of LaserCyte[®] placements have been made at veterinary clinics that already own our VetAutoreadTM Hematology Analyzers. Although we have experienced growth in sales of hematology consumables, LaserCyte[®] consumable sales have been partially offset by declines in sales of VetAutoreadTM consumables. Because the gross margin percentage of LaserCyte[®] consumables exceeds the gross margin percentage of the VetAutoreadTM consumables, gross margin from hematology consumables is expected to increase with continued penetration of the LaserCyte[®] Hematology Analyzer. Our gross margins on LaserCyte[®] Hematology Analyzer sales have been low in the early years of the program due to higher manufacturing, service and warranty costs associated with a new analyzer. As we have gained experience with the analyzer, we have improved manufacturing efficiency and reduced warranty and service costs, which have improved gross margins on these products, particularly in 2005. While we expect that LaserCyte[®] gross margins will continue to improve, they will continue to have a negative impact on overall CAG gross margins.

With all of our instrument lines, we seek to differentiate our products based on breadth of diagnostic menu, flexibility of menu selection, accuracy, reliability, ease of use, ability to handle compromised samples, time to result, analytical capability of software, integration with the IDEXX VetLab[®] system, education and training, and superior sales and customer service. Our equipment and consumables typically are sold at a premium price to competitive offerings. Our success depends, in part, on our ability to maintain a premium price strategy.

Rapid Assays. Our rapid assay business comprises single-use kits for in-clinic testing and microwell-based kits for large clinic and laboratory testing for canine and feline diseases and conditions. Our rapid assay strategy is to develop, manufacture, market and sell proprietary tests with superior performance that address important medical needs. As in our other lines of business, we also seek to differentiate our products through superior sales and customer service. These products carry price premiums over competitive products that we believe do not offer equivalent performance and diagnostic capabilities, and which we believe do not include a similar level of support. We augment our product development and customer service efforts with sales and marketing programs that enhance medical awareness and understanding regarding our target diseases and the importance of diagnostic testing.

Reference Laboratory and Consulting Services. We believe that more than half of all diagnostic testing by U.S. veterinarians is done at outside reference laboratories such as our IDEXX Reference Laboratories. We attempt to differentiate our laboratory testing services from those of our competitors primarily on the basis of quality, customer service, technology employed and specialized test menu. Revenue growth in this business is achieved both through increased sales at existing laboratories and through the acquisition of new customers, including through laboratory acquisitions and opening new laboratories. In 2004, we acquired a laboratory in Columbus, Ohio, opened a laboratory in Seattle, Washington, and acquired Vet Med Lab, which is based in Germany and is the largest European veterinary reference laboratory. In 2005, we acquired laboratories in Switzerland, the United Kingdom, and France and acquired veterinary laboratory customer lists in the U.S. and Germany. Profitability of this business is largely the result of our ability to achieve efficiencies from both volume and operational improvements. New laboratories that we open typically will operate at a loss until testing volumes reach a level that permits profitability. Acquired laboratories frequently operate less profitably than our existing laboratories and those laboratories may not achieve profitability comparable to our existing laboratories for a year or more while we implement operating improvements. Therefore, in the short term, new and acquired reference laboratories generally will have a negative effect on the operating margin of the laboratory and consulting services business.

information solutions, backed by superior customer support and education, to allow the veterinarian to practice better medicine and achieve the practice's business objectives. We differentiate our software systems through continually enhanced functionality through regular software releases. Our veterinary-specific digital radiography systems allow veterinarians to capture digital radiographs with ease and without the use of hazardous chemicals. The digital radiography systems also incorporate IDEXX-PACSTM picture archiving and communication software developed by IDEXX that allows for image enhancement, manipulation, storage and retrieval, and integration with the practice information software. Our strategy in digital radiography is to offer a system that provides superior image quality and software capability at a competitive price, backed by the same customer support provided for our other products and services in the Companion Animal Group.

Water

Our strategy in the water testing business is to develop, manufacture, market and sell proprietary products with superior performance, supported by exceptional customer service. Our customers are primarily water utilities to whom strong relationships and customer support are very important. Over the past several years, the rate of growth of this product line has slowed as a result of increased competition and market penetration. International sales of water-testing products represented 41% of total water product sales in 2005, and we expect that future growth in this business will be significantly dependent on our ability to increase international sales. Growth also will be dependent on our ability to enhance and broaden our product line. Most water microbiological testing is driven by regulation, and, in many countries, a test may not be used for regulatory testing unless it has been approved by the applicable regulatory body. As a result, we maintain an active regulatory program under which we are seeking regulatory approvals in a number of countries, primarily in Europe.

Food Diagnostics Group

Production Animal Services. We develop, manufacture, market and sell a broad range of tests for various poultry, cattle and swine diseases and conditions, and have an active research and development and in-licensing program in this area. Our strategy is to offer proprietary tests with superior performance characteristics. Disease outbreaks are episodic and unpredictable, and certain diseases that are prevalent at one time may be substantially contained or eradicated. In response to outbreaks, testing initiatives may lead to exceptional demand for certain products in certain periods. Conversely, successful eradication programs may result in significantly decreased demand for certain products. The performance of this business, therefore, can be subject to fluctuation. In 2005, approximately 77% of our sales in this business was international. Because of the significant dependence of this business on international sales, the performance of the business is particularly subject to the various risks described below that are associated with doing business internationally.

In 2004, we received USDA approval of our postmortem test for BSE (mad cow disease) and, in February 2005, we were informed that this test was approved by the European Commission for sale in EU member countries. While BSE testing is very limited in the U.S., a larger market for BSE testing exists in Europe.

Dairy Testing. Our strategy in the dairy testing business is to develop, manufacture and sell antibiotic residue testing products that satisfy applicable regulatory requirements for testing of bulk milk by producers and provide reliable field performance. These testing products use, almost exclusively, the SNAP® platform and manufacturing processes of our rapid assay business, incorporating customized reagents for antibiotic detection. Sales of dairy testing products have declined slightly over the last several years largely as a result of increased competition in the domestic market. To increase sales of dairy testing products, we look to increase penetration in geographies outside the United States and in the farm segment of the dairy market, and to develop product line enhancements and extensions.

CRITICAL ACCOUNTING POLICIES AND ESTIMATES

The discussion and analysis of our financial condition and results of operations are based upon our consolidated financial statements, which have been prepared in accordance with accounting principles generally accepted in the U.S. The preparation of these financial statements requires us to make estimates and judgments that affect the reported amounts of assets, liabilities, revenues and expenses, and related disclosure of contingent assets and liabilities. On an ongoing basis, we evaluate our estimates, including those related to inventory, goodwill and other intangible assets, warranty reserves, income taxes, contingencies, and revenue recognition. We base our estimates on historical experience and on various other assumptions that we believe to be reasonable under the circumstances, the results of which form the basis for making judgments about the carrying values of assets and liabilities that are not readily apparent from other sources. Actual results may differ from these estimates. Note 2 to the consolidated financial statements for the year ended December 31, 2005 included in this Form 10-K describes the significant accounting policies used in preparation of these financial statements.

19

We believe the following critical accounting policies reflect our more significant judgments and estimates used in the preparation of our consolidated financial statements.

Inventory

Inventories include material, labor and overhead, and are stated at the lower of cost (first-in, first-out) or market. We write down inventory for estimated obsolescence when warranted by estimates of future demand and market conditions. If actual market conditions are less favorable than those we estimated, additional inventory write-downs may be required, which would have a negative effect on results of operations. Certain major components of inventory are discussed in more detail below.

LaserCyte[®] Hematology Analyzer. At December 31, 2005 and 2004, our net inventories included \$9.8 million and \$11.9 million, respectively, of component parts and finished goods associated with our LaserCyte[®] hematology instrument. In addition, we had firm purchase commitments for an additional \$2.6 million of component parts as of December 31, 2005. At December 31, 2005 and 2004, \$2.3 million and \$1.9 million of the net LaserCyte[®] inventory, respectively, required rework before it could be used to manufacture finished goods. At December 31, 2005 and 2004, the inventory subject to rework was net of \$0.7 million and \$0.3 million write-downs, respectively, for inventory estimated to be obsolete. We expect to fully realize our net investment in inventory and purchase commitments. However, if we alter the design of this product, we may be required to write off some or all of the remaining associated inventory.

VetTest Chemistry® Slides. At December 31, 2005 and 2004, our net inventories included \$14.4 million and \$22.7 million, respectively, of slides used in our VetTest® Chemistry Analyzers. The decrease in slide inventory at December 31, 2005, compared to December 31, 2004, was primarily due to the delay of inventory receipts from the fourth quarter of 2005 to the first quarter of 2006. Most of the slides have a shelf-life of 24 months at the date of manufacture. The average remaining shelf-life at December 31, 2005 was 16.4 months. In addition, we are required to purchase a minimum of \$92.7 million of slides from Ortho through December 31, 2010. During the quarter ended December 31, 2003, we entered into a new contract with Ortho, which extended the term of the supply agreement through 2018 and left the contract minimum purchase commitments unchanged. In June 2005, we further amended this agreement to, among other things, extend its term from 2018 to 2020. As a result of the current and projected demand for VetTest® slides, our commitment to develop a next-generation chemistry analyzer that will utilize these slides, and the ratable decrease in required annual slide purchases from Ortho through 2010, we believe that we will not incur a loss under the contract. See Note 11 to the consolidated financial statements for the year ended December 31, 2005 included in this Form 10-K for additional discussion of our development commitment.

Nitazoxanide. Our nitazoxanide product, Navigator®, for the treatment of equine protozoal myeloencephalitis ("EPM") was approved by the FDA in November 2003. At December 31, 2005, our inventories included \$9.4 million of inventory associated with Navigator®, consisting of \$0.2 million of finished goods and \$9.2 million of active ingredient and other raw materials. In December 2004, we entered into an amendment to our agreement with our supplier of nitazoxanide under which we paid the supplier \$0.9 million in January 2005 and the supplier agreed until 2017 to replace any expiring inventory of nitazoxanide with longer-dated material. The payment was capitalized as inventory

cost. We believe that this agreement has substantially mitigated the risk that we would be required to write down nitazoxanide inventory due to its anticipated expiration prior to sale

Valuation of Goodwill and Other Intangible Assets

Intangible assets, other than goodwill, are valued at fair value when acquired. If a market value is not readily available, the fair value of the intangible asset is estimated based on expected cash flows of the associated business acquired that are attributable to the intangible asset. Goodwill is initially valued based on the excess of the purchase price of a business combination over the other net assets acquired.

20

We assess the impairment of identifiable intangible assets and other long-lived assets whenever events or changes in circumstances indicate that the carrying value may not be recoverable. Factors we consider important that could trigger an impairment review include, but are not limited to, the following:

- Significant under-performance relative to historical or projected future operating results;
- Failure to obtain regulatory approval of certain products;
- Significant changes in the manner of our use of the acquired assets or the strategy for our overall business;
- Significant increase in the discount rate assumed to calculate the present value of future cash flow;
- Significant negative industry or economic trends;
- Significant advancements or changes in technology; and
- Cancellation or significant changes in contractual relationships.

We continually assess the realizability of intangible assets other than goodwill in accordance with Statement of Financial Accounting Standards ("SFAS") No. 144, "Accounting for the Impairment or Disposal of Long-Lived Assets" ("SFAS No. 144"). If an impairment review is triggered, we evaluate the carrying value of long-lived assets by determining if impairment exists based on estimated undiscounted future cash flows over the remaining useful life of the assets and comparing that value to the carrying value of the assets. If the carrying value of the asset is greater than the estimated future cash flows, the asset is written down to its estimated fair value. In determining expected future cash flows, assets are grouped at the lowest level for which cash flows are identifiable and independent of cash flows from other asset groups. The cash flow estimates that are used contain our best estimates, using appropriate and customary assumptions and projections at the time.

Under SFAS No. 142, "Goodwill and Other Intangible Assets" ("SFAS No. 142"), we are required to perform annual tests of goodwill for impairment or additional tests whenever events or circumstances indicate an impairment may exist. For our annual impairment tests, we identify our reporting units, allocate assets and liabilities (including goodwill) to the reporting units and compare the reporting units' net book value to their estimated fair value. The fair value of the reporting units is estimated using a discounted cash flow approach. The cash flow estimates used contain our best estimates, using appropriate and customary assumptions and projections at the time. If a reporting unit's net book value exceeds its fair value, then the implied fair value of goodwill is determined. If the net book value of goodwill exceeds the implied fair value of goodwill impairment loss is recognized in an amount equal to that excess. No impairment has been identified as a result of the annual reviews.

Warranty Reserves

We provide for the estimated cost of product warranties in cost of product revenue at the time revenue is recognized. Our actual warranty obligation is affected by product failure rates and service costs incurred in correcting a product failure. Should actual product failure rates or service costs differ from our estimates, which are based on historical data, revisions to the estimated warranty liability would be required. At December 31, 2005 and 2004, we had accrued \$3.2 million and \$3.7 million for estimated warranty expense, respectively, including warranty reserves of \$2.5 million and \$3.3 million, respectively, for LaserCyte® Hematology Analyzers. Warranty expense was \$2.2 million, \$3.6 million and \$3.6 million for the years ended December 31, 2005, 2004 and 2003, respectively.

The decrease in the warranty liability during 2005, compared to 2004, was due to the improved reliability of the LaserCyte® Hematology Analyzer, partially offset by the impact of the growing installed base of LaserCyte® Hematology Analyzers. The increase in warranty liability during 2004 compared to 2003 was due to the impact of the growing installed base of LaserCyte® Hematology Analyzers, partially offset by a reduction of warranty cost resulting from our improved service experience for these instruments. We charge warranty expense to the cost of LaserCyte® revenue at the time revenue is recognized on the system based on the estimated cost to repair the instrument over its warranty period. Cost of revenue reflects not only estimated warranty expense for the systems sold in the current period, but also any changes in estimated warranty expense for the installed base that results from our quarterly evaluation of service experience. The reduction in estimated warranty costs per instrument resulted in a reduction of \$0.3 million and \$0.6 million in cost of product revenue for the years ended December 31, 2005 and 2004, respectively.

21

Income Taxes

We account for income taxes under SFAS No. 109, "Accounting for Income Taxes." This statement requires that we recognize a current tax liability or asset for current taxes payable or refundable, respectively; and a deferred tax liability or asset, as the case may be, for the estimated future tax effects of temporary differences between book and tax treatment of assets and liabilities and carryforwards to the extent they are realizable. We record a valuation allowance to reduce our deferred tax assets to the amount that is more likely than not to be realized. While we consider future taxable income and ongoing prudent and feasible tax planning strategies in assessing the need for a valuation allowance, in the event we were to determine that we would be able to realize our deferred tax assets in the future in excess of the net recorded amount, an adjustment to the deferred tax asset would increase income in the period such determination was made. Likewise, should we determine that we would not be able to realize all or part of our net deferred tax asset in the future, an adjustment to the deferred tax asset would be charged to income in the period such determination was made. Significant judgment is required in determining our worldwide provision for income taxes and our income tax filings are regularly under audit by tax authorities. We periodically assess our exposures related to our worldwide provision for income taxes and believe that we have appropriately accrued taxes for contingencies. Any reduction of these contingent liabilities or additional assessment would increase or decrease income, respectively, in the period such determination was made.

We consider the operating earnings of non-United States subsidiaries to be indefinitely invested outside the United States, the cumulative amount of which was \$79.6 million at December 31, 2005. No provision has been made for United States federal and state, or international taxes that may result from future remittances of undistributed earnings of non-

United States subsidiaries. During 2005, we repatriated approximately \$30.0 million under the *American Jobs Creation Act of 2004* and recorded a tax provision of \$1.0 million related to this repatriation. Should we repatriate non-United States earnings in the future, we would have to adjust the income tax provision in the period in which the decision to repatriate earnings is made.

Estimates for Certain Contingencies

Under our workers' compensation insurance policy for U.S. employees for the years ended December 31, 2005, 2004 and 2003, we retain the first \$250,000 in claim liability per incident and \$2.8 million, \$3.0 million and \$1.4 million, respectively, in aggregate claim liability. We entered into a similar workers' compensation insurance policy effective January 1, 2006. We estimate claim liability based on claims incurred and the estimated ultimate cost to settle the claims. Based on this analysis, we have recognized cumulative expenses of \$0.7 million, \$0.6 million and \$0.7 million for claims incurred during the years ended December 31, 2005, 2004 and 2003, respectively.

Under our employee health care insurance policy, we retain claims liability risk up to \$125,000 per incident and an aggregate claim limit based on the number of employees enrolled in the plan per month, which was estimated to be \$13.9 million at December 31, 2005. We estimate our liability for the uninsured portion of employee health care obligations based on individual and aggregate coverage, our claims experience, the number of employees enrolled in the program, and the average time from when a claim is incurred to the time it is reported. Should actual employee health care claims liability exceed estimates, we are liable for up to an additional \$1.7 million for potential uninsured obligations at December 31, 2005. We have insurance coverage of \$1.0 million for claims above the aggregate limit. Should employee health insurance claims exceed this coverage, we would have further obligations for the amount in excess of such coverage.

From time to time, we are notified that a claim is being made against us. We evaluate each claim based on the facts and circumstances of that claim. If warranted, we provide for our best estimate of the cost to settle or litigate the claim and evaluate the liability recorded quarterly.

22

Revenue Recognition

We recognize revenue when four criteria are met. These include (i) persuasive evidence that an arrangement exists, (ii) delivery has occurred or services have been rendered, (iii) the sales price is fixed and determinable, and (iv) collectibility is reasonably assured.

- We recognize revenue at the time of shipment to distributors for substantially all products sold through distributors, as title and risk of loss pass to these customers on delivery to the common carrier. We recognize revenue for the remainder of our customers when the product is delivered, except as noted below. Our distributors do not have the right to return products.
- We recognize revenue from the sales of instruments, noncancelable software licenses and hardware systems upon installation (and completion of training if
 applicable) and the customer's acceptance of the instrument or system because at this time we have no significant further obligations.
- We recognize service revenue at the time the service is performed.
- We recognize revenue associated with extended maintenance agreements ratably over the life of the contracts. Amounts collected in advance of revenue recognition are recorded as a current or long-term liability based on the time from the balance sheet date to the future date of revenue recognition.
- We recognize revenue on certain instrument systems under rental programs over the life of the rental agreement. Amounts collected in advance of revenue recognition are recorded as a current or long-term liability based on the time from the balance sheet date to the future date of revenue recognition.

When instruments are sold together with extended maintenance agreements, we allocate revenue to the extended maintenance agreement under the Emerging Issues Task Force consensus on Issue 00-21, "Accounting for Revenue Arrangements with Multiple Deliverables." Accordingly, the total consideration received is allocated to the elements based on their relative fair values, which is determined by amounts charged separately for the delivered and undelivered elements to other customers. The deferred revenue related to the extended maintenance agreements is recognized ratably over the maintenance period. The delivered elements are recognized as revenue when appropriate under the policies described above. Shipping costs reimbursed by the customer are included in revenue.

We record estimated reductions to revenue in connection with customer programs and incentive offerings, which may give customers credits, award points, or trade-in rights. Awards points may be applied to trade receivables owed to us and/or toward future purchase of our products and services. We estimate these reductions based on our experience with similar customer programs in prior years. Revenue reductions are recorded on a quarterly basis based on issuance of credits, points actually awarded, and estimates of points to be awarded in the future based on current revenue. For the SNAP-Up-the-SavingsTM program, estimates of future points are revised quarterly and finalized annually in the third quarter of each year upon the issuance of points to customers. For our Practice DeveloperTM volume discount program, we have reduced revenue assuming all points granted will result in future credits because we do not have sufficient experience with this program to estimate customer point forfeitures. During 2005, we notified customers that, effective November 30, 2005, unused points awarded prior to January 1, 2004, including points issued under the SNAP-Up-the-SavingsTM program, would be canceled and, that on November 30 of each subsequent year, unused points issued prior to January 1 of the prior year would also be canceled. The value of points canceled in 2005 was less than \$0.1 million.

We may offer customers the right to trade in instruments for credit against the purchase price of other instruments acquired in the future. For trade-in rights, we have reduced revenue using estimates regarding the percentage of qualifying instruments that will be traded in and the average trade-in value.

We recognize revenue only in those situations where collection from the customer is reasonably assured. We maintain allowances for doubtful accounts for estimated losses resulting from the inability of our customers to make required payments. We base our estimates on detailed analysis of specific customer situations and the percentage of our accounts receivable by aging category. If the financial condition of our customers were to deteriorate, resulting in an impairment of their ability to make payments, additional allowances might be required.

23

For the Twelve Months Ended December 31.

Net Revenue (in thousands)	 2005	2004		 Dollar Change	Percentage Change	Percentage Change from Currency (i)	Percentage Change Net of Currency Effect
CAG	\$ 520,830	\$	448,687	\$ 72,143	16.1%	%	16.1%
Water	56,760		53,098	3,662	6.9%	0.3%	6.6%
FDG	60,505		47,396	13,109	27.7%	(0.1%)	27.8%
Total Company	\$ 638,095	\$	549,181	\$ 88,914	16.2%	%	16.2%

¹⁾ Represents the percentage change in revenue attributed to the effect of changes in currency rates from the 12 months ended December 31, 2004 to the 12 months ended December 31, 2005.

Companion Animal Group. Revenue for CAG increased \$72.1 million, or 16%, to \$520.8 million from \$448.7 million for the prior year. Incremental sales from businesses acquired during 2004 and 2005, consisting of veterinary reference laboratories and a digital radiography business, contributed approximately 7% to CAG revenue growth during the period. The following table presents revenue by product and service categories for CAG:

For the Twelve Months Ended December 31,

Net Revenue (in thousands)		2005	2004	Dollar Change	Percentage Change	Percentage Change from Currency (1)	Percentage Change Net of Currency Effect
Instruments and consumables	\$	217,537	\$ 197,939	\$ 19,598	9.9%	%	9.9%
Rapid assay products		100,255	93,506	6,749	7.2%	0.2%	7.0%
Reference laboratory and consulting services		156,425	118,596	37,829	31.9%	(0.3%)	32.2%
Practice information management and digital radiography systems		32,589	28,163	4,426	15.7%	0.2%	15.5%
Pharmaceutical products		14,024	10,483	3,541	33.8%	%	33.8%
Net CAG Revenue	<u> </u>	520,830	\$ 448,687	\$ 72,143	16.1%	%	16.1%

⁽¹⁾ Represents the percentage change in revenue attributed to the effect of changes in currency rates from the 12 months ended December 31, 2004 to the 12 months ended December 31, 2005.

The following revenue analysis reflects the results of operations net of the impact of currency exchange rates on sales outside the U.S.

Because our instrument consumables, rapid assay products, and pharmaceutical products are sold in the U.S. and certain other geographies by distributors, distributor purchasing dynamics have an impact on our reported sales of these products. Distributors purchase products from us and sell them to veterinary practices, who are the end-users. Distributor purchasing dynamics may be affected by many factors and may be unrelated to underlying end-user demand for our products. Fluctuations in distributors' inventories may cause reported results in a period not to be representative of underlying end-user demand. Therefore, we believe it is important to track distributor sales to end-users and to distinguish between the impact of end-user demand and the impact of distributor purchasing dynamics on reported revenue growth.

24

Where growth rates are affected by changes in end-user demand, we refer to the impact of practice-level sales on growth. Where growth rates are affected by distributor purchasing dynamics, we refer to the impact of changes in distributors' inventories. If during the comparable period of the prior year, distributors' inventories grew by more than those inventories grew in the current year, then changes in distributors' inventories have a negative impact on our reported sales in the current period. Conversely, if during the comparable period of the prior year, distributors' inventories grew by less than those inventories grew in the current year, then distributors' inventories have a positive impact on our reported sales in the current period.

The increase in sales of instruments and consumables was due mainly to increased sales volume. The increased sales volume of consumables was due primarily to higher worldwide practice-level sales of VetTest[®] slides. To a lesser extent, increased domestic sales of consumables used with our VetLyte[®] Electrolyte Analyzers and higher practice-level sales of tubes used with our hematology instruments also resulted in increased sales volume of consumables. Increased VetTest[®] chemistry and hematology consumables sales volume was due primarily to an increase in our installed base of instruments throughout 2004 and 2005. The increase in sales of VetLyte[®] consumables was due, in part, to lower sales in the fourth quarter of 2004 due to product unavailability, which had a favorable impact of 1% on the growth rate for instruments and consumables during 2005. Increased instrument sales volume resulted mainly from higher sales of LaserCyte[®] Hematology Analyzers and, to a lesser extent, the launch of our VetStatTM Electrolyte and Blood Gas Analyzer.

The increase in sales of rapid assay products was due primarily to increased domestic practice-level sales volume of our canine combination test, the SNAP® 3Dx® Canine Test, and to higher average unit sales prices for canine and feline products.

The increase in sales of laboratory and consulting services resulted primarily from the inclusion of sales from laboratories acquired in the fourth quarter of 2004 and in 2005 and, to a lesser extent, the impact of price increases and higher testing volume. Incremental sales from laboratories acquired in the fourth quarter of 2004 and in 2005 contributed approximately 23% to laboratory and consulting services revenue growth during 2005.

The increase in sales of practice information management and digital radiography systems resulted from increased sales volume of digital radiography instruments. The increase in digital radiography revenue was primarily due to an increase in the number of systems sold, including sales attributable to a business acquired in the third quarter of 2005. Incremental sales from this acquired business contributed approximately 7% to practice information management and digital radiography systems revenue growth during 2005.

The increase in sales of pharmaceutical products resulted primarily from increased practice-level demand and, to a lesser extent, from price increases on certain products. We expect pharmaceutical revenue for 2006, as a percentage of 2005 revenue, to grow at a lower rate of 15% to 20%.

Water. Revenue for Water increased \$3.7 million, or 7%, to \$56.8 million from \$53.1 million for the prior year. The increase resulted primarily from higher worldwide sales volume, partly offset by lower average unit sales prices attributable to both greater price competition in certain geographies and higher relative sales in geographies where products

are sold at lower unit prices. The favorable impact of currency exchange rates contributed an aggregate of \$0.2 million, or less than 1%, to the increase in Water revenue.

Food Diagnostics Group. Revenue for FDG increased \$13.1 million, or 28%, to \$60.5 million from \$47.4 million for the prior year. Incremental sales from businesses acquired during 2004 contributed approximately 11% to FDG revenue growth during the year. The following table presents revenue by product and service categories for FDG:

For the Twelve Months Ended December 31.

Net Revenue (in thousands)	2005	2004	Dollar Change	Percentage Change	Percentage Change from Currency (1)	Percentage Change Net of Currency Effect
Production animal products	\$ 44,945	\$ 31,690	\$ 13,255	41.8%	(0.3%)	42.1%
Dairy testing products	15,560	15,706	(146)	(0.9%)	0.3%	(1.2%)
	 	 	 			
Net FDG revenue	\$ 60,505	\$ 47,396	\$ 13,109	27.7%	(0.1%)	27.8%

⁽¹⁾ Represents the percentage change in revenue attributed to the effect of changes in currency rates from the 12 months ended December 31, 2004 to the 12 months ended December 31, 2005.

24

The following revenue analysis reflects the results of operations net of the impact of currency exchange rates on sales outside the U.S.

The increase in sales of production animal products resulted primarily from higher worldwide sales volume of livestock and, to a lesser extent, poultry diagnostics, including sales attributable to acquisitions in 2004. Incremental sales from businesses acquired during 2004 contributed approximately 17% to production animal products revenue growth during the period.

The decrease in sales of dairy testing products resulted primarily from the divestiture of the ParalluxTM product line and from lower average unit sales prices attributable to greater price competition in certain geographies and to higher relative sales in geographies where products are sold at lower unit prices. These decreases were partially offset by higher unit sales of SNAP[®] tests.

Gross Profit

Total Company. Gross profit increased \$43.9 million, or 16%, to \$322.9 million from \$279.0 million for the prior year and, as a percentage of total revenue, was approximately constant at 51%. The following table presents gross profit and gross profit percentage by operating segment:

For the Twelve Months Ended December 31,

Gross Profit (in thousands)	 2005	Percent of Sales	 2004	Percent of Sales	 Dollar Change	Percentage Change
CAG	\$ 250,409	48.1%	\$ 214,927	47.9%	\$ 35,482	16.5%
Water	38,277	67.4%	35,885	67.6%	2,392	6.7%
FDG	34,214	56.5%	28,205	59.5%	6,009	21.3%
Total Company	\$ 322,900	50.6%	\$ 279,017	50.8%	\$ 43,883	15.7%

We adopted the provisions of Statement of Financial Accounting Standards No. 123(R), "Share-Based Payment" and will expense share-based compensation beginning on January 1, 2006, which will have a negative impact on our future gross profit percentages and on operating margins for all of our segments.

Companion Animal Group. Gross profit for CAG increased \$35.5 million, or 17%, to \$250.4 million from \$214.9 million for the prior year due primarily to increased sales volume across the CAG product lines. As a percentage of revenue, CAG gross profit was approximately constant at 48%. The gross profit percentage was positively impacted by relatively higher selling prices, particularly for laboratory and consulting services and rapid assay products; lower product and service costs associated with the LaserCyte[®] Hematology Analyzer and lower product cost of slides sold for use in our VetTest[®] Chemistry Analyzers under the agreement with our supplier; and the favorable impact of foreign currency rates on sales denominated in those currencies, net of foreign exchange hedge contract losses. The increases in the gross profit percentage were largely offset by higher overall net product and service costs, apart from the favorable LaserCyte[®] and slide costs mentioned above; greater relative sales of lower margin products and services, mainly from higher sales growth of laboratory services; and write-downs of excess pharmaceutical product inventory.

Water. Gross profit for Water increased \$2.4 million, or 7%, to \$38.3 million from \$35.9 million for the prior year due primarily to increased sales volume, partly offset by a slight decrease in the gross profit percentage to 67% from 68%. The gross profit percentage was unfavorably impacted by costs related to a manufacturing issue during the third quarter of 2005 and by lower average unit sales prices. These decreases in the gross profit percentage were partly offset by the favorable impact of foreign currency rates on sales denominated in those currencies, net of foreign exchange hedge contract losses.

Food Diagnostics Group. Gross profit for FDG increased \$6.0 million, or 21%, to \$34.2 million from \$28.2 million for the prior year due to increased sales volume, partly offset by a decrease in the gross profit percentage to 57% from 60%. During the same period of the prior year, a reduction of approximately \$1.8 million in an estimated liability for a third party claim was accounted for as a reduction in cost of revenue and increased the 2004 gross profit percentage by four percentage points. For 2005, an unfavorable impact on the gross margin percentage of two percentage points was attributable to incremental acquisition integration costs. The gross profit percentage was favorably impacted by higher relative sales of higher margin livestock products and by the favorable impact of foreign currency rates on sales denominated in those currencies, net of foreign exchange hedge contract losses, partially offset by higher net product costs.

Operating Expenses and Operating Income

Operating income increased \$7.3 million to \$115.3 million from \$108.0 million for the prior year. As a percentage of revenue, operating income decreased to 18% from 20%. During 2005, operating income was reduced by acquisition integration costs associated with businesses acquired in the fourth quarter of 2004 and in 2005, including costs incurred in connection with the centralization of our European production animal diagnostics operations in Bern, Switzerland. During 2004, operating income benefited from the settlement of a third party claim, described above, and a payment received in settlement of litigation, partly offset by acquisition integration costs. These discrete items in both years resulted in a reported decrease in operating income as a percentage of total company revenue of one percentage point. The remaining difference in the operating income percentage for 2005, compared to the prior year, was attributable, in part, to the expansion of the CAG sales, customer service and marketing organization during 2004 and the first half of 2005; amortization expense for intangible assets purchased in connection with businesses acquired in the fourth quarter of 2004 and in 2005; and other changes in gross profit and operating expenses described in this narrative.

The following tables present operating expenses and operating income by operating segment:

For the Twelve Months Ended December 31.

Operating Expenses (in thousands)	 2005	Percent of Sales	 2004	Percent of Sales	 Dollar Change	Percentage Change
CAG	\$ 167,439	32.1%	\$ 137,804	30.7%	\$ 29,635	21.5%
Water	12,303	21.7%	11,626	21.9%	677	5.8%
FDG	24,320	40.2%	18,374	38.8%	5,946	32.4%
Other	3,507	N/A	 3,178	N/A	 329	10.3%
Total Company	\$ 207,569	32.5%	\$ 170,982	31.1%	\$ 36,587	21.4%

Operating Income (in thousands)	 2005	Percent of Sales	 2004	Percent of Sales	Dollar Change	Percentage Change
CAG	\$ 82,970	15.9%	\$ 77,123	17.2%	\$ 5,847	7.6%
Water	25,974	45.8%	24,259	45.7%	1,715	7.1%
FDG	9,894	16.4%	9,831	20.7%	63	0.6%
Other	(3,507)	N/A	(3,178)	N/A	(329)	(10.3%)
Total Company	\$ 115,331	18.1%	\$ 108,035	19.7%	\$ 7,296	6.8%

Companion Animal Group. Operating expenses for CAG increased \$29.6 million, or 22%, to \$167.4 million from \$137.8 million for the prior year and, as a percentage of revenue, increased to 32% from 31%. The increase was attributable to a 21% (\$15.1 million) increase in sales and marketing expense, a 25% (\$10.1 million) increase in general and administrative expense, and a 17% (\$4.4 million) increase in research and development expense. The increase in sales and marketing expense resulted primarily from the expansion of the worldwide sales, customer service and marketing organization; ongoing expenses attributable to the Vet Med Lab business acquired in the fourth quarter of 2004 and, to a lesser extent, the digital radiography business acquired in the third quarter of 2005 and higher sales commissions as a result of revenue performance. The increase in general and administrative expense resulted primarily from expenses attributable to businesses acquired in the fourth quarter of 2004 and in 2005, comprised of general and administrative expenses of a recurring nature, amortization expense for intangible assets acquired, and integration costs. To a lesser extent, the increase in general and administrative expense was also attributable to higher spending on information technology and other general support functions; the unfavorable impact of exchange rates on foreign currency denominated expenses; and the positive impact in 2004 of a payment received in the second quarter of 2004 to settle certain litigation. The increase in research and development expense resulted primarily from increased spending related to instrument development and, to a lesser extent, rapid assay and pharmaceutical product development.

27

Water. Operating expenses for Water increased \$0.7 million, or 6%, to \$12.3 million from \$11.6 million for the prior year and, as a percentage of revenue, were approximately constant at 22%. The dollar increase was attributable to a 13% (\$0.6 million) increase in general and administrative expense and a 12% (\$0.2 million) increase in research and development expense, partly offset by a 2% (\$0.1 million) decrease in sales and marketing expense. The increase in general and administrative expense resulted primarily from higher spending on information technology and other corporate functions, and, to a lesser extent, from the unfavorable impact of exchange rates on foreign currency denominated expenses. The increase in research and development expense resulted primarily from increased spending on Cryptosporidium testing product development. There were no significant individual events or fluctuations in the nature and amounts of sales and marketing expense.

Food Diagnostics Group. Operating expenses for FDG increased \$5.9 million, or 32%, to \$24.3 million from \$18.4 million for the prior year and, as a percentage of revenue, increased to 40% from 39%. The increase resulted from a 72% (\$4.0 million) increase in general and administrative expense, a 16% (\$1.3 million) increase in sales and marketing expense, and a 13% (\$0.6 million) increase in research and development expense. The increase in general and administrative expense resulted primarily from expenses associated with the acquisition of Bommeli in the fourth quarter of 2004 and the subsequent centralization of our European production animal diagnostics operations in Bern, Switzerland. These costs are composed mainly of general and administrative expenses of a recurring nature to support the Bommeli business, costs related to the cessation of production in our Sweden-based facility, and amortization expense for intangible assets acquired. To a lesser extent, higher spending on information technology and other corporate functions and the unfavorable impact of exchange rates on foreign currency denominated expenses also contributed to the increase in general and administrative expense. The increase in sales and marketing expense resulted primarily from the addition of Bommeli sales and marketing activities and from sales and marketing costs to support the launch of our HerdChek. BSE Antigen Test Kit. The increase in research and development activities and to higher compensation costs, partly offset by reduced development activity following the launch of our HerdChek. BSE Antigen Test Kit.

Other. Operating expenses, consisting primarily of corporate research and development, increased \$0.3 million, or 10%, to \$3.5 million from \$3.2 million for the prior year due mainly to increased long-term development activities.

Interest Income

Net interest income was \$3.1 million for 2005 and 2004. The impact of higher interest rates was substantially offset by the impact of lower average invested cash balances.

Provision for Income Taxes

Our effective income tax rate was 34.2% for the year ended December 31, 2005 compared with 29.7% for the year ended December 31, 2004. The majority of this rate differential resulted from the favorable impact of the resolution in 2004 of an IRS income tax audit through the year 2001. As a result of completing this audit, we reduced previously accrued taxes. Other rate reductions resulted from the release in 2004 of a valuation allowance on international deferred tax assets as a result of a foreign subsidiary demonstrating consistent sustained profitability and changes in certain state and international tax estimates. In addition, 2005 tax expense increased by \$1.0 million and the 2005 effective income tax rate increased by 0.8 percentage points due to incremental taxes on the repatriation of \$30.0 million pursuant to the *American Jobs Creation Act of 2004*.

Twelve Months Ended December 31, 2004 Compared to Twelve Months Ended December 31, 2003

Revenue

Total Company. Revenue for the total company increased \$73.2 million, or 15%, to \$549.2 million from \$476.0 million in the same period of the prior year. The following table presents revenue for the Company and its operating segments:

For the Twelve Months Ended December 31,

Net Revenue (in thousands)	2004	2003		Dollar Change	Percentage Change	Percentage Change from Currency (1)	Percentage Change Net of Currency Effect
CAG	\$ 448,687	\$	384,419	\$ 64,268	16.7%	2.7%	14.0%
Water	53,098		46,936	6,162	13.1%	4.0%	9.1%
FDG	47,396		44,637	2,759	6.2%	5.4%	0.8%
Total Company	\$ 549,181	\$	475,992	\$ 73,189	15.4%	3.1%	12.3%

⁽¹⁾ Represents the percentage change in revenue attributed to the effect of changes in currency rates from the 12 months ended December 31, 2003 to the 12 months ended December 31, 2004.

Companion Animal Group. Revenue for CAG increased \$64.3 million, or 17%, to \$448.7 million from \$384.4 million in the same period of the prior year. The following table presents revenue by product and service categories for CAG:

For the Twelve Months Ended December 31.

Net Revenue (in thousands)	 2004	 2003		Dollar Change	Percentage Change	Percentage Change from Currency (1)	Percentage Change Net of Currency Effect
Instruments and consumables	\$ 197,939	\$ 177,374	\$	20,565	11.6%	3.9%	7.7%
Rapid assay products	93,506	82,978		10,528	12.7%	1.2%	11.5%
Reference laboratory and consulting services	118,596	94,650		23,946	25.3%	2.9%	22.4%
Practice information management and digital radiography systems	28,163	22,463		5,700	25.4%	0.3%	25.1%
Pharmaceutical products	10,483	6,954		3,529	50.8%	%	50.8%
Net CAG Revenue	\$ 448,687	\$ 384,419	s	64,268	16.7%	2.7%	14.0%

⁽¹⁾ Represents the percentage change in revenue attributed to the effect of changes in currency rates from the 12 months ended December 31, 2003 to the 12 months ended December 31, 2004.

The following revenue analysis reflects the results of operations net of the impact of currency exchange rates on sales outside the U.S.

The increase in sales of instruments and consumables (an increase of \$20.6 million, or 12%) was due mainly to increased sales volume, including higher domestic practice-level sales of VetTest[®] slides and, to a lesser extent, tubes used with our hematology instruments, as well as higher volume outside the U.S.; the favorable impact of currency exchange rates on sales outside the U.S.; increased instrument sales, due primarily to increased sales of the LaserCyte[®] Hematology Analyzer; and the impact of changes in distributors' inventory levels. Increased consumables sales volume was due primarily to an increase in our installed base of instruments during 2003 and 2004. Shipments to distributors during the twelve months ended December 31, 2003 were reduced as a result of the Company's continuing efforts to improve efficiency in the distribution channel. The reduced shipments during 2003 had a positive impact on sales growth in the 2004 period. The collective impact of favorable currency exchange and favorable comparisons resulting from lower distributor purchases in 2003 caused reported growth for 2004 to be higher than our estimates of the underlying practice-level growth of instruments and consumables.

29

The increase in sales of rapid assay products (an increase of \$10.5 million, or 13%) was due primarily to increased domestic practice-level sales volume of canine and, to a lesser extent, feline products, as well as demand for our SNAP® test to screen dogs and cats for Giardia infection, which was launched during the first quarter of 2004; the impact of changes in distributors' inventory levels; and the favorable impact of currency exchange rates on sales outside the U.S. Shipments to distributors during 2003 were reduced as a result of the Company's efforts to improve efficiency in the distribution channel, which contributed to a reported sales growth in the 2004 period. The collective impact of changes in distributor inventory levels and favorable currency exchange caused reported growth for 2004 to be higher than our estimates of the underlying practice-level growth of rapid assay products.

The increase in sales of laboratory and consulting services (an increase of \$23.9 million, or 25%) resulted primarily from higher testing volume at established laboratories, mainly in the U.S. and, to a lesser extent, in the United Kingdom and Australia; the inclusion of sales from laboratories acquired in late 2003 and in 2004; and, to a lesser extent, the favorable impact of currency exchange rates on sales at our laboratories outside the U.S. and higher pricing.

The increase in sales of practice information management and digital radiography systems (an increase of \$5.7 million, or 25%), resulted primarily from higher volume of complete system sales and increased hardware sales and placements of digital radiography systems, partly offset by lower service sales.

The increase in sales of pharmaceutical products (an increase of \$3.5 million, or 51%) resulted in part from sales of new products launched in 2003 and 2004.

Water. Revenue for Water increased \$6.2 million, or 13%, to \$53.1 million from \$46.9 million for the same period of the prior year. The increase resulted primarily from higher sales volume and, to a lesser extent, the favorable impact of currency exchange rates on sales outside the U.S., partly offset by lower average unit prices due to price competition in certain foreign countries and higher relative sales in geographies where products are sold at lower unit prices. The favorable impact of currency exchange rates on sales outside the U.S. contributed an aggregate of \$1.9 million, or 4%, to the increase in Water revenue.

Food Diagnostics Group. Revenue for FDG increased \$2.8 million, or 6%, to \$47.4 million from \$44.6 million for the same period of the prior year. Businesses acquired during 2004 contributed approximately 1% to FDG revenue growth during the year. The following table presents revenue by product and service categories for FDG:

For the Twelves Months Ended December 31,

Net Revenue (in thousands)	2004	2003	Dollar Change	Percentage Change	Percentage Change from Currency (1)	Percentage Change Net of Currency Effect
Production animal products	\$ 31,690	\$ 28,580	\$ 3,110	10.9%	6.1%	4.8%
Dairy testing products	 15,706	16,057	(351)	(2.2%)	4.1%	(6.3%)
Net FDG revenue	\$ 47,396	\$ 44,637	\$ 2,759	6.2%	5.4%	0.8%

¹⁾ Represents the percentage change in revenue attributed to the effect of changes in currency rates from the 12 months ended December 31, 2003 to the 12 months ended December 31, 2004.

The increase was due primarily to the favorable impact of currency exchange rates on sales outside the U.S. and higher sales volume of production animal diagnostics. These increases were partly offset by lower average unit prices of production animal diagnostics and dairy-testing products, and by decreased sales volume of dairy-testing products. The increase in production animal diagnostics sales was due to increased sales volume of livestock products outside the U.S. The lower average unit prices were attributable to greater price competition in certain geographies and, to a lesser extent, to higher relative sales in geographies where products are sold at lower unit prices. The favorable impact of currency exchange rates on sales outside the U.S. contributed an aggregate of \$2.4 million, or 5%, to the increase in FDG revenue.

30

Gross Profit

Total Company. Gross profit for the total company increased \$48.7 million, or 21%, to \$279.0 million from \$230.3 million for the same period in the prior year. As a percentage of total company revenue, gross profit increased to 51% in 2004 from 48% in 2003. The following table presents gross profit and gross profit percentage for the Company and its operating segments:

For the Twelve Months Ended December 31.

Gross Profit (in thousands)	 2004	Percent of Sales	 2003	Percent of Sales	 Dollar Change	Percentage Change
CAG	\$ 214,927	47.9%	\$ 175,612	45.7%	\$ 39,315	22.4%
Water	35,885	67.6%	31,483	67.1%	4,402	14.0%
FDG	28,205	59.5%	23,209	52.0%	4,996	21.5%
Total Company	\$ 279,017	50.8%	\$ 230,304	48.4%	\$ 48,713	21.2%

Companion Animal Group. Gross profit for CAG increased \$39.3 million, or 22%, to \$214.9 million from \$175.6 million in the same period of the prior year due to increased sales volume across the CAG product lines and to an increase in the gross profit percentage. As a percentage of CAG revenue, gross profit increased to 48% from 46% for the same period in the prior year. The increase in gross profit percentage was attributable primarily to productivity improvements across CAG product lines and services; the favorable impact of foreign currency rates on sales denominated in those currencies, net of foreign exchange hedge contract losses; and, to a lesser extent, favorable pricing from our supplier of slide consumables compared to the same period of the prior year. The productivity improvements were partly due to manufacturing efficiencies and reductions in service costs related to our LaserCyte® Hematology Analyzer and, to a lesser extent, to fixed costs spread over a higher revenue base. The LaserCyte® service cost improvements generated a favorable change in our accruals for cost of product warranties and extended maintenance agreements for all placed instruments for which we have such future obligations.

These increases in gross profit percentage were partially offset by a lower gross margin percentage recognized from laboratories acquired in 2004, including due to the purchase accounting impact of writing off supplies and, to a lesser extent, by other laboratory service expansion costs, including start-up costs of laboratories opened in the fourth quarters of 2003 and 2004.

Water. Gross profit for Water increased \$4.4 million, or 14%, to \$35.9 million from \$31.5 million for the same period in the prior year, primarily due to increased revenue. As a percentage of Water revenue, gross profit increased to 68% from 67% for the same period in the prior year. The increase in gross profit percentage was attributable primarily to the favorable impact of foreign currency rates on sales denominated in those currencies, net of foreign exchange hedge contract losses.

Food Diagnostics Group. Gross profit for FDG increased \$5.0 million, or 22%, to \$28.2 million from \$23.2 million for the same period in the prior year, primarily due to an increase in the gross profit percentage. As a percentage of FDG revenue, gross profit increased to 60% from 52% for the same period in the prior year. The increase in gross profit percentage was attributable primarily to reductions in an accrual related to a third-party claim resulting from the settlement of that claim and the favorable impact of foreign currency rates on sales denominated in those currencies, net of foreign exchange hedge contract losses, partly offset by unfavorable product costs. The reduction in the accrual for the third-party claim resulted in an aggregate benefit recognized in 2004 of \$1.8 million or a four-percentage-point increase in the gross margin percentage. The unfavorable product costs were due to fixed costs spread over lower production volume in Europe and, to a lesser extent, the impact of expensing a portion of the purchase accounting fair market value adjustment of inventory obtained in connection with the 2004 acquisitions.

31

Operating Expenses

Total Company. Total company operating expenses increased \$21.1 million to \$171.0 million from \$149.9 million for the same period of the prior year. As a percentage of revenues, operating expenses remained relatively constant at 31%. The following tables present operating expenses and operating income for the Company and its operating segments:

CAG	\$ 137,804	30.7%	\$ 120,396	31.3%	\$	17,408	14.5%
Water	11,626	21.9%	10,549	22.5%		1,077	10.2%
FDG	18,374	38.8%	15,603	35.0%		2,771	17.8%
Other	 3,178	N/A	 3,369	N/A		(191)	(5.7%)
Total Company	\$ 170,982	31.1%	\$ 149,917	31.5%	\$	21,065	14.1%
Operating Income (in thousands)	2004	Percent of Sales	2003	Percent of Sales		Dollar Change	Percentage Change
CAG	\$ 77,123	17.2%	\$ 55,216	14.4%	\$	21,907	39.7%
Vater	24,259	45.7%	20,934	44.6%		3,325	15.9%
FDG	9,831	20.7%	7,606	17.0%		2,225	29.3%
Other	(3,178)	N/A	(3,369)	N/A		191	5.7%
Total Company	\$ 108,035	19.7%	\$ 80,387	16.9%	S	27,648	34.4%

Percent of

2004

Dollar

Change

Percentage

Change

Percent of

2003

Companion Animal Group. Operating expenses for CAG increased \$17.4 million, or 14%, to \$137.8 million from \$120.4 million in the same period of the prior year and were approximately constant at 31% from year to year as a percent of sales. The increase was attributable to a 23% (\$13.4 million) increase in sales and marketing expense, a 10% (\$2.4 million) increase in research and development expense, and a 4% (\$1.6 million) increase in general and administrative expense. The increase in sales and marketing expense resulted primarily from increased sales and sales support personnel and marketing program costs; the unfavorable impact of foreign currency denominated expenses; and, to a lesser extent, expenses associated with the Vet Med Lab acquisition in the fourth quarter of 2004. The increase in research and development expense resulted primarily from increased staffing and higher spending to support instrument and pharmaceutical product development. The increase in general and administrative expense reflects higher spending on information technology and other corporate functions, partly due to expenses associated with Sarbanes-Oxley Act compliance efforts; expenses associated with laboratory acquisitions in the first and fourth quarters of 2004, including amortization of intangible assets; and the unfavorable impact of foreign currency denominated expenses, partly offset by the nonrecurrence in 2004 of expenses incurred in 2003 in connection with the write-down of fixed assets of \$7.4 million associated with the discontinuation of development of a clinical chemistry instrument. In October 2003, we extended our relationship with Ortho, the supplier of our VetTest[®] slides. We committed to develop a next-generation clinical chemistry system based on Ortho's dry-slide technology and discontinued efforts to develop the alternative system.

Water. Operating expenses for Water increased \$1.1 million, or 10%, to \$11.6 million from \$10.5 million in the same period of the prior year and were approximately constant at 22% from year to year as a percent of sales. The increase was attributable to a 25% (\$0.8 million) increase in general and administrative expense, a 7% (\$0.1 million) increase in research and development expense, and a 2% (\$0.1 million) increase in sales and marketing expense. The increase in general and administrative expense reflects higher spending on information technology and other corporate functions, partly due to expenses associated with Sarbanes-Oxley Act compliance efforts; the unfavorable impact of foreign currency denominated expenses; and the impact of a gain from a legal settlement in 2003 that was recorded as a reduction to general and administrative expense. There were no significant fluctuations in the nature and amounts of research and development expense or of sales and marketing expense.

32

Food Diagnostics Group. Operating expenses for FDG increased \$2.8 million, or 18%, to \$18.4 million from \$15.6 million in the same period of the prior year and, as a percent of sales, increased to 39% from 35% in the same period of the prior year. The increase was attributable to a 25% (\$1.1 million) increase in general and administrative expense, a 14% (\$0.6 million) increase in research and development expense, a 5% (\$0.3 million) increase in sales and marketing expense, and a \$0.7 million decrease in other income. The increase in general and administrative expense resulted primarily from ongoing expenses associated with the China joint venture formed in 2003; higher spending on information technology and other corporate functions, partly due to expenses associated with Sarbanes-Oxley Act compliance efforts; the unfavorable impact of foreign currency denominated expenses; and expenses associated with the Bommeli acquisition in the fourth quarter of 2004. The increase in research and development expense was due primarily to higher compensation costs for additional personnel; increased spending in support of our HerdChek® BSE Antigen Test Kit; and integration expenses associated with the Bommeli acquisition in the fourth quarter of 2004. The increase in sales and marketing expense resulted primarily from increased spending in support of our HerdChek® BSE Antigen Test Kit and the unfavorable impact of foreign currency-denominated expenses, partly offset by the nonrecurrence in 2004 of expenses incurred in 2003 in connection with the formation of the China joint venture. The decrease in other income results from the nonrecurrence in 2004 of the reduction in an accrual related to a third-party claim recorded as other income in 2003.

Other. Operating expenses for 2004, consisting primarily of corporate research and development, decreased \$0.2 million, or 6%, to \$3.2 million from \$3.4 million for the same period of the prior year.

Interest Income

Operating Expenses

(in thousands)

Net interest income was \$3.1 million for 2004 compared with \$2.9 million during 2003. The increase in interest income was due to higher average invested cash balances partially offset by lower effective interest rates.

Provision for Income Taxes

Our effective income tax rate was 29.7% for the year ended December 31, 2004, compared with 31.5% for the year ended December 31, 2003. The majority of this rate reduction resulted from the resolution in 2004 of an IRS income tax audit through the year 2001. As a result of completing this audit, the Company reduced previously accrued taxes. Other rate reductions resulted from the release in 2004 of a valuation allowance on international deferred tax assets as a result of a foreign subsidiary demonstrating consistent sustained profitability and changes in certain state and international tax estimates, partially offset by revisions in 2003 to international tax estimates and a charge to write-down fixed assets occurring in a high-tax jurisdiction.

RECENT ACCOUNTING PRONOUNCEMENTS

A discussion of recent accounting pronouncements is included in Note 2(p) to the consolidated financial statements for the year ended December 31, 2005 included in this Form 10-K

LIQUIDITY AND CAPITAL RESOURCES

Liquidity

We fund the capital needs of our business through cash generated from operations. At December 31, 2005 and December 31, 2004, we had \$132.7 million and \$137.3 million of cash and cash equivalents and short-term investments, respectively, and working capital of \$192.7 million and \$201.6 million, respectively. At December 31, 2004, we also had long-term investments, primarily in municipal bonds, of \$19.7 million.

We consider the operating earnings of non-United States subsidiaries to be indefinitely invested outside the U.S. Subject to this policy, we manage our worldwide cash requirements considering available funds among all of our subsidiaries. While the repatriation of foreign earnings could have adverse tax consequences, foreign cash balances are generally available without legal restrictions to fund ordinary business operations.

We believe that current cash and cash equivalents, short-term investments and funds generated from operations will be sufficient to fund our operations, capital purchase requirements, and strategic growth needs.

33

Sources and Uses of Cash

Cash provided by operating activities was \$116.6 million for the year ended December 31, 2005, compared to \$95.4 million for 2004. In 2005, cash increased \$10.7 million due to changes in operating assets and liabilities, whereas in 2004 cash decreased by \$13.6 million due to changes in operating assets and liabilities, resulting in a year-to-year change of \$24.4 million. The increase in cash provided by changes in operating assets and liabilities, compared to 2004, was attributable to incremental changes in cash provided by accrued liabilities of \$15.6 million, accounts payable of \$10.7 million, and inventories of \$6.7 million, partly offset by incremental uses of cash attributable to accounts receivable of \$4.1 million and to other assets and liabilities of \$4.5 million.

The increase in accrued expenses was due to the impact of higher income tax accruals in 2005, compared to 2004, and the impact of higher accruals for points granted to customers under our Practice DeveloperTM volume discount program in 2005, compared to 2004. The higher income tax accruals were due in part to the rate reduction in 2004 resulting from the resolution of an IRS income tax audit through the year 2001, which lowered the accrual in 2004 relative to 2005, and the impact of differences in the timing of expense recognition in the financial statements compared to income tax deductibility. The incremental cash generated from accounts payable was due primarily to the timing of payments to vendors, including contractual purchase commitments to Ortho for VetTest[®] slides. The incremental cash generated from inventory was due to lower inventories of VetTest[®] slides and LaserCyte[®] Hematology Analyzers, partially offset by higher VetAutoreadTM instrument and consumable inventory. The decrease in the VetTest[®] slide inventory was due to the deferral of receipts of slides from Ortho from the fourth quarter of 2005 to the first quarter of 2006. The decrease in cash to fund accounts receivable was due to higher sales. The \$5.9 million increase in depreciation and amortization was due primarily to acquisitions in the fourth quarter of 2004 and in 2005.

Cash provided by investing activities was \$12.6 million for the year ended December 31, 2005, compared to cash used by investing activities of \$36.8 million for 2004. The increase in cash provided by investing activities for 2005, compared to 2004, was primarily due to the reduction in cash used for acquisitions. In 2005, we utilized cash of \$7.6 million to acquire veterinary reference laboratories in Switzerland, the United Kingdom, and France, a veterinary laboratory customer list in the U.S. and a digital radiography business. In 2004, we used \$53.9 million to purchase veterinary reference laboratories in the U.S. and Germany and production animal diagnostic companies in the U.S. and Switzerland. We generated \$44.3 million from net sales of short- and long-term investments for the year ended December 31, 2005, compared to \$48.9 million in 2004.

Since 1999, the Board of Directors has authorized the purchase of up to 16,000,000 shares of our common stock in the open market or in negotiated transactions. At December 31, 2005, we had 2,052,000 shares remaining under our share repurchase authorization. During 2005, we repurchased approximately 1,993,000 shares of our common stock for \$123.8 million at an average price of \$62.11 per share. At December 31, 2005, 2004 and 2003, approximately 13,948,000, 11,955,000, and 9,541,000 cumulative shares, respectively, had been repurchased under this program. During 2004 and 2003, the Company received approximately 1,000 and 133,000 shares of stock, respectively, which were owned by the holders for greater than six months, in payment for the exercise price of stock options. The shares of stock had a fair market value of \$0.1 million and \$4.9 million, respectively. See Note 14 to the consolidated financial statements for the year ended December 31, 2005 included in this Form 10-K.

Commitments

Effective January 1, 2003, we entered into a workers' compensation insurance policy for U.S. employees under which we retain the first \$250,000 in claim liability per incident and up to specific limits, based on payroll, in claim liability in the aggregate. We have entered into similar workers' compensation policies effective January 1, 2004, and 2005. The insurance company administers and pays these claims, and we reimburse the insurance company for our portion of these claims. The insurance company provides insurance for claims above the individual occurrence and aggregate limits. In connection with these policies, we have outstanding letters of credit totaling \$1.6 million to the insurance companies as security for these claims at December 31, 2005. See Note 11 to the consolidated financial statements for the year ended December 31, 2005 included in this Form 10-K.

We purchased \$24.2 million in fixed assets and \$2.6 million in rental instruments sold under recourse during the year ended December 31, 2005, principally related to the CAG segment. Our total capital budget for 2006 for fixed assets and rental instruments is approximately \$37.0 million.

34

We have a 40% equity interest in a joint venture formed to assemble and market veterinary diagnostic products for production animals in China. During the year ended December 31, 2005, we made capital contributions of \$0.6 million to the joint venture. We agreed to purchase an additional 55% equity interest in the joint venture from our partner, subject to approval by the Chinese government of the ownership change, and committed to pay \$0.8 million over two years in consideration for the additional equity. In addition, the joint venture entered into a contract with the joint venture partner where the partner will provide promotional and agency services and will receive sales commissions at rates escalating from 2.5% to 8.5% annually based on sales volume. See Note 17 to the consolidated financial statements for the year ended December 31, 2005 included in this Form 10-K

In January 2006, we entered into an agreement to purchase the building in which our headquarters facility is located for \$18.0 million less the face value of the existing mortgage of \$6.5 million and also agreed to assume the mortgage. The closing is subject to certain conditions to closing, including completion of our due diligence review and receipt of certain state and local incentives. In addition, we expect to incur an estimated additional \$20 million over the next two years primarily to renovate the unoccupied portion of the building for the purpose of expanding our research and development, manufacturing and office space. The purchase of the headquarters facility and subsequent renovation is in addition to the 2006 capital budget.

Under the terms of certain supply agreements with suppliers of our veterinary instruments, slides for our VetTest[®] Chemistry Analyzers; electrolyte instruments, components and consumables; our VetAutoreadTM Hematology Analyzers, components and consumables; and certain raw materials, we have aggregate commitments to purchase approximately \$138.7 million of products through 2020. In addition, we have various minimum royalty payments due through 2019 of \$13.5 million.

We committed up to an aggregate of \$4.0 million of capital purchase obligations in connection with the design and construction of automated production equipment at Ortho's facility that will be used to manufacture consumables for use in our next-generation chemistry analyzer. We expect to pay \$1.9 million of our total commitment in 2006, \$1.2 million in 2007 and the remainder in 2008.

In October 2005, our former supplier of VetAutoreadTM Hematology Analyzers and consumables sold this business (including the human hematology testing products division) and we simultaneously entered into a new supply agreement for veterinary products with QBCD, the acquirer of the business. Under this new supply agreement, we received fixed pricing on certain products through December 31, 2020. In partial consideration for this new supply agreement, we paid cash of \$2.5 million to QBCD and guaranteed QBCD's note in the principal amount of \$3.5 million given to our former supplier in partial consideration for the business. See Note 11 to the consolidated financial statements for the year ended December 31, 2005 included in this Form 10-K.

We are required to make the following payments in the years below:

(in thousands)	Total	2006	2007-2008	2009-2010	After 2010
Minimum royalty payments Operating leases	\$ 13,537 45,890	\$ 1,207 7,608	\$ 3,241 12,348	\$ 3,057 9,231	\$ 6,032 16,703
Unconditional purchase obligations (1)	138,668	61,145	49,192	21,198	7,133
Total contractual cash obligations	\$ 198,095	\$ 69,960	\$ 64,781	\$ 33,486	\$ 29,868

 $^{^{(1)} \}quad \text{Of this amount, \$92.7 million represents our minimum purchase obligation under our VetTest}^{\circledR} \text{ slide supply agreement with Ortho.}$

ITEM 7A. QUANTITATIVE AND QUALITATIVE DISCLOSURE ABOUT MARKET RISK

Our financial market risk consists primarily of foreign currency exchange rate risk. We operate subsidiaries in 15 foreign countries and transact business in local currencies. We attempt to hedge the majority of our cash flow on intercompany sales to minimize foreign currency exposure.

35

The primary purpose of our foreign currency hedging activities is to protect against the volatility associated with foreign currency transactions. Corporate policy prescribes the range of allowable hedging activity. We primarily utilize forward exchange contracts with durations of less than 18 months. Gains and losses related to qualifying hedges of foreign currency from commitments or anticipated transactions are deferred in prepaid expenses or accruals, as appropriate, and are included in the basis of the underlying transaction. Our hedging strategy is consistent with prior periods. Our hedging strategy provides that we employ the full amount of our hedges for the succeeding year at the conclusion of our budgeting process for that year, which is complete by the end of the preceding year. Quarterly, we enter into contracts to hedge incremental portions of anticipated foreign currency transactions for the following twelve months. Accordingly, our risk with respect to foreign currency exchange rate fluctuations may vary throughout each annual cycle. At December 31, 2005, we had \$0.6 million in net unrealized gains on foreign exchange contracts designated as hedges recorded in other comprehensive income, which is net of \$0.3 million in taxes.

Our currency rate exposure at December 31, 2005 consisted of local currency revenues and expenses, the impact of hedge contracts and balances denominated in a currency other than the Company's or its subsidiaries' functional currency. Based on our overall currency rate exposure, excluding unrealized gains of \$0.8 million at December 31, 2005 and unrealized losses of \$4.3 million at December 31, 2004 on foreign exchange contracts designated as hedges, a 10% strengthening of the U.S. dollar relative to foreign currencies would reduce operating income by approximately \$2.6 million for 2006 and a 10% strengthening of the U.S. dollar from December 31, 2004 would have reduced operating income for 2005 by approximately \$3.3 million. A 10% weakening of the U.S. dollar relative to foreign currencies at December 31, 2005 would increase operating income by approximately \$2.6 million in 2006. A 10% weakening of the U.S. dollar from December 31, 2004 would have increased operating income by approximately \$3.3 million in 2005. As of December 31, 2005, a 10% strengthening of the U.S. dollar relative to foreign currencies, excluding the impact of hedge contracts currently in place, would reduce operating income by approximately \$9.7 million in 2006, compared to \$9.9 million in 2005.

ITEM 8. FINANCIAL STATEMENTS AND SUPPLEMENTARY DATA

The response to this item is submitted as a separate section of this report commencing on page F-1.

ITEM 9. CHANGES IN AND DISAGREEMENTS WITH ACCOUNTANTS ON ACCOUNTING AND FINANCIAL DISCLOSURE

Not applicable.

ITEM 9A. CONTROLS AND PROCEDURES

Disclosure Controls and Procedures

Our management is responsible for establishing and maintaining disclosure controls and procedures, as defined by the Securities and Exchange Commission in its Rules 13a-15(e) and 15d-15(e) under the Securities Exchange Act of 1934. Generally, these are controls and procedures designed to ensure that the information required to be disclosed in the reports that we file or submit under the Securities Exchange Act is recorded, processed, summarized, and reported within the time periods specified in the SEC's rules and forms. Our management, with the participation of our chief executive officer and chief financial officer, has concluded that as of the end of the period covered by this report, our disclosure controls and procedures are effective to achieve their stated purpose.

Report of Management on Internal Control Over Financial Reporting

We are responsible for establishing and maintaining adequate internal control over financial reporting as defined in Rules 13a-15(f) and 15d-15(f) under the Securities Exchange Act of 1934. The Company's internal control over financial reporting is designed to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles in the United States of America. Internal control over financial reporting includes those policies and procedures that: (i) pertain to the maintenance of records that, in reasonable detail, accurately and fairly reflect the transactions and dispositions of the assets of the company; (ii) provide reasonable assurance that transactions are recorded as necessary to permit preparation of financial statements in accordance with generally accepted accounting principles in the United States of America, and that receipts and expenditures of the company are being made only in accordance with authorizations of management and directors of the company; and (iii) provide reasonable assurance regarding prevention or timely detection of unauthorized acquisition, use or disposition of the company's assets that could have a material effect on the consolidated financial statements.

deteriorate.

We conducted an evaluation of the effectiveness of internal control over financial reporting based on the framework in Internal Control—Integrated Framework issued by the Committee of Sponsoring Organizations of the Treadway commission. Based on this evaluation, we conclude that, as of December 31, 2005, our internal control over financial reporting was effective.

Our assessment of the effectiveness of our internal control over financial reporting as of December 31, 2005 has been audited by PricewaterhouseCoopers LLP, an independent registered public accounting firm, as indicated in their report that is included herein.

Changes in Internal Control Over Financial Reporting

There were no changes in our internal control over financial reporting during the fourth quarter ended December 31, 2005 that materially affected, or are reasonably likely to materially affect, the Company's internal control over financial reporting.

Certifications

The certifications with respect to disclosure controls and procedures and internal control over financial reporting of the Company's chief executive officer and chief financial officer are attached as Exhibits 31.1 and 31.2 to this Annual Report on Form 10-K.

ITEM 9B. OTHER INFORMATION

Not applicable.

PART III.

ITEM 10. DIRECTORS AND EXECUTIVE OFFICERS OF THE REGISTRANT

The information required by this Item is omitted from this Annual Report on Form 10-K and, pursuant to Regulation 14A of the Securities Exchange Act of 1934, as amended, is incorporated herein by reference from the sections entitled "Corporate Governance" and "Election of Directors" in the Company's definitive proxy statement with respect to its 2006 Annual Meeting of Stockholders, which proxy statement will be filed with the Securities and Exchange Commission within 120 days after the end of the fiscal year covered by this report.

ITEM 11. EXECUTIVE COMPENSATION

The information required by this Item is omitted from this Annual Report on Form 10-K and, pursuant to Regulation 14A of the Securities Exchange Act of 1934, as amended, is incorporated herein by reference from the section entitled "Executive Compensation and Related Information" in the Company's definitive proxy statement with respect to its 2006 Annual Meeting of Stockholders, which proxy statement will be filed with the Securities and Exchange Commission within 120 days after the end of the fiscal year covered by this report.

37

ITEM 12. SECURITY OWNERSHIP OF CERTAIN BENEFICIAL OWNERS AND MANAGEMENT AND RELATED STOCKHOLDER MATTERS

The information required by this Item is omitted from this Annual Report on Form 10-K and, pursuant to Regulation 14A of the Securities Exchange Act of 1934, as amended, is incorporated herein by reference from the section entitled "Ownership of Common Stock by Directors and Officers" in the Company's definitive proxy statement with respect to its 2006 Annual Meeting of Stockholders, which proxy statement will be filed with the Securities and Exchange Commission within 120 days after the end of the fiscal year covered by this report.

ITEM 13. CERTAIN RELATIONSHIPS AND RELATED TRANSACTIONS

The information required by this Item is omitted from this Annual Report on Form 10-K and, pursuant to Regulation 14A of the Securities Exchange Act of 1934, as amended, is incorporated herein by reference from the section entitled "Executive Compensation and Related Information—Employment Agreements" in the Company's definitive proxy statement with respect to its 2006 Annual Meeting of Stockholders, which proxy statement will be filed with the Securities and Exchange Commission within 120 days after the end of the fiscal year covered by this report.

ITEM 14. PRINCIPAL ACCOUNTING FEES AND SERVICES

The information required by this Item is omitted from this Annual Report on Form 10-K and, pursuant to Regulation 14A of the Securities Exchange Act of 1934, as amended, is incorporated herein by reference from the section entitled "Ratification of Appointment of Independent Auditors—Independent Auditors' Fees" in the Company's definitive proxy statement with respect to its 2006 Annual Meeting of Stockholders, which proxy statement will be filed with the Securities and Exchange Commission within 120 days after the end of the fiscal year covered by this report.

PART IV.

ITEM 15. EXHIBITS, FINANCIAL STATEMENT SCHEDULES

(a)

- (1) and (2) The financial statements set forth in the Index to Consolidated Financial Statements and the Consolidated Financial Statement Schedule are filed as a part of this Annual Report on Form 10-K commencing on page F-1.
- (a)(3) and (c) The exhibits in the Exhibit Index immediately preceding the exhibits are filed as part of this Annual Report on Form 10-K and incorporated by reference herein.

Pursuant to the requirements of Section 13 or 15(d) of the Securities Exchange Act of 1934, the Registrant has duly caused this report to be signed on its behalf by the undersigned, thereunto duly authorized:

IDEXX LABORATORIES, INC.

By:/s/Jonathan W. Ayers

Jonathan W. Ayers President and Chief Executive Officer March 10, 2006

Pursuant to the requirements of the Securities Exchange Act of 1934, this report has been signed below by the following persons on behalf of the Registrant and in the capacities and on the dates indicated:

SIGNATURE	TITLE	DATE
<u>/s/Jonathan W. Ayers</u> Jonathan W. Ayers	President, Chief Executive Officer and Chairman of the Board of Directors	March 10, 2006
/s/Merilee Raines Merilee Raines	Vice President, Chief Financial Officer and Treasurer(Principal Financial and Accounting Officer)	March 10, 2006
<u>/s/Thomas Craig</u> Thomas Craig	Director	March 10, 2006
/s/Errol B. De Souza, PhD Errol B. De Souza, PhD	Director	March 10, 2006
/s/William T. End William T. End	Director	March 10, 2006
/s/Rebecca M. Henderson, PhD Rebecca M. Henderson, PhD	Director	March 10, 2006
/s/Brian P. McKeon Brian P. McKeon	Director	March 10, 2006
/s/Robert J. Murray Robert J. Murray	Director	March 10, 2006
	39	

EXHIBIT INDEX

Exhibit No.	<u>Description</u>
3.1	Restated Certificate of Incorporation of the Company, as amended (filed as Exhibit No. 3.1 to Annual Report on Form 10-K for the year ended December 31, 1996, File No. 0-19271, and incorporated herein by reference).
3.2	Amended and Restated By-Laws of the Company (filed as Exhibit No. 3.2 to Quarterly Report on Form 10-Q for the quarter ended September 30, 2000, File No. 0-19271, and incorporated herein by reference).
4.1	Amended and Restated Rights Agreement, dated as of January 22, 2001, between the Company and American Stock Transfer & Trust Company as Rights Agent, which includes as Exhibit A the Form of Certificate of Designations, as Exhibit B the Form of Rights Certificate, and as Exhibit C the Summary of Rights to Purchase Preferred Stock (filed as Exhibit No. 1 to Amendment No. 2 to Registration Statement on Form 8-A/A dated March 14, 2001, File No. 0-19271, and incorporated herein by reference).
4.2	Amendment No. 1 to Amended and Restated Rights Agreement, dated as of March 8, 2005, between the Company and American Stock Transfer & Trust Company as Rights Agent (filed as Exhibit No. 4.1 to Current Report on Form 8-K filed on March 9, 2005, File No. 0-19271, and incorporated herein by reference).
4.3	Instruments with respect to other long-term debt of the Company and its consolidated subsidiaries are omitted pursuant to Item 601(b)(4)(iii) of Regulation S-K since the total amount authorized under each such omitted instrument does not exceed 10 percent of the total assets of the Company and its subsidiaries on a consolidated basis. The Company hereby agrees to furnish a copy of any such instrument to the Securities and Exchange Commission upon request.
10.1†	1991 Stock Option Plan of the Company, as amended (filed as Exhibit No. 10.2 to Annual Report on Form 10-K for the year ended December 31, 2001, File No. 0-19271 ("2001 Form 10-K"), and incorporated herein by reference).
10.2†	1991 Director Option Plan of the Company, as amended (filed as Exhibit No. 10.4 to Quarterly Report on Form 10-Q for the quarter ended June 30, 2001, File No. 0-19271 ("June 2001 10-Q"), and incorporated herein by reference).
10.3†	1997 Director Option Plan of the Company, as amended, with the form of option agreement granted thereunder attached thereto (filed as Exhibit No. 10.1 to Quarterly Report on Form 10-Q for the quarter ended June 30, 1997, File No. 0-19271, and incorporated herein by reference).
10.4†	1999 Director Stock Plan of the Company (filed as Exhibit No. 10.1 to Quarterly Report on Form 10-Q for the quarter ended June 30, 1999, File No. 0-19271, and incorporated herein by reference).

10.5* U.S. Supply Agreement, effective as of October 16, 2003, between the Company and Ortho-Clinical Diagnostics, Inc. ("Ortho") (filed as Exhibit No. 10.7 to Annual Report on Form 10-K for the year ended December 31, 2003, File No. 0-19271 ("2003 Form 10-K"), and incorporated herein by reference). 10.6* Amendment No. 1 to U.S. Supply Agreement effective as of January 1, 2005, between the Company and Ortho (filed as Exhibit No. 10.1 to Quarterly Report on Form 10-Q for the quarter ended June 30, 2005, File No. 0-19271 ("June 2005 10-Q"), and incorporated herein by reference). 10.7* European Supply Agreement, effective as of October 17, 2003, between the Company and Ortho (filed as Exhibit No. 10.8 to 2003 Form 10-K, and incorporated herein by reference). 10.8* Amendment No. 1 to European Supply Agreement effective as of January 1, 2005, between the Company and Ortho (filed as Exhibit No. 10.2 to June 2005 10-Q, and incorporated herein by reference). 10.9† 1998 Stock Incentive Plan of the Company, as amended (filed as Exhibit No. 10.1 to Quarterly Report on Form 10-Q for the quarter ended June 30, 2002, File No. 0-19271, and incorporated herein by reference). 10.10† 2000 Director Option Plan of the Company (filed as Exhibit No. 10.5 to June 2001 10-Q, and incorporated herein by reference). 10.11† Employment Agreement dated January 22, 2002, between the Company and Jonathan W. Ayers (filed as Exhibit No. 10.13 to 2001 Form 10-K, and incorporated herein by reference). 10.12† Executive Employment Agreement dated January 28, 2002, between the Company and Jonathan W. Ayers (filed as Exhibit No. 10.14 to 2001 Form 10-K, and incorporated herein by reference). 10.13† Executive Employment Agreement dated September 8, 2003, between the Company and William C. Wallen (filed as Exhibit No. 10.13 to 2003 Form 10-K, and incorporated herein by reference). 10.14† Letter Agreement dated August 12, 2003, between the Company and William C. Wallen (filed as Exhibit No. 10.14 to 2003 Form 10-K, and incorporated herein by reference). 10.15† Form of Executive Employment Agreement between the Company and each of Robert S. Hulsy, Merilee Raines, Quentin Tonelli, S. Sam Fratoni, Conan R. Deady, Jennifer Joiner, Laurel LaBauve and Ali Naqui (filed as Exhibit No. 10.6 to June 2001 10-Q, and incorporated herein by reference). 10.16 Amendment, Release and Settlement Agreement dated as of September 12, 2002, among the Company, IDEXX Europe B.V., and Ortho (filed as Exhibit No. 10.1 to Quarterly Report on Form 10-Q for the quarter ended September 30, 2002, File No. 0-19271, and incorporated herein by reference). 10.17† Director Deferred Compensation Plan, as amended (filed as Exhibit No. 10.1 to Current Report on Form 8-K filed on February 28, 2006, File No. 0-19271 ("February 28, 2006 Form 8-K"), and incorporated herein by reference). 10.18† 2003 Stock Incentive Plan, as amended (filed as Exhibit No. 10.1 to Quarterly Report on Form 10-Q for the quarter ended September 30, 2005, File No. 0-19271, and incorporated herein by reference). 10.19† Form of Stock Option Agreement, as amended pursuant to the 2003 Stock Incentive Plan (filed as Exhibit No. 10.18 to Annual Report on Form 10-K for the year ended December 31, 2004, File No. 0-19271 ("2004 Form 10-K"), and incorporated herein by reference). 10.20† 1997 Employee Stock Purchase Plan, as amended (filed as Exhibit No. 10.19 to 2004 Form 10-K, and incorporated herein by reference). 10.21† Executive Deferred Compensation Plan, as amended (filed as Exhibit No. 10.2 to February 28, 2006 Form 8-K, and incorporated herein by reference). 10.22† Form of Restricted Stock Unit Agreement (filed herewith). 10.23 Purchase and Sale Agreement dated as of January 17, 2006, between the Company and CW Westbrook Limited Partnership (filed herewith). 21 Subsidiaries of the Company (filed herewith). 23 Consent of PricewaterhouseCoopers LLP (filed herewith). 31.1 Certification by Chief Executive Officer (filed herewith). 31.2 Certification by Vice President, Chief Financial Officer and Treasurer (filed herewith). 32.1 Certification by Chief Executive Officer pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002 (filed herewith). 32.2 Certification by Vice President, Chief Financial Officer and Treasurer pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002 (filed herewith). Confidential treatment requested as to certain portions.

Management contract or compensatory arrangement required to be filed as an exhibit pursuant to Item 15(c) of Form 10-K.

	TAGE
Report of Independent Registered Public Accounting Firm	F-2
Consolidated Balance Sheets as of December 31, 2005 and 2004	F-4
Consolidated Statements of Operations for the Years Ended December 31, 2005, 2004 and 2003	F-5
Consolidated Statements of Stockholders' Equity for the Years Ended December 31, 2005, 2004 and 2003	F-6
Consolidated Statements of Cash Flows for the Years Ended December 31, 2005, 2004 and 2003	F-7
Notes to Consolidated Financial Statements	F-8
Schedule II	
Valuation and Qualifying Accounts	F-34

F-1

REPORT OF INDEPENDENT REGISTERED PUBLIC ACCOUNTING FIRM

To the Board of Directors and Stockholders of IDEXX Laboratories, Inc.:

We have completed integrated audits of IDEXX Laboratories, Inc.'s 2005 and 2004 consolidated financial statements and of its internal control over financial reporting as of December 31, 2005 and an audit of its 2003 consolidated financial statements in accordance with the standards of the Public Company Accounting Oversight Board (United States). Our opinions based on our audits, are presented below.

Consolidated financial statements and financial statement schedule

In our opinion, the consolidated financial statements listed in the index appearing under Item 15(a)(1) present fairly, in all material respects, the financial position of IDEXX Laboratories, Inc. and its subsidiaries at December 31, 2005 and 2004, and the results of their operations and their cash flows for each of the three years in the period ended December 31, 2005 in conformity with accounting principles generally accepted in the United States of America. In addition, in our opinion, the financial statement schedule listed in the index appearing under Item 15(a)(2) presents fairly, in all material respects, the information set forth therein when read in conjunction with the related consolidated financial statements. These financial statements and financial statement schedule are the responsibility of the Company's management. Our responsibility is to express an opinion on these financial statements and financial statement schedule based on our audits. We conducted our audits of these statements in accordance with the standards of the Public Company Accounting Oversight Board (United States). Those standards require that we plan and perform the audit to obtain reasonable assurance about whether the financial statements are free of material misstatement. An audit of financial statements includes examining, on a test basis, evidence supporting the amounts and disclosures in the financial statements, assessing the accounting principles used and significant estimates made by management, and evaluating the overall financial statement presentation. We believe that our audits provide a reasonable basis for our opinion.

Internal control over financial reporting

Also, in our opinion, management's assessment, included in Management's Report on Internal Control Over Financial Reporting appearing under Item 9A, that the Company maintained effective internal control over financial reporting as of December 31, 2005 based on criteria established in Internal Control—Integrated Framework issued by the Committee of Sponsoring Organizations of the Treadway Commission (COSO), is fairly stated, in all material respects, based on those criteria. Furthermore, in our opinion, the Company maintained, in all material respects, effective internal control over financial reporting as of December 31, 2005, based on criteria established in Internal Control—Integrated Framework issued by the COSO. The Company's management is responsible for maintaining effective internal control over financial reporting and for its assessment of the effectiveness of internal control over financial reporting based on our audit. We conducted our audit of internal control over financial reporting in accordance with the standards of the Public Company Accounting Oversight Board (United States). Those standards require that we plan and perform the audit to obtain reasonable assurance about whether effective internal control over financial reporting was maintained in all material respects. An audit of internal control over financial reporting includes obtaining an understanding of internal control over financial reporting, evaluating management's assessment, testing and evaluating the design and operating effectiveness of internal control, and performing such other procedures as we consider necessary in the circumstances. We believe that our audit provides a reasonable basis for our opinions.

F-2

A company's internal control over financial reporting is a process designed to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles. A company's internal control over financial reporting includes those policies and procedures that (i) pertain to the maintenance of records that, in reasonable detail, accurately and fairly reflect the transactions and dispositions of the assets of the company; (ii) provide reasonable assurance that transactions are recorded as necessary to permit preparation of financial statements in accordance with generally accepted accounting principles, and that receipts and expenditures of the company are being made only in accordance with authorizations of management and directors of the company; and (iii) provide reasonable assurance regarding prevention or timely detection of unauthorized acquisition, use, or disposition of the company's assets that could have a material effect on the financial statements.

Because of its inherent limitations, internal control over financial reporting may not prevent or detect misstatements. Also, projections of any evaluation of effectiveness to future periods are subject to the risk that controls may become inadequate because of changes in conditions, or that the degree of compliance with the policies or procedures may deteriorate

/s/PRICEWATERHOUSECOOPERS LLP

Boston, Massachusetts March 10, 2006

CONSOLIDATED BALANCE SHEETS

(in thousands, except per share amounts)

	For the Years End	led December 31,
	2005	2004
ASSETS		
Current Assets:	6 (7.151	47.156
Cash and cash equivalents	\$ 67,151	\$ 47,156
Short-term investments Accounts receivable, less reserves of \$1,221 and \$1,494 in 2005	65,580	90,116
and 2004, respectively	71,688	65,639
Inventories	69,369	76,424
Deferred income taxes	13,778	13,460
Other current assets	11,679	8,797
Total current assets	299,245	301,592
Long-term Investments		19,687
Property and Equipment, at Cost:	1.570	2.216
Land Buildings	1,570 7,457	2,216 5,273
Leasehold improvements	34,645	33,240
Machinery and equipment	58,126	52,564
Office furniture and equipment	35,978	37,000
Construction in progress	5,001	7,558
		125.051
Less accumulated depreciation and amortization	142,777 77,080	137,851 75,221
	65,697	62,630
Other Long-term Assets:		
Goodwill Other intangible assets, net of accumulated amortization of \$9.874	88,127	92,937
and \$6,472 for 2005 and 2004, respectively	30,619	31,557
Other noncurrent assets, net	6,988	5,834
	125,734	130,328
TOTALASSETS	\$ 490,676	\$ 514,237
LLABILITIES AND STRUCKHOLDERS FAULTS		
LIABILITIES AND STOCKHOLDERS' EQUITY Current Liabilities:		
Accounts payable	\$ 19,842	\$ 14,723
Accrued expenses	17,756	20,551
Accrued employee compensation and related expenses	27,550	26,163
Accrued taxes	19,960	15,461
Accrued marketing and customer programs	10,751	8,825
Warranty and extended maintenance agreement reserves	2,191	2,785
Notes payable Deferred revenue	551 7,965	1,291
		10 153
		10,153
Total current liabilities	106,566	99,952
Long-term Liabilities:	106,566	99,952
Long-term Liabilities: Deferred tax liabilities	106,566	99,952
ong-term Liabilities: Deferred tax liabilities Notes payable	6,026	99,952 8,450 519
Ong-term Liabilities: Deferred tax liabilities Notes payable Warranty and extended maintenance agreement reserves	6,026 968	99,952 8,450 519 1,011
Long-term Liabilities: Deferred tax liabilities Notes payable	6,026	99,952 8,450 519
Long-term Liabilities: Deferred tax liabilities Notes payable Warranty and extended maintenance agreement reserves	6,026 968	99,952 8,450 519 1,011
Long-term Liabilities: Deferred tax liabilities Notes payable Warranty and extended maintenance agreement reserves Deferred revenue Total long-term liabilities Commitments and Contingencies (Note 11):	106,566 6,026 968 7,806	8,450 519 1,011 6,253
Long-term Liabilities: Deferred tax liabilities Notes payable Warranty and extended maintenance agreement reserves Deferred revenue Total long-term liabilities Commitments and Contingencies (Note 11): Partner's Interest in Consolidated Subsidiary	106,566 6,026 	8,450 519 1,011 6,253
Long-term Liabilities: Deferred tax liabilities Notes payable Warranty and extended maintenance agreement reserves Deferred revenue Total long-term liabilities Commitments and Contingencies (Note 11): Partner's Interest in Consolidated Subsidiary Stockholders' Equity: Common stock, \$0.10 par value; Authorized: 60,000 shares; Issued: 45,938	106,566 6,026 968 7,806 14,800	99,952 8,450 519 1,011 6,253 16,233
Long-term Liabilities: Deferred tax liabilities Notes payable Warranty and extended maintenance agreement reserves Deferred revenue Total long-term liabilities Commitments and Contingencies (Note 11): Partner's Interest in Consolidated Subsidiary Stockholders' Equity: Common stock, \$0.10 par value; Authorized: 60,000 shares; Issued: 45,938 and 45,217 shares in 2005 and 2004, respectively	106,566 6,026 968 7,806 14,800 300	99,952 8,450 519 1,011 6,253 16,233 392
Long-term Liabilities: Deferred tax liabilities Notes payable Warranty and extended maintenance agreement reserves Deferred revenue Total long-term liabilities Commitments and Contingencies (Note 11): Partner's Interest in Consolidated Subsidiary Stockholders' Equity: Common stock, \$0.10 par value; Authorized: 60,000 shares; Issued: 45,938 and 45,217 shares in 2005 and 2004, respectively Additional paid-in capital	106,566 6,026 968 7,806 14,800	8,450 519 1,011 6,253 16,233
Long-term Liabilities: Deferred tax liabilities Notes payable Warranty and extended maintenance agreement reserves Deferred revenue Total long-term liabilities Commitments and Contingencies (Note 11): Partner's Interest in Consolidated Subsidiary Stockholders' Equity: Common stock, \$0.10 par value; Authorized: 60,000 shares; Issued: 45,938 and 45,217 shares in 2005 and 2004, respectively Additional paid-in capital Deferred share-based compensation; Issued: 25 and 14 units in 2005 and	106,566 6,026 968 7,806 14,800 300 4,594 437,394	99,952 8,450 519 1,011 6,253 16,233 392 4,522 410,817
Deferred tax liabilities: Deferred tax liabilities Notes payable Warranty and extended maintenance agreement reserves Deferred revenue Total long-term liabilities Commitments and Contingencies (Note 11): artner's Interest in Consolidated Subsidiary Stockholders' Equity: Common stock, \$0.10 par value; Authorized: 60,000 shares; Issued: 45,938 and 45,217 shares in 2005 and 2004, respectively Additional paid-in capital Deferred share-based compensation; Issued: 25 and 14 units in 2005 and 2004, respectively	106,566 6,026 968 7,806 14,800 300 4,594 437,394 1,316	99,952 8,450 519 1,011 6,253 16,233 392 4,522 410,817 665
Comp-term Liabilities: Deferred tax liabilities Notes payable Warranty and extended maintenance agreement reserves Deferred revenue Total long-term liabilities Commitments and Contingencies (Note 11): Partner's Interest in Consolidated Subsidiary Stockholders' Equity: Common stock, S0.10 par value; Authorized: 60,000 shares; Issued: 45,938 and 45,217 shares in 2005 and 2004, respectively Additional paid-in capital Deferred share-based compensation; Issued: 25 and 14 units in 2005 and 2004, respectively Retained earnings	106,566 6,026 968 7,806 14,800 300 4,594 437,394 1,316 396,936	99,952 8,450 519 1,011 6,253 16,233 392 4,522 410,817 665 318,682
Deferred tax liabilities: Deferred tax liabilities Notes payable Warranty and extended maintenance agreement reserves Deferred revenue Total long-term liabilities Commitments and Contingencies (Note 11): Partner's Interest in Consolidated Subsidiary Stockholders' Equity: Common stock, \$0.10 par value; Authorized: 60,000 shares; Issued: 45,938 and 45,217 shares in 2005 and 2004, respectively Additional paid-in capital Deferred share-based compensation; Issued: 25 and 14 units in 2005 and 2004, respectively Retained earnings Accumulated other comprehensive income	106,566 6,026 968 7,806 14,800 300 4,594 437,394 1,316	99,952 8,450 519 1,011 6,253 16,233 392 4,522 410,817 665
Long-term Liabilities: Deferred tax liabilities Notes payable Warranty and extended maintenance agreement reserves Deferred revenue Total long-term liabilities Commitments and Contingencies (Note 11): Partner's Interest in Consolidated Subsidiary Stockholders' Equity: Common stock, 50.10 par value; Authorized: 60,000 shares; Issued: 45,938 and 45,217 shares in 2005 and 2004, respectively Additional paid-in capital Deferred share-based compensation; Issued: 25 and 14 units in 2005 and 2004, respectively Retained earnings	106,566 6,026 968 7,806 14,800 300 4,594 437,394 1,316 396,936	99,952 8,450 519 1,011 6,253 16,233 392 4,522 410,817 665 318,682
Long-term Liabilities: Deferred tax liabilities Notes payable Warranty and extended maintenance agreement reserves Deferred revenue Total long-term liabilities Commitments and Contingencies (Note 11): Partner's Interest in Consolidated Subsidiary Stockholders' Equity: Common stock, \$0.10 par value; Authorized: 60,000 shares; Issued: 45,938 and 45,217 shares in 2005 and 2004, respectively Additional paid-in capital Deferred share-based compensation; Issued: 25 and 14 units in 2005 and 2004, respectively Retained earnings Accumulated other comprehensive income Treasury stock (14,118 and 12,125 shares in 2005 and 2004, respectively),	106,566 6,026 968 7,806 14,800 300 4,594 437,394 1,316 396,936 866	99,952 8,450 519 1,011 6,253 16,233 392 4,522 410,817 665 318,682 11,301
Long-term Liabilities: Deferred tax liabilities Notes payable Warranty and extended maintenance agreement reserves Deferred revenue Total long-term liabilities Commitments and Contingencies (Note 11): Partner's Interest in Consolidated Subsidiary Stockholders' Equity: Common stock, \$0.10 par value; Authorized: 60,000 shares; Issued: 45,938 and 45,217 shares in 2005 and 2004, respectively Additional paid-in capital Deferred share-based compensation; Issued: 25 and 14 units in 2005 and 2004, respectively Retained earnings Accumulated other comprehensive income Treasury stock (14,118 and 12,125 shares in 2005 and 2004, respectively), at cost	106,566 6,026 968 7,806 14,800 300 4,594 437,394 1,316 396,936 866 (472,096)	99,952 8,450 519 1,011 6,253 16,233 392 4,522 410,817 665 318,682 11,301 (348,327)

The accompanying notes are an integral part of these consolidated financial statements.

F-4

IDEXX LABORATORIES, INC. AND SUBSIDIARIES

CONSOLIDATED STATEMENTS OF OPERATIONS

(in thousands, except per share amounts)

For the Years Ended December 31,

2005 2004 2003

Revenue:						
Product revenue	\$	460,495	\$	411,748	\$	363,284
Service revenue		177,600		137,433		112,708
		638,095		549,181		475,992
Cost of revenue:						
Cost of product revenue		194,252		174,618		166,382
Cost of service revenue		120,943		95,546		79,306
		315,195	<u> </u>	270,164	<u> </u>	245,688
Gross profit		322,900		279,017		230,304
Gross profit		322,700		275,017		230,304
Expenses:						
Sales and marketing		101,990		85,710		71,846
General and administrative		64,631		49,870		45,752
Research and development		40,948		35,402		32,319
Income from operations		115,331		108,035		80,387
Interest income		3,141		3,068		2,867
interest income		3,141		3,008		2,807
Income before provisions for income taxes and						
partner's interest		118,472		111,103		83,254
Provision for income taxes		40,670		33,165		26,278
Partner's interest in loss of subsidiary		(452)		(394)		(114)
Net income	\$	78,254	\$	78,332	\$	57,090
Earnings per share:						
Basic	S	2.41	\$	2.29	\$	1.67
		2.11	Ψ	2.2)	<u> </u>	1.07
Diluted	\$	2.30	\$	2.19	\$	1.59
Weighted average shares outstanding:						
Basic		32,521		34,214		34,271
Diluted		34,055		35,800		35,931

The accompanying notes are an integral part of these consolidated financial statements.

F-5

IDEXX LABORATORIES, INC. AND SUBSIDIARIES

CONSOLIDATED STATEMENTS OF STOCKHOLDERS' EQUITY

(in thousands, except per share amounts)

Common Stock

	Number of Shares	\$ Par V	0.10 alue	Additional Paid-in Capital	Def Equity-l Compens		Retained Earnings	umulated Other rehensive Income (Loss)	 Treasury Stock	Stock	Total kholders' Equity
Balance January 1, 2003	42,331	\$ 4	233	\$ 334,348	\$		\$ 183,260	\$ (2,511)	\$ (178,357)	\$	340,973
Purchase of treasury stock									(36,195)		(36,195)
Exercise of stock options (including tax benefit)	1,934		193	48,914					(4,897)		44,210
Exercise of warrants	125		13	(13)					(1,077)		
Issuance of deferred stock units	125			(13)		138					138
Comprehensive income (loss):						150					150
Net income							57,090				
Unrealized loss on investments, net							37,090				
of tax of \$86								(131)			
Unrealized loss on forward exchange contracts,											
net of tax of \$523								(1,334)			
Translation adjustment								8,541			
Total comprehensive income								·			64,166
Balance December 31, 2003	44,390	4	439	383,249		138	240,350	4,565	(219,449)		413,292
Purchase of treasury stock	-								(128,814)		(128,814)
Exercise of stock options (including tax benefit)	827		83	27,568					(64)		27,587
Issuance of deferred stock units						527					527
Comprehensive income (loss):						321		-	-		321
Net income							78,332				
Unrealized loss on investments, net of	_						70,332				
tax of \$57								(89)			
Unrealized gain on forward exchange contracts, net								(4.7)			
of tax of \$24								178			
Translation adjustment								6,647			
Total comprehensive income											85,068
Balance December 31, 2004	45,217	4	522	410,817		665	318,682	11,301	(348,327)		397,660
Purchase of treasury stock									(123,769)		(123,769)
Exercise of stock options (including tax benefit)	721		72	26,577							26,649
Issuance of deferred stock											

units				651				651
Comprehensive income (loss):								
Net income					78,254			
Unrealized gain on								
investments, net of								
tax of \$15						23		
Unrealized gain on								
forward exchange								
contracts, net								
of tax of \$1,703						3,403		
Translation adjustment						(13,861)		
Total comprehensive income								67,819
Balance December 31, 2005	45,938	\$ 4,594	\$ 437,394	\$ 1,316	\$ 396,936	\$ 866	\$ (472,096)	\$ 369,010

The accompanying notes are an integral part of these consolidated financial statements.

F-6

IDEXX LABORATORIES, INC. AND SUBSIDIARIES

CONSOLIDATED STATEMENTS OF CASH FLOWS

(in thousands)

For the Years Ended December 31,

		ror the rears	Enucu December	31,	
	200	5	2004		2003
Cash Flows from Operating Activities:					
Net income	\$ 78,25	1 \$	78,332	\$	57,090
Adjustments to reconcile net income to net cash provided by operating activities:					
Depreciation and amortization	24,36)	18,427		18,897
Write-down of fixed assets	24,50	-	10,427		7,359
Partner's interest in loss of subsidiary	(45	2)	(394)		(114)
Provision for (recovery of) uncollectible accounts	12		(294)		114
Provision for (benefit of) deferred income taxes	(4,47		4,599		(1,192)
Tax benefit on exercise of nonqualified stock options and					
disqualifying dispositions	7,80		8,211		13,045
Provision for deferred share-based compensation	18	1	135		138
Changes in assets and liabilities, net of acquisitions and					
disposals					
Accounts receivable	(9,30		(5,162)		(5,567)
Inventories	7,43		758		71
Other assets	(2,24		(26)		1,062
Accounts payable	4,90		(5,791)		9,560
Accrued liabilities	10,18		(5,442)		16,552
Deferred revenue	(22))	2,026		140
					
Net cash provided by operating activities	116,55	2	95,379		117,155
Cash Flows from Investing Activities:					
Purchase of short- and long-term investments	(63,61	9)	(37,114)		(130,802)
Sales and maturities of short- and long-term investments	107,88		86,010		64,990
Purchase of property and equipment	(24,19		(29,065)		(16,896)
Net proceeds from sale of land and buildings	2,75				(10,000)
Acquisition of equipment leased to customers	(2,61		(2,640)		(2,724)
Acquisition(s) of intangible assets and business(es), net of cash acquired	(7,60	4)	(53,942)		(2,300)
Net cash provided by (used in) investing activities	12,59	1	(36,751)		(87,732)
Cash Flows from Financing Activities:				-	
Payment of notes payable	(2,05	7)	(356)		(510)
Purchase of treasury stock	(123,76		(129,191)		(35,817)
Proceeds from the exercise of stock options	18,84		19,376		31,165
Net cash used in financing activities	(106,98	5)	(110,171)		(5,162)
Net effect of exchange rates on cash	(2,16		1,757		3,168
Net increase (decrease) in cash and cash equivalents	19,99	5	(49,786)		27,429
Cash and cash equivalents at beginning of year	47,15	5	96,942		69,513
		-		-	
Cash and cash equivalents at end of year	\$ 67,15	\$	47,156	\$	96,942
Supplemental Disclosure of Cash Flow Information:					
Interest paid	\$ 4	\$	33	\$	16
Income taxes paid	\$ 34,34	\$	25,862	\$	4,938
Supplemental Disclosure of Non-Cash Information:					
Value of mature shares exchanged in stock option exercises	\$ -	- \$	64	\$	4,897
	<u> </u>				
Payable for treasury stock	\$	- \$		\$	378
Receivable for purchase price adjustment of		Φ		Ψ	576
business acquisitions	\$ 2	2 \$	500	\$	
Notes payable issued as consideration in acquisitions	\$	- \$	1,000	\$	

The accompanying notes are an integral part of these consolidated financial statements.

IDEXX LABORATORIES, INC. AND SUBSIDIARIES

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

NOTE 1 NATURE OF BUSINESS

We develop, manufacture and distribute products and provide services for the veterinary and the food- and water-testing markets. We operate primarily through three business segments: products and services for the veterinary market, which is referred to as the Companion Animal Group ("CAG"), water quality products ("Water") and products for production animal health and dairy quality, which is referred to as the Food Diagnostics Group ("FDG"). Our products and services are sold worldwide. See Note 18 for additional information regarding our reportable operating segments, products and services, and geographical areas.

NOTE 2 SUMMARY OF SIGNIFICANT ACCOUNTING POLICIES

(a) Consolidation

The accompanying consolidated financial statements include our accounts, our wholly-owned subsidiaries, and all other entities in which we have a variable interest and are determined to be the primary beneficiary. All material intercompany transactions and balances have been eliminated in consolidation.

(b) Estimates

The preparation of these financial statements in accordance with accounting principles generally accepted in the United States of America requires us to make estimates and judgments that affect the reported amounts of assets, liabilities, revenues and expenses, and related disclosure of contingent assets and liabilities. On an ongoing basis, we evaluate these estimates, including those related to customer programs and incentives, product returns, bad debts, inventory, investments, goodwill and other intangible assets, income taxes, warranty reserves, and contingencies. We accrue contingent liabilities when it is probable that future expenditures will be made and such expenditures can reasonably be estimated. We base our estimates on historical experience and on various other assumptions that we believe to be reasonable under the circumstances, the results of which form the basis for making judgments about the carrying values of assets and liabilities that are not readily apparent from other sources. Actual results may differ from these estimates.

(c) Inventories

Inventories include material, labor and overhead, and are stated at the lower of cost (first-in, first-out) or market. We write down inventory for estimated obsolescence when warranted by estimates of future demand and market conditions. If actual market conditions are less favorable than those we estimated, additional inventory write-downs may be required, which would have a negative effect on results of operations. Certain major components of inventory are discussed in more detail below.

LaserCyte[®] Hematology Analyzer. At December 31, 2005 and 2004, our net inventories included \$9.8 million and \$11.9 million, respectively, of component parts and finished goods associated with the LaserCyte[®] Hematology Analyzer. In addition, we have firm purchase commitments for an additional \$2.6 million of component parts at December 31, 2005. At December 31, 2005 and 2004, \$2.3 million and \$1.9 million of the net LaserCyte[®] inventory, respectively, required rework before it could be used to manufacture finished goods. At December 31, 2005 and 2004, the inventory subject to rework was net of \$0.7 million and \$0.3 million write-downs, respectively, for inventory estimated to be obsolete. We expect to fully realize our investment in inventory and purchase commitments. However, if we alter the design of this product, we may be required to write off some or all of the remaining associated inventory.

F-8

VetTest® Chemistry Slides. At December 31, 2005 and 2004, our net inventories included \$14.4 million and \$22.7 million, respectively, of slides used in our VetTest® Chemistry Analyzers. The decrease in slide inventory at December 31, 2005, compared to December 31, 2004, was primarily due to the delay of inventory receipts from the fourth quarter of 2005 to the first quarter of 2006. Most of the slides have a shelf-life of 24 months at the date of manufacture. The average remaining shelf-life at December 31, 2005 was 16.4 months. In addition, we are required to purchase a minimum of \$92.7 million of slides from Ortho-Clinical Diagnostics, Inc. ("Ortho") through December 31, 2010. During the quarter ended December 31, 2003, we entered into a new contract with Ortho, which extended the term of the supply agreement through 2018 and left the contract minimum purchase commitments unchanged. In June 2005, we further amended this agreement to, among other things, extend its term from 2018 to 2020. As a result of the current and projected demand for VetTest® slides, our commitment to develop a next-generation chemistry analyzer that will utilize these slides, and the ratable decrease in required annual slide purchases from Ortho through 2010, we believe that we will not incur a loss under the contract. See Note 11 for additional discussion of our development commitment.

Nitazoxanide. Our nitazoxanide product, Navigator[®], for the treatment of equine protozoal myeloencephalitis ("EPM") was approved by the U.S. Food and Drug Administration ("FDA") in November 2003. At December 31, 2005, our inventories included \$9.4 million of inventory associated with Navigator[®], consisting of \$0.2 million of finished goods and \$9.2 million of active ingredient and other raw materials. In December 2004, we entered into an amendment to our agreement with our supplier of nitazoxanide under which we paid the supplier \$0.9 million in January 2005 and the supplier agreed until 2017 to replace any expiring inventory of nitazoxanide with longer-dated material. The payment was capitalized as inventory cost. We believe that this agreement has substantially mitigated the risk that we would be required to write down nitazoxanide inventory due to its anticipated expiration prior to sale.

The components of inventories are as follows (in thousands):

		December 31,				
		2005		2004		
Raw materials	\$	22,517	\$	20,847 10,363 45,214		
Work-in-process		10,583		10,363		
Finished goods		36,269		45,214		
•	·					
	\$	69,369	\$	76,424		

(d) Property and Equipment

We record property and equipment at cost net of accumulated depreciation and amortization. When an item is sold or retired, the cost and related accumulated depreciation is relieved, and the resulting gain or loss, if any, is recognized in the statement of operations. We provide for depreciation and amortization using the declining-balance and straight-line methods by charges to operations in amounts that allocate the cost of property and equipment over their estimated useful lives as follows:

Asset Classification Estimated Useful Life

Leasehold improvements	Shorter of life of lease or useful life
Machinery and equipment	3-5 years
Office furniture and equipment	3-7 years
Buildings	40 years

We recorded depreciation expense of \$17.8 million, \$14.7 million and \$14.5 million for the years ended December 31, 2005, 2004 and 2003, respectively.

(e) Goodwill and Other Intangible Assets

Intangible assets, other than goodwill, are valued at fair value when acquired. If a market value is not readily available, the fair value of the intangible asset is estimated based on expected cash flows of the associated business acquired that are attributable to the intangible asset. Goodwill is initially valued based on the excess of the purchase price of a business combination over the other net assets acquired.

F-9

We provide for amortization using the straight-line and accelerated methods by charges to operations in amounts that allocate the intangible assets over their estimated useful lives as follows:

Asset Classification	Estimated Useful Life
Detects and consolited to be desired.	15
Patents and completed technology	15 years
Noncompete agreements	2-10 years
Contractual relationships	15 years
Customer lists	5 years
Customer relationships	8-15 years
Licenses	5-10 years
Other	5-10 years

We assess the impairment of identifiable intangible assets and other long-lived assets whenever events or changes in circumstances indicate that the carrying value may not be recoverable. Factors we consider important that could trigger an impairment review include, but are not limited to, the following:

- Significant under-performance relative to historical or projected future operating results;
- Failures to obtain regulatory approval of certain products;
- Significant changes in the manner of our use of the acquired assets or the strategy for our overall business;
- Significant increase in the discount rate assumed to calculate the present value of future cash flow;
- Significant negative industry or economic trends;
- · Significant advancements or changes in technology; and
- Cancellation or significant changes in contractual relationships.

We continually assess the realizability of intangible assets other than goodwill in accordance with Statement of Financial Accounting Standards ("SFAS") No. 144, "Accounting for the Impairment or Disposal of Long-Lived Assets" ("SFAS No. 144"). If an impairment review is triggered, we evaluate the carrying value of long-lived assets by determining if impairment exists based on estimated undiscounted future cash flows over the remaining useful life of the assets and comparing that value to the carrying value of the assets. If the carrying value of the asset is greater than the estimated future cash flows, the asset is written down to its estimated fair value. In determining expected future cash flows, assets are grouped at the lowest level for which cash flows are identifiable and independent of cash flows from other asset groups. The cash flow estimates that are used contain our best estimates, using appropriate and customary assumptions and projections at the time.

Under SFAS No. 142, "Goodwill and Other Intangible Assets" ("SFAS No. 142"), we are required to perform annual tests of goodwill for impairment or additional tests whenever events or circumstances indicate an impairment may exist. For our annual impairment tests, we identify our reporting units, allocate assets and liabilities (including goodwill) to the reporting units and compare the reporting units' net book value to their estimated fair value. The fair value of the reporting units is estimated using a discounted cash flow approach. The cash flow estimates used contain our best estimates, using appropriate and customary assumptions and projections at the time. If a reporting unit's net book value exceeds its fair value, then the implied fair value of goodwill is determined. If the net book value of goodwill exceeds the implied fair value of goodwill impairment loss is recognized in an amount equal to that excess. No impairment has been identified as a result of the annual reviews.

F-10

(f) Warranty and Extended Maintenance Agreement Reserves

We provide for the estimated cost of product warranties in cost of product revenue at the time revenue is recognized. Our actual warranty obligation is affected by product failure rates and service costs incurred in correcting a product failure. Should actual product failure rates or service costs differ from our estimates, which are based on historical data, revisions to the estimated warranty liability would be required. Following is a summary of changes in accrued warranty reserve for products sold to customers for the years ended December 31, 2005 and 2004, respectively (in thousands):

	For the Years Ended December 31				
	 2005		2004		
Balance, beginning of year Provision for warranty expense Provision for change in estimate of prior warranty expense Settlement of warranty liability	\$ 3,679 2,479 (276) (2,723)	\$	3,303 4,196 (612) (3,208)		
Balance, end of year Long-term portion	 3,159 968		3,679 910		

Current portion of warranty reserves \$ 2,191 \$ 2,769

The decrease in the warranty liability during 2005, compared to 2004, was due to the improved reliability of the LaserCyte® Hematology Analyzer, partially offset by the impact of the growing installed base of LaserCyte® Hematology Analyzers. We charge warranty expense to the cost of LaserCyte® revenue at the time revenue is recognized on the system based on the estimated cost to repair the instrument over its warranty period. Cost of revenue reflects not only estimated warranty expense for the systems sold in the current period, but also any changes in estimated warranty expense for the installed base that results from our quarterly evaluation of service experience.

We sell extended maintenance agreements covering our instruments and recognize associated revenue over the life of the contracts. At December 31, 2004, we anticipated that \$0.1 million in losses would be incurred for certain of these contracts and recognized a provision for the estimated loss. No loss on extended maintenance agreements is anticipated as of December 31, 2005.

(g) Income Taxes

We account for income taxes under SFAS No. 109, "Accounting for Income Taxes." This statement requires that we recognize a current tax liability or asset for current taxes payable or refundable, respectively; and a deferred tax liability or asset, as the case may be, for the estimated future tax effects of temporary differences between book and tax treatment of assets and liabilities and carryforwards to the extent they are realizable. We record a valuation allowance to reduce our deferred tax assets to the amount that is more likely than not to be realized. While we consider future taxable income and ongoing prudent and feasible tax planning strategies in assessing the need for a valuation allowance, in the event we were to determine that we would be able to realize our deferred tax assets in the future in excess of the net recorded amount, an adjustment to the deferred tax asset would increase income in the period such determination was made. Likewise, should we determine that we would not be able to realize all or part of our net deferred tax asset in the future, an adjustment to the deferred tax asset would be charged to income in the period such determination was made. Significant judgment is required in determining our worldwide provision for income taxes and our income tax filings are regularly under audit by tax authorities. We periodically assess our exposures related to our worldwide provision for income taxes and believe that we have appropriately accrued taxes for contingencies. Any reduction of these contingent liabilities or additional assessment would increase or decrease income, respectively, in the period such determination was made. See Note 9 for additional information regarding income taxes.

(h) Revenue Recognition

We recognize revenue when four criteria are met: (i) persuasive evidence that an arrangement exists, (ii) delivery has occurred or services have been rendered, (iii) the sales price is fixed and determinable, and (iv) collectibility is reasonably assured.

• We recognize revenue at the time of shipment to distributors for substantially all products sold through distributors as title and risk of loss pass to these customers on delivery to the common carrier. We recognize revenue for the remainder of our customers when the product is delivered to the customer except as noted below. Our distributors do not have the right to return products.

F-11

- We recognize revenue from the sales of instruments, noncancelable software licenses and hardware systems upon installation (and completion of training if
 applicable) and the customer's acceptance of the instrument or system because at this time we have no significant further obligations.
- We recognize service revenue at the time the service is performed.
- We recognize revenue associated with extended maintenance agreements over the life of the contracts. Amounts collected in advance of revenue recognition are
 recorded as a current or long-term liability based on the time from the balance sheet date to the future date of revenue recognition.
- We recognize revenue on certain instrument systems under rental programs over the life of the rental agreement. Amounts collected in advance of revenue recognition are recorded as a current or long-term liability based on the time from the balance sheet date to the future date of revenue recognition.

When instruments are sold together with extended maintenance agreements, we allocate revenue to the extended maintenance agreement under the Emerging Issues Task Force ("EITF") consensus on Issue 00-21, "Accounting for Revenue Arrangements with Multiple Deliverables." Accordingly, the total consideration received is allocated to the elements based on their relative fair values, which is determined by amounts charged separately for the delivered and undelivered elements to other customers. The deferred revenue related to the extended maintenance agreements is recognized ratably over the maintenance period. The delivered elements are recognized as revenue when appropriate under the policies described above. Shipping costs reimbursed by the customer are included in revenue.

We record estimated reductions to revenue in connection with customer programs and incentive offerings, which may give customers credits, award points, or trade-in rights. Awards points may be applied to trade receivables owed to us and/or toward future purchase of our products and services. We estimate these reductions based on our experience with similar customer programs in prior years. Revenue reductions are recorded on a quarterly basis based on issuance of credits, points actually awarded, and estimates of points to be awarded in the future based on current revenue. For the SNAP-Up-the-SavingsTM program, estimates of future points are revised quarterly and finalized annually in the third quarter of each year upon the issuance of points to customers. For our Practice DeveloperTM volume discount program, we have reduced revenue assuming all points granted will result in future credits because we do not have sufficient experience with this program to estimate customer point forfeitures. During 2005, we notified customers that, effective November 30, 2005, unused points awarded prior to January 1, 2004, including points issued under the SNAP-Up-the-SavingsTM program, would be canceled and, that on November 30 of each subsequent year, unused points issued prior to January 1 of the prior year would also be canceled. The value of points canceled in 2005 was less than \$0.1 million.

We may offer customers the right to trade in instruments for credit against the purchase price of other instruments acquired in the future. For trade-in rights, we have reduced revenue using estimates regarding the percentage of qualifying instruments that will be traded in and the average trade-in value.

We recognize revenue only in those situations where collection from the customer is reasonably assured. We maintain allowances for doubtful accounts for estimated losses resulting from the inability of our customers to make required payments. We base our estimates on detailed analysis of specific customer situations and a percentage of our accounts receivable by aging category. If the financial condition of our customers were to deteriorate, resulting in an impairment of their ability to make payment, additional allowances might be required.

(i) Research and Development and Software Development Costs

Research and Development costs are expensed as incurred. In accordance with SFAS No. 86, "Accounting for the Costs of Computer Software to be Sold, Leased or Otherwise Marketed" ("SFAS No. 86"), we evaluate our software research and development costs for capitalization after the technological feasibility of software and products containing

(j) Advertising and Promotion Costs

We expense advertising costs to sales and marketing expense in the period they are incurred.

(k) Share-Based Compensation

Prior to January 1, 2006, we measured costs related to employee share-based compensation plans in accordance with Accounting Principles Board ("APB") Opinion No. 25, "Accounting for Stock Issued to Employees" ("APB No. 25"), and elected to disclose the pro forma impact of accounting for share-based compensation plans under the provisions of SFAS No. 123, "Accounting for Stock-Based Compensation" and SFAS No. 148, "Accounting for Stock-Based Compensation—Transition and Disclosure—An Amendment of FASB No. 123" (collectively, "SFAS No. 123, as Amended"). Accordingly, no employee compensation cost has been recognized for these plans based on SFAS No. 123, as Amended

Had compensation cost for our share-based compensation and employee stock purchase plans been determined consistent with the provisions of SFAS No. 123, as Amended, our net income and net income per common and common equivalent share would have been reduced to the following pro forma amounts (in thousands, except per share amounts):

	 For the Years Ended December 31,						
	 2005		2004		2003		
Net income:							
As reported	\$ 78,254	\$	78,332	\$	57,090		
APB No. 25 compensation recorded, net of tax			-				
Pro forma share-based employee compensation, net of tax	(8,701)		(7,975)		(7,999)		
Pro forma net income	\$ 69,553	\$	70,357	\$	49,091		
	 		·	_			
Earnings per share:							
Basic: as reported	\$ 2.41	\$	2.29	\$	1.67		
Basic: pro forma	2.14		2.06		1.43		
Diluted: as reported	2.30		2.19		1.59		
Diluted: pro forma	2.05		1.97		1.37		

See Note 15 for discussion of our share-based compensation plans.

(l) Foreign Currency Translation

Assets and liabilities of our foreign subsidiaries are translated using the exchange rate in effect at the balance sheet date. Revenue and expense accounts are translated using a weighted average of exchange rates in effect during the period. Cumulative translation gains and losses are shown in the accompanying consolidated balance sheets as a separate component of accumulated other comprehensive income (loss). Exchange gains and losses arising from transactions denominated in foreign currencies other than our subsidiary's functional currency are included in current operations. Included in general and administrative expenses are aggregate foreign exchange currency transaction losses of \$0.8 million, and gains of \$0.4 million and \$1.0 million for the years ended December 31, 2005, 2004 and 2003, respectively. Additionally, for the year ended December 31, 2005, a cumulative translation loss of \$0.5 million was transferred from accumulated other comprehensive income and included in general and administrative expenses as a result of the closure of our Sweden-based operation and the associated centralization of our European production animal diagnostics operations products manufacturing in Switzerland.

(m) Derivative Instruments and Hedging

We follow SFAS No. 133, "Accounting for Derivative Instruments and Hedging Activities" as amended by SFAS No. 137, "Accounting for Derivative Instruments and Hedging Activities—Deferral of the Effective Date of SFAS No. 133" and SFAS No. 138, "Accounting for Certain Derivative Instruments and Hedging Activities—An Amendment of SFAS No. 133" ("SFAS No. 133, as Amended"). SFAS No. 133, as Amended requires that all derivatives, including forward currency exchange contracts, be recognized on the balance sheet at fair value. Derivatives that are not hedges must be recorded at fair value through earnings. If a derivative is a hedge, depending on the nature of the hedge, changes in the fair value of the derivative are either offset against the change in fair value of the hedged item through earnings or recognized in other comprehensive income until the hedged item is recognized in earnings. We immediately record in earnings the extent to which a hedge is not effective in achieving offsetting changes in fair value.

F_13

Our subsidiaries enter into foreign currency exchange contracts of their anticipated intercompany inventory purchases for the next twelve months in order to minimize the impact of foreign currency fluctuations on these transactions. Our accounting policies for these contracts are based on our designation of such instruments as hedging transactions. We also utilize some natural hedges to mitigate our transaction and commitment exposures. The contracts we enter into are firm foreign currency commitments, and, therefore, market gains and losses are deferred until the contract matures, which is the period when the related obligation is settled. We enter into these exchange contracts with large multinational financial institutions. We do not hold or engage in transactions involving derivative instruments for purposes other than risk management. We hedge less than the full value of forecasted intercompany sales and thus no significant ineffectiveness has resulted or been recorded through the statement of operations. At December 31, 2005, we recorded \$0.8 million in unrealized gains through accumulated other comprehensive income from foreign exchange contracts with 2006 expiration dates. At December 31, 2004, we recorded \$4.3 million in unrealized losses through accumulated other comprehensive loss from foreign exchange contracts with 2005 expiration dates. The foreign currency contracts, which extend through December 31, 2006 and 2005, respectively, consisted of the following (in thousands):

II S Dollar Fauivalent

Currency Solu		U.S. Donai Equivaient			
		2005		2004	
Euro	\$	44,511	\$	35,000	
British Pound		18,046		17,360	
Canadian Dollar		11,825		11,082	
Swiss Franc		7,664			
Australian Dollar		2,756		1,800	
Japanese Yen		2,644		575	
•		· · · · · · · · · · · · · · · · · · ·			
	\$	87,446	\$	65,817	

Gains and losses on foreign exchange contracts intended as hedges for intercompany sales of goods are recorded in cost of sales. Included in cost of goods sold are foreign exchange losses of \$0.1 million, \$5.2 million and \$6.7 million for the years ended December 31, 2005, 2004 and 2003, respectively.

(n) Disclosure of Fair Value of Financial Instruments and Concentration of Risk

Financial instruments consist mainly of cash and cash equivalents, investments, accounts receivable, accounts payable and notes payable. Financial instruments that potentially subject us to concentrations of credit risk are principally cash, cash equivalents, investments and accounts receivable. We place our investments in highly rated financial institutions and investment grade money market funds, municipal bonds and preferred stock. Concentration of credit risk with respect to accounts receivable is limited to certain customers to whom we make substantial sales. To reduce risk, we routinely assess the financial strength of our customers and, as a consequence, believe that our accounts receivable credit risk exposure is limited. We maintain an allowance for potential credit losses, but historically have not experienced any significant credit losses related to an individual customer or group of customers in any particular industry or geographic area. The carrying amounts of our financial instruments approximate fair market value. See Note 18 for further discussion of concentration of credit risk of accounts receivable.

We currently purchase certain products and materials from single sources or a limited number of sources. Some of the products that we purchase from these sources are proprietary, and, therefore, may not be available from other sources. If we are unable to obtain adequate quantities of these products in the future, we could face cost increases or reductions or delays in product shipments, which could have a material adverse effect on our results of operations.

F-14

(o) Comprehensive Income

SFAS No. 130, "Reporting Comprehensive Income," requires us to report all changes in equity during a period, resulting from net income and transactions or other events and circumstances from non-owner sources, in a financial statement for the period in which they are recognized. We have chosen to disclose comprehensive income, which encompasses net income, foreign currency translation adjustments and the difference between the cost and the fair market value of investments in debt securities and foreign exchange contracts, in the Consolidated Statement of Stockholders' Equity. We consider the foreign currency cumulative translation adjustment to be permanently invested and, therefore, have not provided income taxes on those amounts.

Accumulated other comprehensive income (loss) consists of the following at December 31, 2005 and 2004, respectively, (in thousands):

		December 31,			
		2005		2004	
Unrealized loss on investments, net of tax Unrealized gain (loss) on forward exchange	s	(46)	\$	(69)	
contracts, net of tax		553		(2,850)	
Cumulative translation adjustment		359		(2,850) 14,220	
	\$	866	\$	11,301	

(p) Recent Accounting Pronouncements

In October 2004, the *American Jobs Creation Act of 2004* (the "Jobs Creation Act") was signed into law. The Jobs Creation Act allows for a reduced rate of United States tax on qualifying repatriations of earnings held outside the United States in either the last tax year that began before the enactment date or the first tax year that began during the one-year period beginning on the date of enactment. In December 2004, the Financial Accounting Standards Board ("FASB") issued Staff Position 109-2, "Accounting and Disclosure Guidance for the Foreign Earnings Repatriation Provision within the American Jobs Creation Act of 2004" ("FSP 109-2"). FSP 109-2 allows time beyond the financial reporting period of enactment to evaluate the effect of the Jobs Creation Act on a company's plan for reinvestment or repatriation of foreign earnings. FSP 109-2 was effective upon its issuance. Accordingly, we recognized the tax impact of foreign earnings repatriated in December 2005 when we decided on our repatriation plan during the fourth quarter of 2005.

In December 2004, the FASB issued Statement of Financial Accounting Standards ("SFAS") No. 123(R), "Share-Based Payment" ("SFAS No. 123(R)"), which is a revision of SFAS No. 123, "Accounting for Stock-Based Compensation" and SFAS No. 148, "Accounting for Stock-Based Compensation—Transition and Disclosure—An Amendment of FASB No. 123" (collectively, "SFAS No. 123, as Amended") and supersedes Accounting Principles Board ("APB") Opinion No. 25, "Accounting for Stock Issued to Employees". During 2005, the FASB also issued Staff Positions No. FAS 123(R)-1, 2, and 3 to provide application guidance related to SFAS No. 123(R). SFAS No. 123(R) requires all sharebased compensation to employees, including grants of employee stock options, to be valued at fair value on the date of grant, and to be expensed over the applicable vesting period. Pro forma disclosure of the income statement effects of share-based payments is no longer an alternative. Implementation of SFAS No. 123(R) is required as of the beginning of the first annual period that begins after June 15, 2005. Compensation expense related to any awards that are not fully vested must also be recognized as of the effective date. Compensation expense for the unvested awards will be measured based on the fair value of the awards previously calculated in developing the pro forma disclosures in accordance with the provisions of SFAS No. 123, as Amended. We adopted the provisions of SFAS No. 123(R) on January 1, 2006 and do not plan to adjust financial statements for prior periods. In 2006, we modified our share-based compensation programs to shift from the grant of stock options only to the grant of a mix of restricted stock and stock options. Also in connection with the adoption of SFAS 123(R), we adopted the straight-line method to prospectively expense future share-based grants. Previously, we utilized and we will continue to utilize the graded method to expense stock option grants prior to January 1, 2006. We expect our 2006 share-based compensation expense, as a percentage of net income excluding share-based compensation expense, to be slightly less than the pro forma expense disclosed in accordance with SFAS No. 123, as Amended, above in the financial statements for prior periods. The total compensation cost for unvested awards granted prior to January 1, 2006, before future forfeitures, that will be recognized in the years ending December 31, 2006 through December 31, 2010 is \$16.6 million. Approximately half of this expense will be recognized in 2006 and decreasing amounts of the total expense will be recognized over the subsequent four years, resulting in a weighted average expense period of approximately 1.7 years. We do not expect the adoption of SFAS No. 123(R) to have a material impact on our cash flows or financial position.

F-15

In March 2005, the FASB issued FASB Interpretation ("FIN") 47, "Accounting for Conditional Asset Retirement Obligations" ("FIN 47"). FIN 47 clarifies the timing of liability recognition for conditional asset retirement obligations in accordance with SFAS No. 143, "Accounting for Asset Retirement Obligations". The provisions of FIN 47 are effective no later than the end of fiscal years ending after December 15, 2005. The adoption of FIN 47 did not have a material impact on our results of operations or financial position for the year ended December 31, 2005.

In May 2005, the FASB issued SFAS No. 154, "Accounting Changes and Error Corrections—a replacement of APB Opinion No. 20 and FASB Statement No. 3" ("SFAS No. 154"). SFAS No. 154 replaces APB Opinion No. 20, "Accounting Changes" and SFAS No. 3 "Reporting Accounting Changes in Interim Financial Statements" and changes the requirements for the accounting for and reporting of a change in accounting principle. SFAS No. 154 applies to all voluntary changes in accounting principle. It also applies to

changes required by an accounting pronouncement in the unusual instance that the pronouncement does not include specific transition provisions. When a pronouncement includes specific transition provisions, those provisions should be followed. SFAS No. 154 requires retrospective application to prior periods' financial statements of changes in accounting principle. The provisions of SFAS No. 154 are effective for fiscal years beginning after December 15, 2005. We do not expect the adoption of SFAS No. 154 to have a material impact on our results of operations or financial position.

In June 2005, the FASB ratified the EITF consensus on Issue 05-6, "Determining the Amortization Period for Leasehold Improvements Purchased After Lease Inception or Acquired in a Business Combination" ("EITF 05-6"). EITF 05-6 requires that leasehold improvements acquired in a business combination or placed in service significantly after the beginning of the lease term be amortized over the shorter of the useful life of the assets or the lease term, including renewal periods that are reasonably assured. EITF 05-6 is effective for leasehold improvements acquired in reporting periods beginning after June 29, 2005. The adoption of EITF 05-6 did not have a material impact on our results of operations or financial position for the year ended December 31, 2005.

In November 2005, the FASB ratified the EITF consensus on Issue 04-10, "Determining Whether to Aggregate Operating Segments That Do Not Meet the Quantitative Thresholds" ("EITF 04-10"). EITF 04-10 provides implementation guidance regarding the aggregation of operating segments referenced by SFAS No. 131, "Disclosures about Segments of an Enterprise and Related Information" ("SFAS No. 131"). EITF 04-10 concludes that operating segments that do not meet the quantitative thresholds described by SFAS No. 131 can be aggregated only if aggregation is consistent with the objective and basic principles of SFAS No. 131, the segments have similar economic characteristics, and the segments share a majority of the aggregation criteria listed in SFAS No. 131. EITF 04-10 is effective for fiscal years ending after September 15, 2005. The adoption of EITF 04-10 did not have an impact on our determination of our reportable operating segments and the related disclosures.

NOTE 3 BUSINESS ACQUISITIONS

In February 2004, we acquired certain assets and assumed certain liabilities of a veterinary reference laboratory located in Ohio. We paid cash of \$5.3 million, issued a note for \$1.0 million and assumed liabilities of \$0.5 million, for a total purchase price of \$6.8 million. Goodwill and amortizable intangible assets of \$1.9 million and \$3.9 million, respectively, were assigned to the Companion Animal Group segment.

In August 2004, we paid cash of \$1.5 million to acquire all of the shares of a production animal diagnostics company located in New York. Amortizable intangible assets of \$2.2 million were assigned to the Food Diagnostics Group segment.

In November 2004, we acquired all of the shares of the Institut für klinische Prüfung Ludwigsburg GmbH, which conducted business under the name Vet Med Lab ("Vet Med Lab"). Vet Med Lab and subsidiaries comprise veterinary reference laboratories in Germany and Switzerland, and additional customer service locations in the Netherlands, the United Kingdom, France, Italy, Austria and Denmark. We paid cash, including acquisition costs and net of cash acquired, of \$31.0 million and assumed liabilities of \$10.7 million for a total purchase price of \$41.7 million. Goodwill and amortizable intangible assets of \$26.1 million and \$11.2 million, respectively, were assigned to the Companion Animal Group segment.

F-16

In December 2004, we acquired all of the shares of Dr. Bommeli AG, a production animal diagnostics company based in Switzerland. We paid cash, including acquisition costs and net of cash acquired, of \$15.8 million, and assumed liabilities of \$3.1 million for a total purchase price of \$18.9 million. Goodwill and amortizable intangible assets of \$7.4 million and \$8.8 million, respectively, were assigned to the Food Diagnostics Group.

In 2005, we paid cash of \$5.5 million and assumed liabilities of \$0.7 million to acquire certain assets of veterinary reference laboratories in Switzerland, the UK, Germany and France and customer lists in the U.S. and Germany. Goodwill and other intangible assets of \$2.1 million and \$2.8 million, respectively, were assigned to the Companion Animal Group segment.

In September 2005, we paid cash of \$2.0 million and assumed liabilities of \$1.3 million to acquire the business of a Georgia-based veterinary-specific digital radiography systems company. Intangible assets of \$2.5 million were assigned to the Companion Animal Group segment. We also agreed to make additional purchase price payments of up to \$2.3 million, contingent on the achievement by the acquired business of certain milestones.

The final purchase price allocation of certain businesses acquired in 2005 is subject to finalization of the valuation of certain assets and liabilities.

The results of operations of the acquired businesses have been included with our results since the respective acquisition dates. Pro forma information has not been presented because such information is not material to our financial statements taken as a whole.

NOTE 4 CASH EQUIVALENTS, SHORT-TERM AND LONG-TERM INVESTMENTS

Cash equivalents are highly liquid investments purchased with original maturities of less than three months.

We account for investments under SFAS No. 115, "Accounting for Certain Investments in Debt and Equity Securities" as available-for-sale. Investments are recorded at amortized cost and adjusted to fair market value through other comprehensive income. Gains on sales of investments were not significant for the years ended December 31, 2005, 2004 and 2003. Short-term investments, which had cost bases of \$65.7 million and \$90.2 million at December 31, 2005 and 2004, respectively, are investment securities with maturities of greater than three months, but less than one year, and consist of the following (in thousands):

		December 31,				
	•		2005		2004	
	-					
Municipal bonds	\$	\$	19,263	\$	49,716 40,400	
Municipal auction rate securities			44,600		40,400	
Canadian certificates of deposit			1,717		·	
	-					
	\$	\$	65,580	\$	90,116	

At December 31, 2005 and 2004, we held \$65.6 million and \$90.1 million, respectively, of short-term investments, which included \$44.6 million and \$40.4 million, respectively, of auction rate municipal securities classified as available-for-sale securities. Our investments in these securities are recorded at cost, which approximates fair market value due to their variable interest rates, which typically reset every 28 to 35 days, and, despite the long-term nature of their stated contractual maturities, we have the ability to quickly liquidate these securities. As a result, we had no cumulative gross unrealized holding gains (losses) or gross realized gains (losses) from these short-term investments. All income generated from these short-term investments was recorded as interest income.

We include auction rate municipal bonds in short-term investments. Previously, such investments had been classified as cash and cash equivalents. We adjusted our Consolidated Statements of Cash Flows for the period ended December 31, 2003 to reflect the gross purchases and sales of these securities as investing activities rather than as a

Long-term investments, which had a cost basis of \$19.8 million at December 31, 2004 are investment securities with maturities of greater than one year and less than five years and consist of municipal bonds. We held no long-term investments at December 31, 2005.

NOTE 5 OTHER NONCURRENT ASSETS, INTANGIBLE ASSETS AND GOODWILL

Other noncurrent assets are as follows (in thousands):

	December 31,					
Description	 2005		2004			
Deferred tax asset	\$ 420	\$	239			
Cost of rental instruments sold under recourse, net Other assets	4,115 2,453		4,127 1,468			
	\$ 6,988	<u> </u>	5,834			

Rental instruments sold under recourse are amortized over their useful life of three years.

Intangible assets other than goodwill consist of the following (in thousands):

		December 31, 2005			December 31, 2004			
		Cost		Accumulated Amortization		Cost		Accumulated Amortization
Existing technologies	S	9,168	S	3,193	S	10,309	\$	1,987
Licenses	*	3,800		1,757	-	3,800	*	1,409
Customer relationships		15,111		1,413		14,249		362
Customer lists		703		465		638		377
Noncompete agreements		3,207		1,003		2,686		591
Patents		5,810		1,934		6,211		1,744
Contractual relationships and other		2,694		109		136		2
	\$	40,493	\$	9,874	\$	38,029	\$	6,472

Amortization expense of intangible assets was \$3.9 million, \$1.6 million and \$0.5 million for the years ended December 31, 2005, 2004 and 2003, respectively. During the year ended December 31, 2005, we acquired \$5.3 million of amortizable intangible assets related to acquisitions in the CAG segment, with a weighted average amortization period of 13.8 years. During the year ended December 31, 2004, we acquired \$15.2 million of amortizable intangible assets related to acquisitions in the CAG segment, with a weighted average amortization period of 10.6 years, and \$11.0 million of amortizable intangible assets related to acquisitions in the FDG segment, with a weighted average amortization period of 14.7 years.

Amortization expense of intangible assets for each of the next five years is expected to be as follows (in thousands):

		Amortization Expense
2006 2007 2008 2009 2010	\$	4,338
2007		3,849
2008		3,849 3,488
2009		3,118
2010		2,765

F-18

Goodwill consists of the following (in thousands):

	December 31,		
	 2005		2004
Companion Animal Group Segment:		<u> </u>	
Veterinary reference laboratories	\$ 51,311	\$	53,088
Pharmaceuticals	13,745		13,745
Practice information management and digital			
radiography systems	1,453		1,453
Other goodwill	113		119
Water Segment:			
Water testing products	15,184		16,885
Food Diagnostics Group Segment: Production animal diagnostics			
Production animal diagnostics	6,321		7,647
	\$ 88,127	\$	92,937

During the year ended December 31, 2005, we acquired \$2.1 million of goodwill (of which \$1.3 million is expected to be tax deductible) related to acquisitions in the CAG segment. During the year ended December 31, 2004, we acquired \$28.0 million of goodwill (of which \$2.0 million is expected to be tax deductible) related to acquisitions in the

CAG segment and \$7.1 million of goodwill (which is not expected to be tax deductible) related to an acquisition in the FDG segment. The remaining change in goodwill noted above is a result of changes in foreign currency exchange rates.

NOTE 6 IMPAIRMENT OF LONG-LIVED ASSETS

During the fourth quarter of 2003, we entered into a new agreement with Ortho. Under this agreement, we are developing and will introduce a next-generation chemistry analyzer for the veterinary market based on Ortho's dry-slide technology, and Ortho will supply us with the slide consumables used in both the new instrument and the VetTest[®] Chemistry Analyzer currently sold by us. As a result of this agreement, we decided to discontinue efforts to develop an alternative chemistry system and incurred a noncash charge of \$7.4 million to write down equipment purchased to manufacture the consumable used in the alternative chemistry system, which was included in general and administrative expenses in the consolidated statement of operations.

NOTE 7 NOTES PAYABLE

In connection with the acquisition of a water testing products business in August 2000, we issued \$8.5 million in notes payable to a former shareholder of Genera, of which \$7.0 million was collateralized by cash in escrow. The remaining \$1.5 million was unsecured and noninterest bearing, and was discounted at 6% to a fair value of \$1.3 million. In April 2002, we repaid \$7.5 million, of which \$7.0 million was paid from the cash held in escrow. The remaining unsecured portion of \$1.0 million was due in three annual installments, beginning in August 2002. The note holder elected to defer the August 2002 payment of \$0.5 million until April 2003. The note holder elected to defer the August 2004 payment of \$0.25 million until February 2005. The interest rate on the deferred notes was 3%.

In connection with the February 2004 acquisition of a veterinary reference laboratory described in Note 3, we issued a note payable to the sellers for \$1.0 million. The note bears interest at the prime rate. The balance outstanding at December 31, 2005 and 2004 was \$0.6 million and \$0.9 million, respectively. We paid \$0.4 million in February 2005 and the remaining \$0.5 million, plus accrued interest, in March 2006.

In connection with the November 2004 acquisition of Vet Med Lab described in Note 3, we assumed a note payable to a bank of \$0.6 million. The interest rate on the note was 2.9%. The balance outstanding at December 31, 2004 was \$0.6 million. We paid the note in full during the first quarter of 2005.

NOTE 8 EXIT ACTIVITY

During the year ended December 31, 2005, we centralized our European production animal diagnostics manufacturing operations in Bern, Switzerland, the location of the production animal diagnostics company acquired in December 2004. In connection with this centralization, we ceased operations in Sweden. We recognized expenses of \$1.0 million associated with this exit activity during the year ended December 31, 2005 and do not expect to incur any future costs. The total costs include a cumulative translation adjustment write-off of \$0.5 million, one-time employee termination benefits of \$0.2 million, and building lease termination costs of \$0.1 million, which are included in general and administrative expenses in the consolidated statement of operations. The total costs also include one-time employee termination benefits of \$0.2 million that are included in costs of product revenue. These expenses are attributable to the Food Diagnostics Group segment. At December 31, 2005, accrued expenses include building lease termination costs of \$0.1 million which were subsequently paid in January 2006.

F-19

NOTE 9 INCOME TAXES

Earnings before income taxes for each year were as follows (in thousands):

	 2005		2004		2003
Domestic International	\$ 85,401 33,523	\$	78,605 32,892	\$	58,582 24,786
	\$ 118,924	S	111,497	s	83,368

The provisions for income taxes for the years ended December 31, 2005, 2004 and 2003 are comprised of the following (in thousands):

		For the Years Ended December 31,						
		2005		2004	_	2003		
Current								
Federal	S	30,070	\$	19,438	S	18,122		
State		4,680		2,628		3,811		
International		10,397		6,500		3,811 5,537		
		45,147		28,566	·	27,470		
Deferred								
Federal		(3,020)		5,328		(978) (170)		
State		(277)		556		(170)		
International		(1,180)		(1,285)		(44)		
		(4,477)		4,599		(1,192)		
	\$	40,670	s	33,165	\$	26,278		

The provision for income taxes differs from the amount computed by applying the statutory federal income tax rate as follows:

	December 31,			
	 2005	2004	2003	
U.S. federal statutory rate	35.0%	35.0%	35.0%	
State income tax, net of federal tax benefit	2.4	1.9	2.8	
International income taxes	(1.9)	(3.5)	(3.7)	
Extraterritorial income exclusions	(0.6)	(0.7)	(0.7)	
Nontaxable interest income	(0.6)	(0.6)	(0.8)	

Domestic manufacturing exclusions Tax on dividend repatriations Other, net	(0.5) 0.5 (0.1)	 (2.4)	(1.1)
Effective tax rate	34.2%	29.7%	31.5%

Our effective tax rate was 34.2% for the year ended December 31, 2005, compared with 29.7% for the year ended December 31, 2004. The majority of this rate differential resulted from the favorable impact of several rate reducing items occurring in 2004 (that are described below). In addition, 2005 tax expense increased by \$1.0 million and the 2005 effective income tax rate increased by 0.8 percentage points due to incremental taxes on repatriation of \$30.0 million pursuant to the *American Jobs Creation Act of 2004*.

Our effective rate was 29.7% for the year ended December 31, 2004, compared with 31.5% for the year ended December 31, 2003. The reduction in the effective tax rate from 2004 compared to 2003 primarily resulted from the resolution in 2004 of an IRS income tax audit through the year 2001. As a result of completing this audit, we reduced previously accrued taxes. Other rate reductions resulted from the release in 2004 of a valuation allowance on international deferred tax assets as a result of a foreign subsidiary demonstrating consistent sustained profitability and changes in certain state and international tax estimates, partially offset by revisions in 2003 to international tax estimates and a charge to write down fixed assets occurring in a high-tax jurisdiction.

F-20

The components of the net deferred tax asset (liability) included in the accompanying consolidated balance sheets are as follows (in thousands):

	2005			2004			
	 Current		Long-Term		Current		Long-Term
Assets:							
Accrued expenses	\$ 8,887	\$		\$	8,655	\$	
Accounts receivable reserves	363				284		
Deferred revenue	2,198		2,453		1,900		2,549
Inventory basis differences	2,612				1,419		
Intangible asset basis differences			5				332
Property-based differences			366		-		295
Net operating loss carryforwards	58		4,424		43		4,605
Unrealized losses on foreign exchange							
contracts and investments	29				1,463		
	 			-			
Total assets	14,147		7,248		13,764		7,781
Total assets	14,147		7,240		15,704		7,701
Valuation allowance	(369)		(4,527)		(304)		(4,639)
	 			-			
Total assets, net of valuation allowance	13,778		2,721		13,460		3,142
rotal assets, net of valuation anowance	13,776		2,721		15,400		3,142
Liabilities:							
Cost of rental instruments sold under recourse			(1,158)				(1,133)
Property-based differences			(410)				(2,804)
Intangible basis differences			(6,758)				(7,566)
Unrealized gains on foreign exchange contracts	(307)						
Other							(84)
		-		-			
Total liabilities	(307)		(8,326)				(11,587)
	 (==7)		(*,*=*)				(,,-)
N. 10 1. (1111)	 	·	(5.605)		12.460		(0.445)
Net deferred tax assets (liabilities)	\$ 13,471		(5,605)	\$	13,460	\$	(8,445)

At December 31, 2005, we had United States federal domestic net operating loss carryforwards of approximately \$0.2 million available to offset future taxable income. Net operating loss carryforwards expire at various dates through 2014. The Tax Reform Act of 1986 contains provisions that limit annual availability of the net operating loss carryforwards due to a more than 50% change in ownership that occurred upon the acquisition of some companies.

At December 31, 2005, we had net operating loss carryforwards in foreign and state jurisdictions of approximately \$62.6 million available to offset future taxable income. Most of these net operating loss carryforwards expire at various dates through 2022 and the remainder have indefinite lives. We have recorded a valuation allowance for these assets because realizability is uncertain.

We consider the operating earnings of non-United States subsidiaries to be indefinitely invested outside the United States. No provision has been made for United States federal and state, or international taxes that may result from future remittances of undistributed earnings of non-United States subsidiaries, the cumulative amount of which is \$79.6 million at December 31, 2005. During the year ended December 31, 2005, we repatriated approximately \$30.0 million under the American Jobs Creation Act of 2004. Should we repatriate non-United States earnings in the future, we would have to adjust the income tax provision in the period in which the decision to repatriate earnings is made.

NOTE 10 EARNINGS PER SHARE

Basic earnings per share is computed by dividing net income by the weighted average number of shares of common stock and deferred stock units outstanding during the year. The computation of diluted earnings per share is similar to the computation of basic earnings per share, except that the denominator is increased for the assumed exercise of dilutive options and other potentially dilutive securities using the treasury stock method unless the effect is anti-dilutive.

F-21

The following is a reconciliation of shares outstanding for basic and diluted earnings per share (in thousands):

2005	2004	2003
32,498	34,203	34,270
23	11	1
32,521	34,214	34,271
32,521	34,214	34,271
	32,498 23 32,521	32,498 34,203 23 11 32,521 34,214

Dilutive effect of options issued to employees and directors Dilutive effect of warrants	1,534	1,586	1,613
	34,055	35,800	35,931

Deferred stock units outstanding are included in shares outstanding for basic and diluted earnings per share because the associated shares of our common stock are issuable for no cash consideration, the number of shares of our common stock to be issued is fixed and issuance is not contingent. See Note 15 for additional information regarding deferred compensation plans.

In connection with our acquisition of the capital stock of Blue Ridge Pharmaceuticals, Inc. ("Blue Ridge") in 1998, we issued warrants to acquire 806,000 shares of common stock at \$31.59 per share that expired on September 30, 2003. As of December 31, 2003, all of the warrants were exercised or had expired.

Certain options to acquire shares have been excluded from the calculation of shares outstanding for dilutive earnings per share because they were anti-dilutive. Information about anti-dilutive options and the weighted average market value of shares used to calculate the dilutive effect of options were as follows (in thousands, except per share amounts):

	For the Years Ended December 31				
	 2005	2004		2003	
Weighted average number of shares underlying anti-dilutive options		24		37	
Weighted average exercise price per underlying share of anti-dilutive options	\$ 	\$ 60.70	\$	42.60	
Weighted average market value per share	\$ 61.57	\$ 53.83	\$	39.35	

The following is additional information concerning the exercise prices of vested and unvested options outstanding (in thousands, except per share amounts):

	2005	2004
Closing price per share of our common stock	\$ 71.98	\$ 54.59
Number of shares underlying options outstanding with exercise prices below the closing price Number of shares underlying options outstanding with exercise prices equal to	 3,747	3,861
or above the closing price	 <u>-</u>	 37
Total number of shares underlying outstanding options	 3,747	 3,898

F-22

NOTE 11 COMMITMENTS, CONTINGENCIES AND GUARANTEES

Commitments

We lease our facilities under operating leases that expire through 2018. In addition, we are responsible for the real estate taxes and operating expenses related to these facilities. We also have lease commitments for automobiles and office equipment.

Minimum annual rental payments under these agreements are as follows (in thousands):

Years Ending December 31,		Amount
2007	e	7.600
2006 2007	\$	7,608 6,506 5,842 4,831 4,400 16,703
2008		5,842
2009 2010		4,831
2010 Thereafter		4,400 16.702
Thereare	<u> </u>	10,703
	9	45,890
	3	45,670

Rent expense charged to operations under operating leases was approximately \$7.7 million, \$6.6 million and \$6.0 million for the years ended December 31, 2005, 2004 and 2003, respectively.

Under the terms of certain supply agreements with suppliers of our veterinary instruments, slides for our VetTest[®] Chemistry Analyzers; electrolyte instruments, components and consumables; our VetAutoreadTM Hematology Analyzers, components and consumables; and certain raw materials, we have aggregate commitments to purchase approximately \$138.7 million of products through 2020. In addition, we have various minimum royalty payments due through 2019 of \$13.5 million.

We committed up to an aggregate of \$4.0 million of capital purchase obligations in connection with the design and construction of automated production equipment at Ortho's facility that will be used to manufacture consumables for use in our next-generation chemistry analyzer. We expect to pay \$1.9 million of our total commitment in 2006, \$1.2 million in 2007 and the remainder in 2008.

In January 2006, we entered into an agreement to purchase the building in which our headquarters facility is located for \$18.0 million less the face value of the mortgage in cash and also agreed to assume the mortgage. We believe the face value of the mortgage will be approximately \$6.5 million (unaudited) on the closing date with a fair market value of \$7.6 million (unaudited). The closing is subject to certain conditions to closing, including completion of our due diligence review and receipt of certain state and local incentives.

We have a 40% equity interest in a joint venture formed to assemble and market veterinary diagnostic products for production animals in China. During the year ended December 31, 2005, we made capital contributions of \$0.6 million to the joint venture. We agreed to purchase an additional 55% equity interest in the joint venture from our partner, subject to approval by the Chinese government of the ownership change, and committed to pay \$0.8 million over two years in consideration for the additional equity. In addition, the joint venture entered into a contract with the joint venture partner where the partner will provide promotional and agency services and will receive sales commissions at rates escalating from 2.5% to 8.5% annually based on sales volume. See Note 17 for additional information regarding the joint venture.

Contingencies

We are subject to claims that arise in the ordinary course of business, including with respect to actual and threatened litigation and other matters. We accrue contingent liabilities when it is probable that future expenditures will be made and such expenditures can reasonably be estimated. However, our actual losses with respect to these contingencies could exceed our accruals.

In October 2004, we resolved a contingent liability for a third-party claim related to alleged patent infringement. As a result, we recognized reductions of previously accrued expenses during 2004 of \$1.8 million in cost of product revenue.

F-23

Under our workers' compensation insurance policy for U.S. employees for the years ended December 31, 2005, 2004 and 2003, we retain the first \$250,000 in claim liability per incident and \$2.8 million, \$3.0 million and \$1.4 million, respectively, in aggregate claim liability. We entered into a similar workers' compensation insurance policy effective January 1, 2006. We estimate claim liability based on claims incurred and the estimated ultimate cost to settle the claims. Based on this analysis, we have recognized cumulative expenses of \$0.7 million, \$0.6 million and \$0.7 million for claims incurred during the years ended December 31, 2005, 2004 and 2003, respectively. In connection with these policies, we have outstanding letters of credit totaling \$1.6 million to the insurance companies as security for these claims at December 31, 2005. In 2006, we agreed to increase an existing letter of credit by \$0.6 million for the 2006 policy year and we reduced another letter of credit for the policy years 2003 and 2004 by \$0.6 million.

Under our employee health care insurance policy, we retain claims liability risk up to \$125,000 per incident and an aggregate claim limit based on the number of employees enrolled in the plan per month, which was estimated to be \$13.9 million at December 31, 2005. We estimate our liability for the uninsured portion of employee health care obligations based on individual and aggregate coverage, our claims experience, the number of employees enrolled in the program, and the average time from when a claim is incurred to the time it is reported. Should actual employee health care claims liability exceed estimates, we are liable for up to an additional \$1.7 million for potential uninsured obligations at December 31, 2005. We have insurance coverage of \$1.0 million for claims above the aggregate limit. Should employee health insurance claims exceed this coverage, we would have further obligations for the amount in excess of such coverage.

We have entered into employment agreements with two of our officers whereby payments may be required if we terminate their employment without cause. The amounts payable are based upon the executives' salaries at the time of termination and the cost to us of continuing to provide certain benefits. Had both of such officers been terminated as of December 31, 2005, we would have had aggregate obligations for salaries and benefits of approximately \$1.9 million under such agreements. We have entered into employment agreements with each of our officers that require us to make certain payments in the event the officer's employment is terminated under certain circumstances within a certain period following a change in control of our stock. The amounts payable by us under these agreements is based on the officer's salary and bonus history at the time of termination and the cost us of continuing to provide certain benefits. Had all of our officers been terminated following a change in control as of December 31, 2005, we would have had aggregate obligations of approximately \$10.1 million under these agreements. These agreements also provide for the acceleration of the vesting of all stock options held by two of the officers immediately upon a change in control, and of all stock options held by our other executive officers upon any qualifying termination following a change in control. We also have employment agreements with other employees through 2009 that provide for total payments of \$1.0 million.

We currently purchase many products and materials from single sources or a limited number of sources. Some of the products that we purchase from these sources are proprietary, and, therefore, cannot be readily replaced by alternative sources. These products include our VetTest[®] Chemistry, VetAutoreadTM Hematology, VetLyte[®] Electrolyte, and VetStatTM Electrolyte and Blood Gas Analyzers and related consumables; certain digital radiography system components, specifically image capture plates and readers; active ingredients for pharmaceutical products; and certain components of our SNAP[®] rapid assay devices, water testing products and LaserCyte[®] Hematology Analyzers. If we are unable to obtain adequate quantities of these products in the future, we could face cost increases or reductions or delays in product shipments, which could have a material adverse effect on our results of operations.

From time to time, we have received notices alleging that our products infringe third-party proprietary rights, although we are not aware of any pending litigation with respect to such claims. Patent litigation frequently is complex and expensive, and the outcome of patent litigation can be difficult to predict. There can be no assurance that we will prevail in any infringement proceedings that may be commenced against us. If we lose any such litigation, we may be stopped from selling certain products and/or we may be required to pay damages as a result of the litigation.

In connection with a business acquisition in September 2005, we have a contingent obligation of up to \$2.3 million for additional purchase price payments to sellers, contingent on the achievement by the acquired business of certain milestones. See Note 3 for additional information regarding business acquisitions.

F-24

Guarantees

The following is a summary of our agreements and obligations that we have determined to be within the scope of FIN 45, "Guaranter's Accounting and Disclosure Requirements for Guarantees Including Indirect Guarantees of Indebtedness of Others, an Interpretation of FASB No. 5, 57 and 107 and a Rescission of FASB Interpretation No. 34" ("FIN 45").

Our Amended and Restated Certificate of Incorporation provides that we will indemnify our officers and directors to the maximum extent permitted by Delaware law. The maximum payment that we may be required to make under such provisions is theoretically unlimited and is impossible to determine. We maintain directors' and officers' liability insurance, which may provide reimbursement to us for payments made to, or on behalf of, officers and directors pursuant to the indemnification provisions. Our indemnification obligations are not within the scope of the provisions of FIN 45. Accordingly, we have recorded no liability for such obligations as of December 31, 2005 and 2004.

In October 2005, our former supplier of VetAutoreadTM Hematology Analyzers and consumables sold this business (including the human hematology testing products division) and we simultaneously entered into a new supply agreement for veterinary products with the acquirer of the business. Under this new supply agreement, we received fixed pricing on certain products through December 31, 2020, among other benefits. In partial consideration for this new supply agreement, we paid cash of \$2.5 million to the acquirer and guaranteed the acquirer's note (the "Note") in the principal amount of \$3.5 million given to our former supplier in partial consideration for the business. The acquirer is obligated to pay the Note through quarterly principal and interest payments through 2008. We are obligated to make a second payment of \$1.25 million upon the achievement of certain milestones by the acquirer, which we expect to occur in approximately 2008, and a third payment of \$1.25 million twelve months later. Our obligations to make the second and third payments are subject to the acquirer's payment of all amounts under the Note and the release of our guaranty. We recorded the fair value of the guaranty of \$0.5 million and recognized the associated assets and liabilities as of the effective date of the agreements.

We enter into agreements with third parties in the ordinary course of business under which we are obligated to indemnify such third parties for and against various risks and losses. The precise terms of such indemnities vary with the nature of the agreement. In many cases, we limit the maximum amount of our indemnification obligations, but in some cases, those obligations may be theoretically unlimited. We have not incurred material expenses in discharging any of these indemnification obligations, and based on our analysis of the nature of the risks involved, we believe that the fair value of these agreements is minimal. Accordingly, we have recorded no liabilities for these obligations as of December 31, 2005 and 2004.

When acquiring a business, we sometimes assume liability for certain events or occurrences that took place prior to the date of acquisition. However, we do not believe we have any probable pre-acquisition liabilities or guarantees that should be recognized as of December 31, 2005 and 2004.

NOTE 12 STOCKHOLDERS' EQUITY

(a) Preferred Stock

Our Board of Directors is authorized, subject to any limitations prescribed by law, without further stockholder approval, to issue from time to time up to 500,000 shares of Preferred Stock, \$1.00 par value per share ("Preferred Stock"), in one or more series. Each such series of Preferred Stock shall have such number of shares, designations, preferences, voting powers, qualifications and special or relative rights or privileges as shall be determined by the Board of Directors, which may include, among others, dividend rights, voting rights, redemption and sinking fund provisions, liquidation preferences, conversion rights and preemptive rights.

F-25

(b) Series A Junior Participating Preferred Stock

On December 17, 1996, we designated 100,000 shares of Preferred Stock as Series A Junior Participating Preferred Stock ("Series A Stock") in connection with our Shareholder Rights Plan. In general, each share of Series A Stock will: (i) be entitled to a minimum preferential quarterly dividend of \$10 per share and to an aggregate dividend of 1,000 times the dividend declared per share of Common Stock, (ii) in the event of liquidation, be entitled to a minimum preferential liquidation payment of \$1,000 per share (plus accrued and unpaid dividends) and to an aggregate payment of 1,000 times the payment made per share of Common Stock, (iii) have 1,000 votes, voting together with the Common Stock, (iv) in the event of any merger, consolidation or other transaction in which Common Stock is exchanged, be entitled to receive 1,000 times the amount received per share of Common Stock and (v) not be redeemable. These rights are protected by customary anti-dilution provisions. There are no shares of Series A Stock outstanding. See Note 13 for additional information regarding preferred stock purchase rights.

NOTE 13 PREFERRED STOCK PURCHASE RIGHTS

On December 17, 1996, we adopted a Shareholder Rights Plan and declared a dividend of one preferred stock purchase right for each outstanding share of Common Stock to stockholders of record at the close of business on December 30, 1996. Under certain conditions, each right may be exercised to purchase one one-thousandth of a share of Series A Stock at a purchase price of \$200.00. The rights will be exercisable only if a person or group has acquired beneficial ownership of 25% or more of the Common Stock or commenced a tender or exchange offer that would result in such a person or group owning 30% or more of the Common Stock. We generally will be entitled to redeem the rights, in whole, but not in part, at a price of \$.01 per right at any time until the tenth business day following a public announcement that a 25% stock position has been acquired and in certain other circumstances.

If any person or group becomes a beneficial owner of 25% or more of the Common Stock (except pursuant to a tender or exchange offer for all shares at a fair price as determined by the outside members of our Board of Directors), each right not owned by a 25% stockholder will enable its holder to purchase such number of shares of Common Stock as is equal to the exercise price of the right divided by one-half of the current market price of the Common Stock on the date of the occurrence of the event. In addition, if we thereafter are acquired in a merger or other business combination with another person or group in which we are not the surviving corporation or in connection with which our Common Stock is changed or converted, or if we sell or transfer 50% or more of our assets or earning power to another person, each right that has not previously been exercised will entitle its holder to purchase such number of shares of common stock of such other person as is equal to the exercise price of the right divided by one-half of the current market price of such common stock on the date of the occurrence of the event.

NOTE 14 TREASURY STOCK

Our Board of Directors has approved the repurchase of up to 16,000,000 shares of our common stock in the open market or in negotiated transactions. During the years ended December 31, 2005, 2004 and 2003, we repurchased approximately 1,993,000, 2,413,000 and 927,000 shares, respectively, of common stock for \$123.8 million, \$128.8 million and \$36.2 million, respectively. Additionally, during 2004 and 2003, we received approximately 1,000 and 133,000 shares of stock, respectively, which were owned by the holder for greater than six months, in payment for the exercise price of stock options. The shares of stock had a fair market value of \$0.1 million and \$4.9 million, respectively. From the inception of the stock repurchase program in August 1999 to December 31, 2005, we repurchased 13,948,000 shares for \$466.1 million and received 170,000 shares of stock with a market value of \$6.0 million in payment for the exercise price of stock options.

NOTE 15 SHARE-BASED COMPENSATION PLANS

Our share-based compensation plans are described below. These plans, and any amendments increasing the number of shares issuable thereunder, were approved by our stockholders

Stock Incentive Plan

During 2003, our Board of Directors approved the 2003 Stock Incentive Plan, as amended (the "2003 Stock Plan") pursuant to which our employees and Directors may receive various types of share-based incentives, including stock options, restricted stock, stock appreciation rights and deferred stock units. A total of 1,850,000 shares of common stock are authorized for issuance under the 2003 Stock Plan, as amended, provided that no more than 1,500,000 shares will be available for the grant of incentive stock options, and no more than 600,000 shares will be available for awards other than stock options and stock appreciation rights (such as restricted stock). In addition, if any options granted under our prior plans, including the 1991 Stock Plan, the 1998 Stock Plan or the 2000 Director Plan, terminate, expire or are forfeited without having been exercised in full, the shares subject to such unexercised options are available for issuance under the 2003 Stock Plan. Options granted under the 2003 Stock Plan and prior plans may not be granted at an exercise price less than the fair market value of the common stock on the date granted (or less than 110% of the fair market value in the case of incentive stock options granted to holders of more than 10% of our Common Stock). Options may not be granted for a term of more than ten years. The vesting schedule of all options granted under the 2003 Stock Plan is determined by the Compensation Committee of the Board of Directors at the time of grant.

F-26

Deferred Compensation Plans

Under our Director Deferred Compensation Plan, non-employee Directors may defer a portion of their cash fees in the form of vested Deferred Stock Units, each of which represents the right to receive one unissued share of our common stock. Directors receive a number of Deferred Stock Units equal to the amount of cash fees deferred divided by the closing sale price of the common stock on the date of deferral. In addition, beginning in 2006, we will grant \$75,000 of Deferred Stock Units to Directors annually in lieu of granting stock options under the 2003 Stock Plan ("Vesting Deferred Stock Units"). Vesting Deferred Stock Units vest one year from the date of grant. Except upon a change in control, as defined in the Director Deferred Compensation Plan, or certain limited circumstances, all Deferred Stock Units will be exchanged for an equivalent number of shares of common stock by us one year following a Director's resignation or retirement. The value of Deferred Stock Units is expensed as compensation when earned.

Under our Executive Deferred Compensation Plan (the "Executive Plan"), certain members of our management may elect to defer a portion of their cash compensation in Deferred Stock Units. These Deferred Stock Units will be exchanged for a fixed number of shares of common stock on dates determined by the employee, subject to the limitations of the Executive Plan and applicable law. Except upon a change in control, as defined in the Executive Plan, or certain other limited circumstances, officers may not receive shares of common stock in settlement of Deferred Stock Units earlier than one year following the termination of their employment.

Deferred Stock Units are presented in the stockholders' equity section of the balance sheet as deferred share-based compensation. During the years ended December 31, 2005, 2004 and 2003, approximately 11,300, 10,400 and 3,300 Deferred Stock Units, respectively, valued at \$0.7 million, \$0.5 million and \$0.1 million, respectively, were issued.

Employee Stock Purchase Plans

During 1997, the Board of Directors approved the 1997 Employee Stock Purchase Plan, under which we reserved and may issue up to an aggregate of 620,000 shares of Common Stock in periodic offerings. Also during 1997, the Board of Directors approved the 1997 International Employee Stock Purchase Plan, under which we reserved and could issue up to an aggregate of 30,000 shares of Common Stock in semiannual offerings. The 1997 International Employee Stock Purchase Plan was terminated in February 2005, and there were no shares remaining thereunder at the time of termination. Prior to July 1, 2005, stock was sold under each of these plans at 85% of its fair market value, as defined in the plans as the lower of the closing price of our common stock at the beginning of the period and the closing price of our common stock at the end of the period. Effective July 1, 2005, we amended the 1997 Employee Stock Purchase Plan to provide that stock is sold at 85% of the closing price of the stock on the last day of the period. Therefore, the fair value of the purchase rights under the program equals the discount from the market price at the exercise date. Shares subscribed to and issued under the plans during the years ended December 31, 2005, 2004 and 2003 were 39,000, 44,000 and 50,000, respectively.

F-27

Summary of Transactions Under Stock Incentive Plans

A summary of the status of options granted under our stock incentive plans at December 31, 2003, 2004 and 2005, and changes during the years then ended, are presented in the table below (in thousands, except weighted average exercise price):

	To	Total			Exercisable			
	Number of Shares		Weighted Average Exercise Price	Number of Shares		Weighted Average Exercise Price		
Outstanding December 31, 2002	5,461	\$	21.47	2,659	\$	18.92		
Granted	948		35.37	ŕ				
Exercised	(1,885)		18.37					
Terminated	(251)		25.73					
Outstanding December 31, 2003	4,273		25.67	1,607	\$	21.71		
Granted	595		51.47					
Exercised	(783)		22.45					
Terminated	(185)		29.33					
Outstanding December 31, 2004	3,900		30.07	1,638	\$	23.71		
Granted	606		57.35	· · ·				
Exercised	(682)		24.71					
Terminated	(77)		44.15					
Outstanding December 31, 2005	3,747		35.17	1,707	\$	26.64		
· · · · · · · · · · · · · · · · · · ·				,				

At December 31, 2005, a total of 955,000 shares of Common Stock were available for future grants under our stock incentive plans.

Summary of Stock Options Outstanding

The following summarizes information about all stock options issued and outstanding at December 31, 2005 (in thousands, except exercise price and per share amounts):

					Options Outstanding				Options Exercisable			
	Exercise Ran			Number Of Options		Weighted Average Exercise Price	Weighted Average Remaining Contract Life	Number of Options		Weighted Average Exercise Price		
\$	13.69-	\$	25.20	1,323	\$	22.34	4.65	1,034	\$	21.78		
	26.63-		34.27	1,146		30.48	6.43	504		29.66		
	34.98-		57.31	1,232		52.37	8.47	154		46.05		
	58.54-		63.42	46		60.22	8.75	15		60.99		

Upon any change in control of the company, 25% of the unvested stock options then outstanding under the 1991 Stock Option Plan, 1991 Director Plan, 1998 Stock Plan, 2000 Director Plan and the 2003 Stock Plan will vest and become exercisable.

Fair Value of Share-Based Compensation

As discussed in Note 2(k), we account for share-based compensation to employees in accordance with APB No. 25, and elect to disclose the pro forma impact of accounting for share-based compensation plans under the provisions of SFAS No. 123 and SFAS No. 148. Accordingly, no SFAS No. 123-based employee compensation cost has been recognized for these plans.

	For the Years Ended December 31,					
	 2005	2004	2003			
Dividend yield	None	None	None			
Expected volatility	40%	40%	55%			
Risk-free interest rate	4.2%	3.1%	3.2%			
Expected life from vesting date to exercise date, in years	2.8	2.8	3.1			

Options granted to Directors vest fully on the first anniversary of the date of grant. Options granted to employees during the years ended December 31, 2005 and 2004 vest over five years at a rate of 20% per year on each anniversary of the date of grant.

Effective July 1, 2005, we amended our employee stock purchase plan to eliminate the look-back option feature. Therefore, the fair value of the purchase rights under the program equals the discount from the market price at the exercise date. For periods ending prior to July 1, 2005, in order to determine the proforma impact under SFAS No. 123, as Amended, the fair value of the purchase rights issued under the employee stock purchase plans is estimated on the date of grant using the Black-Scholes option-pricing model with the following weighted average assumptions:

	For the Years Ended December 31,				
	 2005	2004	2003		
Dividend yield	None	None	None		
Expected volatility	33%	33%	40%		
Risk-free interest rate	3.4%	2.0%	1.0%		
Expected life in years	0.5	0.5	0.5		

The weighted average fair value of options and purchase rights granted were as follows:

	 For the Years Ended December 31,				
	 2005		2004		2003
Weighted average fair value per underlying share:					
Options granted	\$ 25.17	\$	21.59	\$	19.07
Purchase rights granted under employee stock purchase plans	\$ 12.33	\$	12.38	\$	8.96

We calculate pro forma expense under SFAS No. 123, as Amended, using the graded-vesting method.

NOTE 16 IDEXX RETIREMENT AND INCENTIVE SAVINGS PLAN

We have established the IDEXX Retirement and Incentive Savings Plan (the "401(k) Plan"). Employees eligible to participate in the 401(k) Plan may contribute specified percentages of their salaries, a portion of which will be matched by us. We matched \$2.4 million, \$2.0 million, and \$1.7 million for the years ended December 31, 2005, 2004 and 2003, respectively. In addition, we may make contributions to the 401(k) Plan at the discretion of the Board of Directors. There were no discretionary contributions in 2005, 2004 and 2003

NOTE 17 JOINT VENTURE

On June 18, 2003, we formed a joint venture, Beijing IDEXX-Yuanheng Laboratories Co. Limited (the "Venture"), with Beijing Anheal Laboratories Company Ltd. ("Anheal"), formerly known as Beijing Fortunate Century Animal Health Co., Ltd., to assemble and market veterinary diagnostic products for production animals in China. The Venture is headquartered in Beijing, China. Our initial equity interest in the Venture is 40%, however, we are committed to acquire an additional 20% of the Venture from Anheal within two years of the formation of the joint venture, subject to Chinese government approval. We bear an economic risk that is greater than our equity interest and also have the ability to make decisions that significantly affect the results of the activities of the Venture through majority board representation. Therefore, the Venture is consolidated into our financial statements in accordance with FIN 46(R), "Consolidation of Variable Interest Entities - an interpretation of ARB No. 51." We contributed \$1.5 million during the year ended December 31, 2003, and \$0.6 million for the additional 20% interest discussed above and to make an additional \$1.5 million capital contribution to the Venture within three months after the approval by the Chinese government of the additional 20% interest. However, as discussed in Note 11, we modified the joint venture agreement and agreed to acquire 55% of the Venture from Anheal, subject to Chinese government approval, where upon the commitments above will become void.

F-29

We are also obligated to make available to the Venture selected technology, know-how and licenses and to assist with certain logistical, management training and operating matters. In connection with the joint venture agreement, we have not entered into indemnification agreements or assumed liabilities predating the establishment of the Venture.

NOTE 18 SEGMENT REPORTING

We disclose information regarding its segments in accordance with the provisions of SFAS No. 131, "Disclosures about Segments of an Enterprise and Related Information" ("SFAS No. 131"). SFAS No. 131 requires disclosures about operating segments in annual financial statements and requires selected information about operating segments in interim financial statements. It also requires related disclosures about products and services and geographic areas. Operating segments are defined as components of an enterprise about which separate financial information is available that is evaluated regularly by the chief operating decision-maker, or decision-making group, in deciding how to allocate resources and in assessing performance. Our chief operating decision-maker is the Chief Executive Officer.

We are organized into business units by market and customer group. Our reportable operating segments include the Companion Animal Group ("CAG"), the Water testing business ("Water") and the Food Diagnostics Group ("FDG") and other. CAG develops, designs, manufactures, and distributes products and performs services for veterinarians. Water develops, designs, manufactures and distributes products to detect disease and contaminants in food animals and food. Other encompasses activities that are not included in our reportable segments and is primarily comprised of corporate research and development and interest income. Assets categorized as other include cash, short-term investments, long-term investments, deferred tax assets and other miscellaneous current and long-term assets.

For the Years Ended December 31,

	 CAG	 Water	 FDG	 Other	 Consolidated Total
2005					
Revenues	\$ 520,830	\$ 56,760	\$ 60,505	\$ 	\$ 638,095
Income (loss) from operations	\$ 82,970	\$ 25,974	\$ 9,894	\$ (3,507)	\$ 115,331
Interest income	 		 		3,141
Income before provisions for income taxes and partner's interest					118,472
Provision for income taxes					40,670
Partner's interest in loss of subsidiary					(452)
Net income					\$ 78,254
Depreciation and amortization	\$ 21,236	\$ 447	\$ 2,686	\$ 	\$ 24,369
Segment assets	266,207	35,696	32,077	156,696	490,676
Expenditures for property (1)	23,402	119	1,311		24,832
2004					
Revenues	\$ 448,687	\$ 53,098	\$ 47,396	\$ 	\$ 549,181
Income (loss) from operations	\$ 77,123	\$ 24,259	\$ 9,831	\$ (3,178)	\$ 108,035
Interest income				 	 3,068
Income before provisions for income taxes and partner's interest					111,103
Provision for income taxes Partner's interest in loss of subsidiary					33,165 (394)
,					 (67.)
Net income					\$ 78,332
Depreciation and amortization	\$ 16,794	\$ 507	\$ 1,126	\$ 	\$ 18,427
Segment assets	263,858	30,832	39,820	179,727	514,237
Expenditures for property (1)	27,541	694	3,630		31,865
2003					
Revenues	\$ 384,419	\$ 46,936	\$ 44,637	\$ 	\$ 475,992
Income (loss) from operations	\$ 55,216	\$ 20,934	\$ 7,606	\$ (3,369)	\$ 80,387
Interest income				 	 2,867
Income before provisions for income taxes and partners's interst					83,254
Provision for income taxes					26,278
Partner's interest in loss of subsidiary					(114)
Net income					\$ 57,090
Depreciation and amortization	\$ 18,079	\$ 317	\$ 501	\$ 	\$ 18,897
Segment assets	198,267	27,330	16,119	280,159	521,875
Expenditures for property	16,115	109	672		16,896

⁽¹⁾ Expenditures for property for the year ended December 31, 2005 include \$0.6 million for property acquired in connection with CAG business acquisitions. Expenditures for property for the year ended December 31, 2004 include \$2.1 million and \$0.7 million for property acquired in connection with CAG and FDG business acquisitions, respectively.

F-31

Revenues by product and service categories were as follows (in thousands):

		December 31,							
		005	2004		2003				
CAG revenue:									
Instruments and consumables	\$ 217	537 \$	197,939	\$	177,374				
Rapid assay products	100		93,506		82,978				
Reference laboratory and consulting services	156	425	118,596		94,650				
Practice information management and digital									
radiography systems		589	28,163		22,463				
Pharmaceutical products	14	024	10,483		6,954				
Net CAG revenue	520	830	448,687		384,419				
Net Water revenue	56	760	53,098		46,936				
FDG revenue									
Production animal products and services		945	31,690		28,580				
Dairy testing products	15	560	15,706		16,057				
Net FDG revenue	60	505	47,396		44,637				
Net revenue	\$ 638	095 \$	549,181	S	475,992				

Revenue by principal geographic area, based on customers' domiciles, was as follows (in thousands):

	For the Years Ended December 31,							
	 2005		2004		2003			
Americas	 							
United States	\$ 418,565	\$	373,615	\$	331,852			
Canada	18,428		16,486		14,688			
Other Americas	6,235		4,766		4,803			
	 443,228		394,867		351,343			
	 443,220		374,807		331,343			
Europe								
United Kingdom	46,419		43,365		36,521			
Germany	38,994		20,595		16,295			
France	19,300		15,148		11,653			
Other Europe	 49,468		35,045		25,936			
	154,181		114,153		90,405			
	 			-				
Asia Pacific Region								
Japan	17,531		16,533		15,077			
Australia	15,618		16,308		13,566			
Other Asia Pacific	 7,537		7,320		5,601			
	40,686		40,161		34,244			
Total	\$ 638,095	\$	549,181	\$	475,992			
	 050,075	Ψ	5 15,101	•	475,772			

In 2005, two of our CAG distributors, The Butler Company and Burns Veterinary Supply, Inc., merged and, as a result, they collectively accounted for 10% of our 2005 revenue and 4% of our net accounts receivable at December 31, 2005. In 2004 and 2003, no customer accounted for greater than 10% of our revenue. Our largest customers are our U.S. distributors of our products in the CAG segment. The largest consumer of our products and services accounts for approximately 1% of our sales.

F-32

Net long-lived assets by principal geographic areas include net property and equipment, goodwill and other intangible assets. These long-lived assets are subject to geographic risks because they are generally difficult to move and effectively utilize in another geographic area in a reasonable time period and because they are relatively illiquid. Net long-lived assets by principal geographic areas were as follows (in thousands):

	December 31,							
	 2005		2004		2003			
Americas								
United States	\$ 97,268	\$	94,573	\$	77,176			
Other Americas	174		170		22			
	 97,442		94,743		77,198			
Europe	 	<u> </u>						
United Kingdom	26,878		25,336		19,266			
Germany	32,282		35,817		54			
Switzerland	17,009		20,351					
France	1,297		58		84			
Netherlands	2,632		2,505		1,753			
Other Europe	 80		659		547			
	80,178		84,726		21,704			
Asia Pacific Region	 							
Japan	386		518		594			
Australia	5,846		6,454		6,517			
Other Asia Pacific	592		683		977			
	 6,824		7,655		8,088			
Total	\$ 184,444	\$	187,124	\$	106,990			

NOTE 19 SUMMARY OF QUARTERLY DATA (UNAUDITED)

A summary of quarterly data follows (in thousands, except per share data):

	_						
	March 3		June 30,		June 30, September 30,		December 31,
2005							
Revenue	\$	152,426	\$	160,630	\$	158,069	\$ 166,970
Gross profit		76,080		80,575		81,329	84,916
Operating income		26,138		28,886		30,123	30,184
Net income		17,690		19,933		20,604	20,027
Earnings per share:							
Basic	\$	0.54	\$	0.61	\$	0.63	\$ 0.63
Diluted	\$	0.51	\$	0.59	\$	0.61	\$ 0.60
2004							
Revenue	\$	133,417	\$	137,379	\$	134,111	\$ 144,274
Gross profit		67,046		72,002		71,058	68,911
Operating income		25,301		31,055		28,404	23,275
Net income		17,791		23,910		19,696	16,935
Earnings per share:							
Basic	\$	0.51	\$	0.69	\$	0.58	\$ 0.51

For the Quarters Ended

F-33

SCHEDULE II IDEXX LABORATORIES, INC. AND SUBSIDIARIES

VALUATION AND QUALIFYING ACCOUNTS

(in thousands)

		Balance at Beginning of Year		Beginning		Beginning		Charges to Costs and Expenses	 Write-Offs/ Cash Payments	 Balance at End of Year
Allowance for doubtful accounts receivable:										
December 31, 2003	\$	2,415	\$	114	\$ 579	\$ 1,950				
December 31, 2004		1,950		(294)	162	1,494				
December 31, 2005		1,494		121	394	1,221				
Accrued severance and lease cancellation reserve (including CEO Succession Charge):										
December 31, 2003	\$	1,042	\$	255	\$ 1,040	\$ 257				
December 31, 2004		257		283	293	247				
December 31, 2005		247		590	448	389				
Valuation allowance for deferred tax assets:										
December 31, 2003	\$	5,142	\$	516	\$ 31	\$ 5,627				
December 31, 2004		5,627		(615)	69	4,943				
December 31, 2005		4,943		541	588	4,896				

F-34

Restricted Stock Unit Agreement Granted Under IDEXX Laboratories, Inc. 2003 Stock Incentive Plan

1. Grant of Restricted Stock.

IDEXX Laboratories, Inc., a Delaware corporation (the "Company"), hereby grants to the Participant a Restricted Stock Unit Award consisting of the number of Restricted Stock Units ("RSUs") stated on the reverse side of this Agreement. Each RSU represents the right to receive one share of common stock, \$.01 par value, of the Company (individually a "Share" and collectively the "Shares"). The Company will record on its books the grant of the RSUs to the Participant and will issue Shares upon vesting of the RSUs as provided below. This award of RSUs is subject to the terms and conditions set forth in this Agreement, the Company's 2003 Stock Incentive Plan (the "Plan"), and the description of the Plan set forth in the Plan Prospectus. The Plan and Plan Prospectus are provided to the Participant with this Agreement. Defined terms not otherwise defined in this Agreement shall have the meanings set forth in the Plan or the Prospectus.

2. Vesting and Forfeiture.

- (a) Vesting. Subject to Section 3(b), the RSUs shall vest and become nonforfeitable in accordance with the vesting schedule set forth on the reverse side of this Agreement.
- **(b) Forfeiture.** Except as otherwise provided in this Section 3, in the event that the Participant ceases to be employed by the Company or a member of the Board of Directors of the Company (an "Eligible Grantee") for any reason or no reason, with or without cause, the balance of RSUs that have not vested as of the date of such cessation will be forfeited and the Participant will have no future rights with respect to any such unvested RSUs. For all purposes of this Agreement, (i) "employment" shall be defined in accordance with the provisions of Section 1.421-7(h) of the Income Tax Regulations or any successor regulations, and (ii) if this Agreement shall be assumed or a new Agreement substituted therefor in a transaction to which Section 425(a) of the Code applies, employment by such assuming or substituting corporation shall be considered for all purposes of this Agreement to be employment by the Company. If the Participant is employed by a parent or subsidiary of the Company, any references in this Agreement to employment with the Company or termination of employment by or with the Company shall instead be deemed to refer to such parent or subsidiary.

3. Restrictions on Tranfer.

The Participant may not sell, assign, transfer, pledge, hypothecate or otherwise dispose of by operation of law or otherwise, any RSUs, or any interest therein, except by will or the laws of descent and distribution.

4. Rights as Stockholder.

Neither the Participant, nor any person claiming through the Participant, will have any of the rights or privileges of a stockholder of the Company with respect to the RSUs unless and until Shares have been issued, recorded on the records of the Company or its transfer agent, and delivered to the Participant upon vesting of the RSUs. No adjustment shall be made for dividends or distributions or other rights for which the record date is prior to the date such Shares are issued. After such issuance, recordation and delivery, the Participant will have all the rights of a stockholder of the Company with respect to the Shares.

5. Delivery of Shares; Compliance with Securities Laws, Etc.

- (a) General. The Company shall, upon vesting of RSUs hereunder, make prompt delivery of vested Shares to the Participant, provided that if any law or regulation requires the Company to take any action with respect to such Shares before the issuance thereof, then the date of delivery of such Shares shall be extended for the period necessary to complete such action.
- (b) Listing, Qualification, Etc. This Agreement shall be subject to the requirement that if, at any time, counsel to the Company shall determine that the listing, registration or qualification of Shares subject hereto upon any securities exchange or under any state or federal law, or the consent or approval of any governmental or regulatory body, or that the disclosure of non-public information or the satisfaction of any other condition is necessary as a condition of, or in connection with, the issuance of Shares hereunder, then such issuance shall be deferred until such listing, registration, qualification, consent or approval, disclosure or satisfaction of such other condition shall have been effected or obtained on terms acceptable to the Board of Directors. Nothing herein shall be deemed to require the Company to apply for, effect disclosure, or to satisfy such other condition.

6. No Special Employment Rights.

Nothing contained in the Plan, the Prospectus or this Agreement shall be construed or deemed to constitute an employment or service contract or confer or be deemed to confer on the Participant any right to continue in the employ or service of, or to continue any other relationship with, the Company or limit in any way the right of the Company to terminate the Participant's employment or service or other relationship at any time, with or without cause.

7. Withholding Taxes; Section 83(b) election.

- (a) No Shares will be delivered pursuant to the vesting of an RSU unless and until the Participant satisfies any federal, state or local withholding tax obligation required by law to be withheld in respect of this award. The Participant acknowledges and agrees that to satisfy any such tax obligation, the Company may deduct and retain from the Shares to be distributed upon vesting of RSUs such number of Shares as is equal in value to the Company's minimum statutory withholding obligations with respect to the income recognized by the Participant upon such vesting (based on minimum statutory withholding rates for federal and state tax purposes, including payroll taxes, that are applicable to such income), based on the closing price of the Company's common stock on the date of vesting.
 - (b) The Participant acknowledges that no election under Section 83(b) of the Internal Revenue Code of 1986 may be filed with respect to this award.

8. Data Privacy.

By entering into this Agreement, the Participant: (i) authorizes the Company and its Subsidiaries, and any agent of the Company and its Subsidiaries administering the Plan or providing Plan recordkeeping services, to disclose to the Company or any of its Subsidiaries such information and data as the Company or any such Subsidiary shall request in order to facilitate administration of the Plan; (ii) waives any data privacy rights he or she may have with respect to such information; and (iii) authorizes the Company and its Subsidiaries to store and transmit such information in electronic form.

9. Miscellaneous.

- (a) Except as provided herein, this Agreement may not be amended or otherwise modified unless evidenced in writing and signed by the Company and the Participant. Any provision for the benefit of the Company contained in this Agreement may be waived, either generally or in any particular instance, by the Board of Directors of the Company. The Board of Directors may amend, alter, suspend, discontinue or terminate the Plan, or any portion thereof, at any time, subject to the requirements for certain amendments or alterations set forth in the Plan.
- **(b)** All notices under this award shall be mailed or delivered by hand to the parties at their respective addresses set forth on the opposite side of this Agreement or at such other address as may be designated in writing by either of the parties to one another.
- (c) This Agreement and the Plan constitute the entire agreement between the parties, and supersedes all prior agreements and understandings, relating to the subject matter of this Agreement. The invalidity or unenforceability of any provision of this Agreement shall not affect the validity or enforceability of any other provision of this Agreement, and each other provision of this Agreement shall be severable and enforceable to the extent permitted by law.
- (d) This Agreement shall be binding upon and inure to the benefit of the Company and the Participant and their respective heirs, executors, administrators, representatives, successors and assigns, subject to the restrictions on transfer set forth in Section 3 hereof.
- (e) The right of the Participant to receive Shares pursuant to this award is an unfunded and unsecured obligation of the Company. The Participant shall have no rights under this award other than those of an unsecured general creditor of the Company.
- (f) This award shall be governed by and construed in accordance with the laws of the State of Delaware and applicable federal law, without regard to applicable conflicts of laws.

PURCHASE AND SALE AGREEMENT

(80 Eisenhower Drive — Westbrook, Maine)

This PURCHASE AND SALE AGREEMENT (this "Agreement") is made and entered into as of the 17th day of January, 2006 (the "Effective Date"), by and among CW WESTBROOK LIMITED PARTNERSHIP, a Delaware limited partnership having an address of c/o S.R. Weiner and Associates, Inc., 1330 Boylston Street, Chestnut Hill, Massachusetts 02467, ("Seller") and IDEXX LABORATORIES, INC., a Delaware corporation having an address of 80 Eisenhower Drive, Westbrook, Maine 04092 ("Buyer").

In consideration of Ten Dollars (\$10.00) and other good and valuable consideration, the receipt, sufficiency and delivery of which are hereby acknowledged, the parties hereto hereby agree as follows:

- 1. <u>AGREEMENT TO PURCHASE AND SELL.</u> Seller hereby agrees to sell, and Buyer hereby agrees to buy, subject to the terms and conditions of this Agreement, the following real and personal property (collectively, the "<u>Property</u>"):
- 1.1 the land, as more particularly described in <u>Exhibit A</u> attached hereto and made a part hereof, together with all mineral and water rights, easements, rights of way, privileges and benefits appurtenant thereto (the "<u>Land</u>"), and all buildings and improvements thereon (the "<u>Improvements</u>"). The Land and the Improvements are collectively referred to as the "<u>Real Property</u>";
- 1.2 all of Seller's right, title and interest, if any, in and to all fixtures and equipment now used in connection with the operation of the Improvements and located therein including, without limiting the generality of the foregoing, any of the following: boilers, pumps, tanks, electric panel switchboards, lighting equipment and wiring, heating, plumbing, ventilating and air conditioning apparatus and equipment, elevators, escalators, and conveyors, and all other personal property owned by Seller and located at the Property (the "Personalty");
- 1.3 all of Seller's right, title and interest in and to all leases and occupancy agreements for any portion of the Real Property, including, without limitation, the Leases (as defined below), together with any and all guarantees, deposits and escrows and prepaid rents relating to or serving as security for the Leases and any files kept by Seller in connection with the Leases; provided, however, that with respect to the Sanmina Lease identified on Exhibit C, Seller shall assign its rights, in common with Seller such that both parties shall have the right to enforce such rights independently of the other, but Buyer shall not assume any obligations thereunder;
- 1.4 all of Seller's right, title and interest, if any, in and to all intangible property used in connection with the foregoing, including, without limitation, all contract rights, licenses, permits, approvals, utility contracts, telephone exchange numbers, advertising materials, plan and specifications, drawings, surveys, governmental approvals and development rights and all warranties and guaranties (if any) issued to or held by Seller by any manufacturer, supplier, contractor, subcontractor or materialman in connection with the construction or installation of the Improvements or equipment or fixtures included as part of the Property or the maintenance of the Property;

1

1.5 the interests of Seller in and to any contracts relating to the operation, ownership, maintenance or management of the Real Property ("Service Contracts") and listed on Exhibit E attached hereto, but only to the extent Buyer elects to assume any such contracts (each an "Assumed Contract"), which election shall be evidenced by written notice to Seller prior to the end of the Inspection Period. Any Service Contract not assumed by Buyer will be terminated by Seller as of the Closing. Notwithstanding the foregoing, the existing property management agreement between Seller and S.R. Weiner and Associates, Inc., if any, shall be terminated at Closing; and

1.6 all of Seller's rights and interests in and to any easements affecting the Premises, including the right to receive any payments in connection therewith.

2. PURCHASE PRICE; ASSUMED LOAN.

- 2.1 The purchase price for the Property (the "Purchase Price") shall be Eighteen Million Dollars (\$18,000,000.00), subject to adjustment as hereinafter provided, minus the sum of (x) the outstanding principal balance of the Assumed Loan, and (y) unpaid interest thereon accrued through the "Closing Date" (as hereinafter defined), and (z) all other amounts then due and owing under the Assumed Loan. Subject to the terms and conditions of this Agreement, the Purchase Price shall be paid as follows:
 - 2.1.1 Within one (1) "Business Day" (as defined in Section 17.14 below) after the Effective Date, a cash deposit by wire transfer of immediately available funds in the amount of Six Hundred Thousand Dollars (\$600,000.00) (together with any interest earned thereon, the "Deposit") shall be paid by Buyer to Lawyers Title Insurance Corporation, as escrow agent ("Escrow Agent"), and the Deposit shall be held and paid in accordance with the terms of this Agreement.
 - 2.1.2 The balance of the Purchase Price, subject to adjustments and prorations provided for in this Agreement, shall be paid by Buyer to Escrow Agent at the Closing (as hereinafter defined) by wire transfer of immediately available federal funds and Escrow Agent shall transfer such funds, together with the Deposit by wire transfer of immediately available funds to such account(s) as Seller may designate in writing.
- 2.2 Escrow Agent is executing this Agreement to acknowledge Escrow Agent's responsibilities hereunder. Any amendment to this Agreement that is not signed by Escrow Agent shall be effective as to the signatories thereto, but shall not be binding on Escrow Agent. Escrow Agent shall accept the Deposit with the understanding of the parties that Escrow Agent is not a party to this Agreement except to the extent of its specific responsibilities hereunder, and does not assume or have any liability for the performance or non-performance of Buyer or Seller hereunder to either of them and need not be joined in any amendment hereto, except to the extent any such amendment modifies Escrow Agent's rights or obligations hereunder. Additional provisions with respect to the Escrow Agent are set forth in Section 16 below. If the Escrow Agent requires an additional agreement or provisions, the parties agree to negotiate such agreement or provisions in good faith.

2

2.3 Subject to the provisions of this Agreement, the Property will be conveyed subject to that certain loan (the "Assumed Loan") in the original principal amount of Nine Million Five Hundred and no/100 Dollars (\$9,500,000.00) as the same may have been modified, amended or extended, made by Sun Life Assurance Company of Canada (the "Lender"), which Assumed Loan is evidenced and secured by the documents listed on Exhibit B attached hereto (the "Assumed Loan Documents"). Prior to the Closing, Seller and Buyer shall use good faith efforts to obtain the approval of the Lender (and/or, if applicable, the loan servicer ("Servicer")) to the transfer of the Property to Buyer subject to the Assumed Loan. Seller shall use good faith efforts to have the Lender include in such approval document such estoppel provisions pertaining to the Assumed Loan as Buyer shall reasonably request. All loan assumption fees and all other costs and fees and expenses (including legal fees) charged by the Lender shall be borne by Seller. Except as set forth in the preceding sentence, any and all costs and expenses incurred by Buyer in connection with the assumption of the Assumed Loan shall be paid by Buyer. Buyer agrees to cooperate

in good faith and to provide, execute and deliver, as applicable, such documents and financial information to Lender or Servicer for review and approval prior to and in connection with the approval of Buyer's assumption of the Assumed Loan as Lender and/or Servicer shall reasonably require, in form and substance reasonably acceptable to Buyer, but in no event on terms less favorable than set forth in the Assumed Loan Documents. If required by Lender or Servicer, Buyer shall form a "single purpose" subsidiary to acquire title to the Property and to assume the Assumed Loan. In no event shall Buyer be obligated to incur or assume any liability in excess of Seller's current obligations under the Assumed Loan Documents. It shall be a condition to Buyer's obligations hereunder, that the Assumed Loan Documents shall be amended to Buyer's reasonable satisfaction to (a) accommodate the fact that Buyer is a publicly traded company and to make the obligations and restrictions thereunder consistent with the rights and obligations of Buyer as a publicly traded company; (b) to eliminate any restrictions on the transfer of stock of Buyer; (c) eliminate any requirements or obligations specific to Seller and not reasonably capable of being satisfied or performed by Buyer; and (d) address such other provisions as Buyer shall reasonably require. In addition, the Assumed Loan Documents shall be amended to provide for the Lender's release of Seller and any guarantor or indemnitor of Seller's obligations under the Assumed Loan Documents from any and all liability thereunder. At Closing, Seller shall assign to Buyer all of Seller's rights to all escrow balances or reserves, if any, maintained by the Lender or Servicer in connection with the Assumed Loan, and Seller shall receive a credit from Buyer in the amount of such escrow balances or reserves, subject to confirmation from Lender or Servicer. Notwithstanding the foregoing, if Seller and Buyer have not obtained the Lender's and Servicer's approval to the assumption of the Assumed Loan by Buyer or its affiliate on or before the Closing Date (the "Lender's Consent"), then the Closing Date shall automatically be extended for up to thirty (30) days to allow Seller to obtain such approval prior to the Closing. In the event approval for the assumption of the Assumed Loan is not obtained prior to the Closing Date (as such date may be extended as aforesaid), Seller or Buyer may terminate this Agreement by written notice given to the other on or before the Closing Date (as extended). If the Loan Documents are not amended as provided above to Buyer's satisfaction, or if Seller is not able to secure from the lender estoppel provisions reasonably satisfactory to Buyer, Buyer may terminate this Agreement by written notice given to Seller on or before the Closing Date (as extended). Upon termination of this Agreement, as provided in either of the preceding two sentences, the Deposit shall be returned to Buyer and this Agreement shall terminate without further recourse to either party, except that those obligations that pursuant to the express terms hereof survive the termination of this Agreement shall remain in effect. Up to and including the Closing Date, Seller shall (x) continue to perform all obligations and to make all required payments in the manner and at the times specified in the Assumed Loan Documents; and (y) use its best efforts to prevent from occurring any event that with notice or the passage of time, or both, would constitute a default under the Assumed Loan Documents.

3

3. REPRESENTATIONS AND WARRANTIES OF SELLER.

- 3.1 Representations and Warranties. Seller represents and warrants to Buyer as follows as of the date hereof and as of the Closing:
 - 3.1.1 <u>Authority.</u> Seller is a limited partnership, duly organized, validly existing and in good standing under the laws of the State of Delaware, and has all requisite power and authority to enter into this Agreement and all documents now or hereafter to be executed and delivered by Seller pursuant to this Agreement (collectively, the "<u>Seller's Documents</u>") and to perform its obligations hereunder and under Seller's Documents.
 - 3.1.2 No Conflict; Due Authorization. The execution and delivery of this Agreement and the consummation of the transactions contemplated hereunder on the part of Seller does not and will not conflict with the Partnership Agreement establishing Seller, and no consents or waivers of or by any third party are necessary to permit the consummation by Seller of the transactions contemplated by this Agreement, other than the consent of Lender or Servicer as provided in Section 2.3. This Agreement and all documents that are to be executed by Seller and delivered to Buyer at the Closing are, and at the Closing shall be, duly authorized, executed and delivered by Seller, and are, and at the Closing will be, legal, valid and binding obligations of Seller, enforceable in accordance with their terms, and, when executed, and also at the time of Closing, will not violate any provisions of any agreement or judicial order to which Seller is a party or to which Seller or the Property is subject.
 - 3.1.3 <u>Leases</u>. There are no leases or occupancy agreements affecting the Property except for the Leases set forth on <u>Exhibit C</u> attached hereto (together, the "<u>Leases</u>", and each individually, a "<u>Lease</u>"). Seller has paid in full all brokerage commissions and leasing fees connected with the Leases. <u>Exhibit D</u> sets forth a true and complete list of all security deposits (and all accrued interest thereon, if any) currently being held by Seller under the Leases, and whether held in cash, letter of credit or otherwise. As of the Closing Date, the term of the lease between Seller, as lessor, and PRECISmetals, Inc. (whose interest is now owned by Sanmina-SCI) as lessee, shall have expired. Seller shall remain responsible for, and hereby represents, covenants and agrees that it will pay, any and all amounts owed to the tenant under the Sanmina Lease, including without limitation the return of any security deposit, reconciliation of operating expenses, real estate taxes and prepaid rent.

4

- 3.1.4 <u>Contracts</u>. Seller has delivered to Buyer true and complete copies of all of the Service Contracts in effect with respect to the Property, a true, correct and complete list of which is attached hereto as <u>Exhibit E</u>. Seller represents and covenants that it shall be responsible for, and shall pay, all amounts due and owing under all Service Contracts not assumed by Buyer, and with respect to Assumed Contracts, all amounts due and owing to the extent that such amounts pertain to work performed or materials supplied prior to the Closing Date.
 - 3.1.5 FIRPTA. Seller is not a "foreign person" as defined in Section 1445(f)(3) of the Internal Revenue Code.
- 3.1.6 <u>Litigation</u>. Other than matters set forth on <u>Exhibit F-1</u>, there is not now pending nor, to Seller's knowledge, has there been threatened any action, claim, suit or proceeding against Seller or the Property before or by any federal or state court, commission, regulatory body, administrative agency or other governmental body, domestic or foreign, which relates to the Seller or the Property. To the best of Seller's knowledge, there are no pending or contemplated insurance claims or filings with respect to the Property and no pending or contemplated tax abatement or reduction proceedings.
- 3.1.7 <u>Violations of Law</u>. Except as set forth on <u>Exhibit F-2</u> attached hereto, Seller has not received any written notice from any governmental authority having jurisdiction asserting (a) that the Property or any part thereof, or any existing use thereof or any existing Tenant, is in violation of any law, ordinance, rule or regulation applicable to the Property which has not been cured, including without limitation any one or more of the "Environmental Laws" (as defined below") and Seller has no knowledge of any such violation, or (b) that Seller lacks any permits required to be obtained by Seller under any applicable laws or regulations.
- 3.1.8 Environmental Matters. In addition to the representations of Seller in Sections 3.1.6 and 3.1.7 of this Agreement, Seller makes the following additional representations with respect to the presence of Hazardous Materials on the Property in violation of Environmental Laws (as hereinafter defined). Attached hereto as Exhibit G is a list of all reports which are in Seller's possession or control and which relate to the presence or absence of Hazardous Materials on, in or under the Property (collectively the "Environmental Reports"). True, correct and complete copies of the Environmental Reports have been delivered to Buyer. To Seller's knowledge, except as disclosed in the Environmental Reports and other than routine amounts of cleaning fluids and similar substances which have customarily been used in the routine operation or maintenance of the Property in compliance with Environmental Laws, as hereinafter defined, there has been no release of Hazardous Materials on, in or under the Property in amounts or concentrations that would require reporting under the Environmental Laws, or that would result in a violation of the Environmental Laws. Except as set forth in the Environmental Reports or as listed on Exhibit G-1 attached hereto, Seller has not received, nor is it aware of, any written notice from any government authority or other entity or person (i) asserting that either any condition existed or exists at the Property which constitutes or has resulted in a violation of any Environmental Laws, or (ii) stating facts which would reasonably be expected to give rise to a claim under Environmental Laws. Notwithstanding the foregoing,

5

The term "Environmental Laws", as used in this Agreement, means all federal, state, or local laws, rules, regulations or requirements (whether now existing or hereafter enacted or promulgated) and including common law, and any judicial or administrative orders or judgments, relating to Hazardous Materials in addition to the following: Stormwater Management Law, 38 M.R.S.A. § 420-D; Natural Resources Protection Act, 38 M.R.S.A. §§ 480-A et seq.; Site Location of Development Law, 38 M.R.S.A. §§ 481 et seq.

The term "Hazardous Materials", as used in this Agreement, means any substance, chemical, compound, product, solid, gas, liquid, waste, byproduct, pollutant, contaminant or material which is hazardous, toxic, ignitable, corrosive, carcinogenic or otherwise dangerous to human, plant or animal life or the environment or which are defined, determined or identified as such in any federal, State of Maine or local laws or regulations and which are regulated or subject to clean-up authority under any such laws or regulations, including, but not limited to materials defined as (a) "hazardous waste" under the Federal Resource Conservation and Recovery Act (b) "hazardous substances" under the Federal Comprehensive Environmental Response, Compensation and Liability Act, (C) "pollutants" under the Federal Clean Water Act; (D) "toxic substances" under the Toxic Substances Control Act; and (E) "oil," "hazardous waste," "hazardous matter" or "hazardous substances" under state law, including, without limitation, the State of Maine's Hazardous Waste, Septage and Solid Waste Management Act, as amended, (38 M.R.S.A. §§ 1301-1319-Y) and the regulations promulgated thereunder, and the Maine Uncontrolled Hazardous Substance Sites Law, as amended (38 M.R.S.A. §§1361-1371).

3.1.9 <u>Assumed Loan.</u> Attached hereto as <u>Exhibit B</u> is a true, correct and complete list of the material Assumed Loan Documents to date. Seller represents that is has delivered to Buyer true, correct and complete copies of the Assumed Loan Documents. Seller has not received written notice from the Lender or Servicer asserting an Event of Default under the Assumed Loan Documents that remains uncured on the date hereof. To the best of Seller's knowledge, there is no outstanding event of default under the Assumed Loan Documents and no event has occurred that with notice or the passage of time, or both, would constitute an event of default under the Assumed Loan Documents. Seller is current in all payments of principal and interest due under the Assumed Loan through the last scheduled payment date. <u>Exhibit B</u> accurately sets forth the unpaid principal balance of the Assumed Loan as of the last scheduled payment date (taking into account such payment), and the principal amount of any deposits, reserves or escrows held or established in connection with therewith as of the date specified on <u>Exhibit B</u>, which is the last date on which such information was provided to Seller by Lender or Servicer.

6

- 3.1.10 Condemnation. Seller has not (i) received any written notice from any governmental authority stating that there are any pending or contemplated condemnation, eminent domain or annexation proceedings affecting the Property or any part thereof or (ii) engaged in any discussions or negotiations with any governmental authority regarding any such proceedings or agreements in lieu thereof.
- 3.2 <u>Seller's Knowledge</u>. As used in this Agreement, the term "to Seller's knowledge" or words of similar tenor shall mean only the current actual (and not constructive, imputed or implied) knowledge of the following designees of Seller, after reasonable inquiry: Thomas DeSimone, Al Rocco or Helen Poulin. Seller represents and warrants that the above-designated individuals are the parties with the day-to-day knowledge and responsibility for the Property. Anything herein to the contrary notwithstanding, but subject to the following provisions of this paragraph, no such designee shall have any personal liability or obligation whatsoever, merely by virtue of the preceding sentence of this Section 3.2, with respect to any of the matters set forth in this Agreement or any of the representations made by Seller being or becoming untrue, inaccurate or incomplete in any respect.
- 3.3 <u>Survival of Representations</u>. The representations and warranties contained in this Agreement and in any other document or instrument delivered pursuant to this Agreement shall survive Closing and delivery of the Deed for the duration of the "<u>Survival Period</u>" and, with respect to any written claim in connection with the representations and warranties made within such Survival Period, until final unappealable adjudication or settlement thereof. For purposes of this Agreement, the "<u>Survival Period</u>" shall mean that period commencing on the Closing Date and ending on the first anniversary of the Closing Date. Any claim must be delivered to Seller on or before the end of the Survival Period, time being of the essence. Any claim under this Article 3 is subject to the provisions of <u>Section 6.8</u> herein. Notwithstanding anything herein to the contrary, Seller shall in no event have any liability for breach of any representation, warranty, indemnity or covenant set forth in this Agreement or in any document, instrument or agreement delivered pursuant hereto (i) unless and until the aggregate amount of all such claims for breaches exceeds Twenty Thousand Dollars (\$20,000.00) in which event Buyer shall be entitled to seek recovery of the full amount of its claim under this Section, or (ii) for any amounts in excess of One Million Dollars (\$1,000,000.00) in the aggregate (the "<u>Liability Limits</u>"). To the extent that prior to Closing Buyer discovers any inaccuracy in a representation and warranty of Seller in this Agreement and Buyer or Seller expressly notifies the other party prior to Closing of any such inaccuracy and notwithstanding such inaccuracy and notification, Buyer proceeds to close this transaction, such representation and warranty shall be deemed modified to reflect the inaccuracy discovered by Buyer.
- 3.4 <u>Continued Existence</u>. Seller covenants and agrees that it will not voluntarily take any action to dissolve or terminate its existence and shall use good faith and diligent efforts to remain in good standing under the laws of the State of Maine until such time as its indemnity obligations under Section 6.8 have been satisfied.

7

3.5 Personal Assurance. Stephen R. Weiner, an individual resident of Massachusetts, hereby covenants and agrees with Buyer that until Seller's indemnification obligations under Section 6.8 have been satisfied, he will make available and will contribute to Seller cash in an amount up to the maximum Liability Limit (as defined above in this Section 3.3 hereof), to the extent such funds are required in order to pay amounts due and owing from Seller to Buyer on account of Seller's indemnification obligations set forth in Section 6.8 of this Agreement from time to time and guaranties payment of such funds by Seller to Buyer, to the extent required in order to satisfy Seller's indemnification obligations set forth in Section 6.8 of this Agreement. This is a personal obligation and undertaking of Stephen R. Weiner. If Stephen R. Weiner fails to contribute such funds, as provided above, or if contributed, such funds are not paid to Buyer to the extent of Seller's indemnification obligations set forth in Section 6.8 of this Agreement, Buyer will have enforceable legal and equitable remedies directly against Stephen R. Weiner for the full amount of Seller's indemnity obligations, up to the Liability Limit. So long as Stephen R. Weiner's obligations under this Section 3.5 shall remain outstanding, Stephen R. Weiner covenants and agrees to maintain a net worth (determined in accordance with generally accepted accounting principles) of not less than \$10,000,000 and liquid assets of not less than \$5,000,000. At the Closing, Stephen R. Weiner shall deliver to Buyer a written certification from his accountant certifying that the above net worth and liquidity requirements of Stephen R. Weiner continue to be satisfied as of the date of such certification (an "Accountant's Letter"). Stephen R. Weiner shall provide Buyer with written notice if at any time prior to the expiration of the time period described in Section 3.4 above there shall occur any material adverse change in his financial condition, which notice shall be accompan

4. INSPECTION AND TERMINATION.

4.1 Inspection Period. Through and including the "Inspection Period" (as defined below), Buyer, its agents, employees and contractors, shall be entitled to enter upon the Property including all leased areas, upon reasonable prior notice to Seller, to perform inspections and tests on and of the Property, including surveys, test borings, environmental sampling and other studies, examinations and tests of all structural and mechanical systems within the Improvements, and to examine the books and records of Seller and Seller's property manager relating to the Property and performing such other investigations and analysis as Buyer deems necessary or appropriate. The "Inspection Period" shall be that period of time from the Effective Date of this Agreement to and including the forty-fifth (45th) day after the Effective Date; provided, however, that if Buyer gives written notice to Seller on or before the end of such forty-fifth (45th) day that Buyer needs additional time to complete its due diligence, as described in the preceding sentence or the "Title Commitment" or "Survey" (as defined in Section 5.3 below), the Inspection Period shall be extended for an additional thirty (30) days. If Buyer wishes to engage in any "invasive testing" (as defined below), Buyer shall obtain Seller's prior consent thereto, which consent shall not be unreasonably withheld, conditioned or delayed. Invasive testing shall be testing of the Real Property that involves digging, drilling, cutting or boring into the ground or Improvements. Seller acknowledges that there are currently monitoring wells on the Real Property and that the taking of samples from existing wells shall not constitute invasive testing. Buyer agrees to coordinate its inspections through Seller's property manager, whose name and phone number is provided below; provided, however, that such property manager shall not unreasonably interfere with Buyer's inspection or ability to inspect and provided further that Buyer shall not unreasonably be required to change its schedule of inspections to accommodate the availability of Seller's property manager. Seller's property manager is Helen Poulin and can be reached at (207) 773-8833. Buyer reserves the right for itself, and its agents and contractors, to contact governmental authorities, including those with authority to administer and enforce Environmental Laws, to identify and collect information related to the Property, including Permit information. Such contacts and information gathering may include, but will not be limited to, information identified in the "All Appropriate Inquiry" regulations promulgated by the U.S. Environmental Protection Agency in 40 C.F.R. Part 312, published at 70 Fed. Reg. 66107-113 (November 1, 2005) regardless of the effective date of those regulations.

8

- 4.2 <u>Insurance</u>. Prior to entering into any portion of the Real Property not currently occupied by Buyer as tenant under the IDEXX Lease, Buyer shall deliver to Seller evidence reasonably satisfactory to Seller that Buyer has obtained comprehensive general liability insurance with limits of not less than One Million Dollars (\$1,000,000.00) per occurrence for property damage, bodily or personal injury, or death.
- 4.3 <u>Limitation on Inspections</u>. In connection with the right to enter upon the Property set forth in this Article 4, Buyer agrees (i) to repair any damage to the Property caused by its inspections, except that if Buyer installs groundwater monitoring wells at its own expense Buyer shall not be required to remove the wells, but shall leave them in a safe condition, and (ii) to hold harmless and indemnify the Seller from any and all damages, claims, losses and liabilities (including, without limitation, legal fees and expenses) incurred by Seller in respect of bodily injury, property damage or mechanics liens arising from or related to Buyer's or Buyer's agents' or contractors' or employees' inspections of the Property or its failure to satisfy the conditions of subsection (i) above. The provisions of this subparagraph shall survive the Closing and delivery of the Deed or termination of this Agreement.
- 4.4 <u>Seller Deliveries</u>. Seller has delivered to Buyer copies of all of the items specified on <u>Exhibit H</u>, attached hereto, as qualified by that certain letter from Tom DeSimone to Buyer dated January 6, 2006, attached hereto as part of Exhibit H (the "<u>Seller Documents</u>"). Except as otherwise expressly set forth in Section 3.1 hereof, Seller makes no representations or warranties of any kind regarding the accuracy of the information contained in any of the Seller Documents.
- 4.5 <u>Termination Option</u>. This Agreement shall automatically terminate at the end of the first Business Day following the Inspection Period, as the same may be extended as provided in Section 4.1 above, unless Buyer, on or before the end of such first Business Day gives written notice to Seller that Buyer is prepared to proceed to Closing in accordance with the terms of this Agreement. Without limiting the automatic termination of the preceding sentence, Buyer shall have the right to unilaterally terminate this Agreement at anytime prior to the end of the Inspection Period. Upon such termination, under either scenario, the Deposit shall be returned to Buyer by the Escrow Agent without any further required action by Buyer or Seller and neither party shall have any further liability or obligation to the other hereunder except for any provision that expressly survives the Closing or the termination of this Agreement.

9

5. <u>TITLE</u>.

- 5.1 <u>Title</u>. Title to the Real Property shall be good and marketable subject only to (a) zoning restrictions, (b) customary utility easements, but only to the extent that they exclusively serve the Property, (c) such taxes for the current tax year as are not due and payable as of the date of Closing (the "<u>Closing Date</u>"); (d) the rights of the tenant under the "<u>IDEXX Lease</u>" (as defined in <u>Exhibit C</u>); and (e) any title and survey matters waived or deemed to be waived by Buyer pursuant to Section 5.3 (all of which are collectively referred to herein as "<u>Permitted Exceptions</u>").
- 5.2 <u>Additional Title Considerations</u>. Without limiting the generality of the requirements of Section 5.1 above, the Property shall not be considered to be in compliance with the provisions of this Agreement with respect to title unless the following conditions are either satisfied or waived (or deemed waived) pursuant to Section 5.3 below:
 - 5.2.1 all Improvements shall be wholly within the lot lines of the Land and shall not encroach upon or under any property not within such lot lines;
 - 5.2.2 no building, structure, improvement, parking area, driveway, or property of any kind encroaches on to the Land; and
 - 5.2.3 title to the Real Property is insurable, for the benefit of Buyer, at customary rates, in the ALTA form currently in use, subject only to those Permitted Exceptions (as such terms are defined above) approved or waived (or deemed waived) by Buyer.
- 5.3 <u>Title and Survey.</u> Buyer shall obtain each of the following at its own cost and expense: (i) a current Survey of the Real Property; and (ii) a title insurance commitment for the Real Property issued by a title insurance company selected by Buyer (the "<u>Title Commitment</u>"). Buyer shall notify Seller on or before the expiration of the Inspection Period, as the same may be extended in accordance with Section 4.1 above, of any matters reflected on the Title Commitment or on the Survey that would make Seller unable to give title to the Premises as stipulated herein, including without limitation, any title or survey matters that are objectionable to Buyer (referred to herein as "<u>Defects of Title</u>", or individually, a "<u>Defect of Title</u>"). Any such written notice is referred to herein as a "<u>Buyer's Title Notice</u>". Seller shall have five (5) business days following its receipt of Buyer's Title Notice (a "<u>Seller's Title Notice</u>"). If Seller notifies Buyer in its Seller's Title Notice that Seller has elected not to cure the matter or matters objected to by Buyer in such Buyer's Title Notice, or if, having elected to attempt to cure a Defect of Title, Seller is unable to do so within the Cure Period as defined below, after the exercise of good faith efforts, Buyer shall elect, within five (5) business days following its receipt of Seller's Title Notice, or the end of the Cure Period, as the case may be, to either (a) waive its objection or objections to the matter or matters specified in Buyer's Title Notice and not being cured by Seller, in which case such matter or matters shall become Permitted Exceptions; or (b) terminate this Agreement by written notice to Seller, in which case the Deposit shall be returned forthwith to Buyer by the Escrow Agent without any further required action by either Buyer or Seller and neither party shall have any further liability or obligation to the other hereunder, unless such Defect of Title arises as a result of Seller's default hereunder, and except for any matter that expressl

any objection to any Defect of Title unless Buyer notifies Seller of such Defect of Title on or before the end of the Inspection Period, as the same may be extended as provided in Section 4.1. Accordingly, any Defect of Title not timely objected to by Buyer shall be deemed to be a "Permitted Exception". In all events Seller shall be required to remove at or prior to Closing (X) any mortgages affecting the Property and any other liens encumbering the Property that secure monetary obligations (including, without limitation, mechanic's liens) to the extent such liens are voluntarily created by Seller or arise out of any work performed by or on behalf of Seller (other than documents evidencing and securing the Assumed Loan), and (Y) any tax liens or amounts affecting the Property being conveyed due for taxes, water or sewer for any period of time prior to Closing. With respect to any monetary liens that are not included within clauses (X) or (Y) above, Seller agrees to cause such liens to be released from the Property by posting a bond or otherwise.

10

5.4 <u>Title Policy.</u> Buyer may obtain, at its sole cost and expense, an Owner's Title Insurance policy in the amount of the Purchase Price insuring fee simple title to the Real Property, subject only to the Permitted Exceptions (a "<u>Title Policy</u>"). Buyer may, at its election, request (i) extended coverage endorsement over general exceptions, with the standard survey exception limited to matters shown on the surveys, and (ii) other title endorsements (collectively, "<u>Buyer Requested Title Endorsements</u>"). Seller agrees to cooperate reasonably with Buyer and the title company in connection with issuing the Title Policy and Buyer Requested Title Endorsements and to assist in satisfying the requirements to the issuance of the Title Policy, to the extent reasonable and customary for sellers of commercial property. Such cooperation shall include, but not be limited to, Seller providing evidence of authority, copies of organizational documents, an affidavit or affidavits necessary to satisfy the reasonable and customary requirements of the title company in deleting the standard printed exceptions and in issuing the Buyer Requested Title Endorsements, provided that the form and content of such affidavits are reasonably acceptable to Seller and provided the same do not materially increase the obligations or decrease the rights of Seller contempated by this Agreement.

6. AS IS PURCHASE; MAINTENANCE OF PROPERTY; INTERIM COVENANTS.

6.1 <u>AS IS PURCHASE</u>. Except as expressly set forth in this Agreement (including, without limitation, Article 3), Buyer acknowledges and agrees that Buyer is acquiring the Property in its "AS IS" condition, WITH ALL FAULTS, AND WITHOUT ANY WARRANTY, EXPRESS, IMPLIED OR STATUTORY, all of which are hereby waived and disclaimed by Buyer; provided, however, that such waiver and disclaimer shall not, and do not, relieve Seller from any liability for breach of any representation or warranty set forth herein. Other than as expressly set forth herein, neither Seller nor any agents, representatives, or employees of Seller have made any representations or warranties, direct or indirect, oral or written, express or implied, to Buyer or any agents, representatives, or employees of Buyer with respect to the condition of the Property, its fitness for any particular purpose, or its compliance with any laws. Seller and Buyer acknowledge that the Inspection Period is intended to provide Buyer the opportunity to make such inspections (or have such inspections made by consultants) as it desires of the Property and all facts relevant to its use, including, without limitation, the interior, exterior, and structure of all Improvements, and the condition of soils and subsurfaces. Buyer acknowledges that it is relying on its own investigation of the Property and not on any information provided or to be provided by Seller except as expressly set forth in this Agreement, and agrees to accept the property at the Closing and waive all objections or claims against Seller (including, without limitation, any right or claim of contribution) arising from or related to the Property or to any Hazardous Materials on the Property except as expressly set forth in this Agreement. The provisions of this Section 6.1 shall survive the Closing.

11

- 6.2 Maintenance of Property. From and after the date of this Agreement through the Closing, the Property will be operated and managed on behalf of Seller in the normal course of business, in a manner consistent with the way the Property is presently being operated and managed and Seller shall keep the Property and Improvements in substantially the same condition and repair as it is in on the date hereof, reasonable wear and tear, damage by casualty and actions by Buyer excepted. During the term of this Agreement, Seller shall not enter into any new leases or modify the Leases without the prior written approval of Buyer. In addition, Seller shall maintain property and liability insurance on the Property with such coverages and in such amounts as are presently maintained by Seller on the date hereof. Subject to the adjustments prescribed in Article 9 hereof, Seller will, in the ordinary course of business, cause to be paid all trade accounts and costs and expenses of operation and maintenance of the Property incurred or attributable to a period prior to the Closing Date. Seller will pay and discharge, in the ordinary course of business, but in any event prior to delinquency, all ownership, leasing, operating, management and maintenance fees, costs and expenses incurred with respect to periods prior to the Closing, including without limitation, all amounts due and owing under the Service Contracts. Seller shall not take any affirmative action that would cause any Defect of Title, cause the Property not to conform with the provisions of this Agreement, would cause any representations of Seller set forth in this Agreement to be untrue or incorrect in any material respect or would otherwise cause Seller to be unable to perform its obligations under this Agreement.
- 6.3 <u>Performance of Assumed Loan Documents</u>. From and after the Effective Date of this Agreement through the Closing, Seller shall (a) continue to perform all obligations and to make all required payments in the manner and at the times specified in the Assumed Loan Documents; and (b) use its best efforts to prevent from occurring any event that with notice or the passage of time, or both, would constitute a default under the Assumed Loan Documents.
 - 6.4 No Transfers. Seller shall not sell, transfer, encumber or convey or mortgage the Property, or any part thereof, or interest therein, without the written consent of the Buyer.
- 6.5 <u>Leases</u>. Seller shall continue to observe and perform, within any applicable cure period, all of the obligations imposed upon Seller pursuant to the Leases from the Effective Date through the Closing Date. Seller will not modify any of the Leases from the Effective Date through the Closing without the prior written approval of Buyer, which approval may be withheld in Buyer's sole discretion.

12

6.6 Reasonable Good Faith Efforts; Further Assurances. Subject to the terms and conditions of this Agreement, each party will use commercially reasonable efforts to take, or cause to be taken, all action and to do or cause to be done, all things necessary or desirable under applicable law to consummate the transactions contemplated by this Agreement and to satisfy the conditions to Closing set forth herein. The parties further agree to execute and deliver such other documents, certificates, agreements and other writings and to take such other actions as may be necessary or desirable in order to consummate or implement expeditiously the transactions contemplated by this Agreement, provided the same are customary and do not materially increase the obligations or decrease the rights of the parties contemplated by this Agreement.

6.7 [Intentionally Omitted.]

6.8 Cross Indemnification.

6.8.1 Seller shall defend, indemnify and hold harmless Buyer from all losses, costs, damages, liabilities, claims and expenses (including reasonable attorneys' fees), resulting from (a) any breach of Seller's representations or warranties set forth in this Agreement, including without limitation Seller's representations and warranties set forth in Section 3.1 or (b) any liabilities or alleged liabilities under Environmental Laws arising from the presence of Hazardous Materials on the Property prior to and as of the Closing Date or arising from acts or omissions of Seller related to the Property in violation of Environmental Laws, except that Seller shall have no responsibility for Hazardous Materials used, disposed of or released in, on or under the Property by Buyer (or its employees, contractors or invitees) in its capacity as tenant under the IDEXX Lease. Any claim for breach of Seller's representations set forth in Section 3.1 shall be subject to the provisions of Article 3.

6.8.2 Buyer shall defend, indemnify and hold harmless Seller from all losses, costs, damages, liabilities, claims and expenses (including reasonable attorneys' fees), damage and liability resulting from (a) any breach of Buyer's representations or warranties set forth in this Agreement, including without limitation Buyer's representations and warranties set forth in Section 13.1, or (b) any liabilities or alleged liabilities under Environmental Laws arising from the presence of Hazardous Materials on the Property prior to and as of the Closing Date that were used, disposed of or released in, on or under the Property by Buyer (or its employees, contractors or invitees) in its capacity as tenant under the IDEXX Lease.

6.8.3 Notwithstanding anything else contained herein to the contrary, (a) Claims by Buyer against Seller for indemnification under clause (a) of Section 6.8.1 of this Agreement shall be subject to the time limitation and the Liability Limits set forth in Article 3, (b) claims by Buyer for indemnification under clause (b) of Section 6.8.1 of this Agreement shall be subject to the Liability Limits set forth in Article 3 and must be delivered to Seller on or before the date that is twelve (12) months after the Closing Date, (c) claims by Seller against Buyer for indemnification under clause (a) of Section 6.8.2 of this Agreement shall be subject to the time limitations and the limitations on liability set forth in Sections 13.2 and 13.3, and (d) claims by Seller for indemnification under clause (b) of Section 6.8.2 of this Agreement shall be subject to the limitations on liability set forth in Section 13.3 (but not the time limits set forth in Section 13.2).

13

6.8.4 If either party receives notice of a claim (the "Claim") against it that such party believes is subject to the indemnification provisions of this Section 6.8, the party seeking indemnification (the "Indemnitee") shall promptly inform the indemnifying party (the "Indemnitor") of the Claim as soon as reasonably practicable after the Indemnitee receives notice thereof. Once the Indemnitor has received notice of the Claim, it must promptly either (i) acknowledge, in writing, its obligation to defend and indemnify Indemnitee against the Claim or (ii) explain in writing and in reasonable detail the reasons why it believes it is not required to defend and indemnify Indemnitee. Provided that Indemnitor has acknowledged its obligation to defend and indemnify Indemnitee, (i) Indemnitor shall have sole control and authority with respect to the defense or settlement thereof; provided, however, that Indemnitor shall not enter into any settlement that obligates Indemnitee to take any action or incur any expense or which otherwise prejudices Indemnitee without the prior written consent of the Indemnitee (which shall not be unreasonably withheld, conditioned or delayed), and further provided that Indemnitee shall have the right to be represented separately by counsel of its own choosing, at its own expense, in connection with any such claim; and (ii) Indemnitee shall cooperate with Indemnitor, at Indemnitor's expense, in a reasonable way to facilitate settlement or defense. If Indemnitor does not acknowledge its obligation to defend and indemnify Indemnitee against the Claim, Indemnitee shall reasonable way to facilitate settlement or defense. If Indemnitor does not acknowledge its obligation to defend and indemnify Indemnitee against the Claim, Indemnitor indemnities sole control and authority with respect to the defense or settlement thereof without prejudice to its ability to seek indemnification from Indemnitor under this Section 6.8. The parties agree and acknowledge that if an Indemnitor's obligation to indemnify I

6.8.5 In the event an Indemnifying Party pays an Indemnified Party's losses pursuant to this Section, such Indemnifying Party shall be subrogated to the rights of the Indemnified Party, up to the amount of such payment, has against any insurer or other third party with respect thereto (and, upon the reasonable request of the Indemnifying Party, the Indemnified Party shall take appropriate actions necessary to transfer and assign such rights to the Indemnifying Party).

6.8.6 The provisions of this Section 6.8 shall survive the Closing hereunder and shall terminate on the date on which all indemnity obligations under this Section 6.8 have been satisfied.

14

6.9 Application to DEP.

6.9.1 Subject to the provisions of this Section 6.9, Buyer reserves the right, at its own expense, prior to March 16, 2006, to apply (an "Application") for a No Further Action Assurance and/or Certificate of Completion from the Maine Department of Environmental Protection ("DEP") Voluntary Response Action Program ("VRAP") under 38 M.R.S.A. §343-E, which may supply to Buyer certain liability protections allowed by law. Such application would involve supplying information, data and environmental sampling results with the DEP. Subject to the provisions of this Section 6.9, Seller agrees reasonably to cooperate with Buyer in connection with any such application and related process, and reasonably to assist in supplying any existing information, data and sampling results. Upon request, Buyer will add Seller as a coapplicant to obtain a No Further Action Assurance and/or Certificate of Completion. In the event that Buyer seeks to submit to DEP an Application prior to Closing, Buyer shall provide a draft of the Application and all supporting documentation (including, without limitation, all environmental data generated or gathered by Buyer that Buyer wishes to submit to DEP as part of the Application process) to Seller not less than three (3) business days prior to its submission to the DEP or any other governmental agency. Seller shall have the right, within such three (3) business day period, to disapprove of the draft Application but only if Seller reasonably concludes that the Application is likely to lead the DEP to conclude there is a current significant threat to health or the environment. If Seller so disapproves, then, within five (5) days after the date of Buyer's receipt of written notice of Seller's disapproval, Buyer may provide Seller with written notice terminating this Agreement. Upon such termination, the Deposit shall be returned to Buyer by the Escrow Agent without any further required action by Buyer or Seller and neither party shall have any further liability or obligation to the other hereunder e

6.9.2 If, as set forth above, Buyer applies for a No Further Action Assurance and/or Certificate of Completion prior to March 16, 2006, then Buyer shall diligently pursue and attempt to obtain a No Further Action Assurance or Certificate of Completion from DEP by no later than May 1, 2006 and the Closing Date shall automatically be extended to May 5, 2006. If Seller has chosen to be a co-applicant, then Buyer will (i) use reasonable efforts to provide Seller with the opportunity to participate in all material verbal and written communications with DEP concerning the Application and/or VRAP process; and (ii) promptly share with Seller all correspondence from DEP concerning the Application and/or VRAP process including, without limitation, any draft form of No Further Action Assurance or Certificate of Completion.

6.9.3 If, prior to May 1, 2006, Buyer receives from DEP a No Further Action Assurance or Certificate of Completion that is acceptable to Buyer, in its sole discretion, then Buyer shall, subject to the other terms and conditions of this Agreement, proceed to Closing. If, prior to May 1, 2006, Buyer has not received a No Further Action Assurance or Certificate of Completion from DEP that is acceptable to Buyer, in its sole discretion, despite the diligent effort of the parties, then Buyer shall have the right to terminate this Agreement until 5:00 p.m. on May 5, 2006, by providing written notice to Seller and Escrow Agent that Buyer is terminating this Agreement. Upon such termination, the Deposit shall be returned to Buyer by the Escrow Agent without any further required action by Buyer or Seller and neither party shall have any further liability or obligation to the other hereunder except for any provision that expressly survives the Closing or the termination of this Agreement. If Buyer does not so terminate, then Buyer shall, subject to the other provisions of this Agreement, proceed to Closing.

6.9.4 If DEP issues a No Further Action Assurance and/or Certificate of Completion prior to Closing, Seller, in its sole discretion, shall decide at Closing whether such instrument shall be recorded before or after the deed from Seller to Buyer is recorded. Seller acknowledges that the VRAP No Further Action Assurance and/or Certificate of Completion may provide that liability is conditioned on recording of such documents and on adopting and recording activity and use limitations in the form of an Environmental Covenant pursuant to the Uniform Environmental Covenants Act, 38 M.R.S.A. § 3001 et seq., which shall be drafted by Buyer. In any event, Seller shall use best efforts prior to Closing to obtain any subordination agreement(s) from Lender that DEP or Buyer deem necessary to ensure any prior recorded interest of Lender that may impair or threaten the effectiveness of the Environmental Covenant will be subordinated to the Environmental Covenant. If such subordination agreement(s) should not have been obtained from Lender notwithstanding Seller's best efforts by May 1, 2006, then the Closing shall be extended to May 20, 2006 to afford Seller the opportunity to obtain such subordination agreement. If such subordination agreement is not obtained by May 20, 2006, Buyer shall have the right to terminate this Agreement by providing written notice to Seller that Buyer is terminating this Agreement. Upon such termination, the Deposit shall be returned to Buyer by the Escrow Agent without any further required action by Buyer or Seller and neither party shall have any further liability or obligation to the other hereunder except for any provision that expressly survives the Closing or the termination of this Agreement. If Buyer does not so terminate despite such subordination agreement(s) not having been obtained from Lender, then Buyer shall, subject to the other terms and conditions of this Agreement (including the other provisions of this Section 6.9), proceed to Closing.

7. CONTINGENCIES AND CONDITIONS TO CLOSING.

- 7.1 Without limiting any other conditions to Buyer's obligations to close set forth in this Agreement, the obligations of Buyer under this Agreement are subject to the satisfaction at the time of Closing (except as expressly provided below) of each of the following conditions (any of which may be waived in whole or in part by Buyer in writing at or prior to Closing):
 - 7.1.1 All of the representations and warranties by Seller set forth in this Agreement, including without limitation Section 3.1 hereof, shall be true and correct in all material respects as of the Closing Date (other than changes thereto which result from actions or conditions caused by the tenant under the IDEXX Lease).

16

- 7.1.2 Seller shall have delivered or caused to be delivered to the Buyer all documents required to be delivered or caused to be delivered by the Seller pursuant to Article 8 hereof.
- 7.1.3 Lender under the Assumed Loan shall have consented to the assumption by Buyer of the Assumed Loan and Buyer and such lender shall have entered into satisfactory agreements providing for such assumption and the modification of the Assumed Loan Documents as provided in Section 2.3 above, and shall have provided such estoppel information with respect to the Assumed Loan as Buyer shall have reasonably requested, all on terms reasonably acceptable to Buyer.
- 7.1.4 No order, writ, injunction, moratorium, referendum or decree shall have been entered and be in effect or be threatened or pending and no statute, rule, regulation or other requirement shall have been promulgated or enacted and be in effect, that restrains, enjoins, invalidates or prohibits the transactions contemplated hereby or the current use of the Real Property or the expansion of Buyer's operations into 100% of the Improvements.
- 7.1.5 No suit or other proceeding shall be pending or threatened by any third party not affiliated with or acting at the request of Buyer before any court or governmental authority seeking to restrain or prohibit or declare illegal, or seeking substantial damages against Buyer in connection with the transactions contemplated by this Agreement or the current use of the Real Property or the expansion of Buyer's operations into 100% of the Improvements.
- 7.1.6 If it is determined that Seller lacks any permits required by applicable law or regulation, and if Buyer gives Seller written notice of the same at or prior to the end of the Inspection Period, as the same may be extended as provided in Section 4.1, then in such events, Seller and Buyer shall cooperate in obtaining or transferring such permit or permits to Buyer and if such permits are not obtained or transferred, Buyer shall have the right to terminate this Agreement by giving written notice at or prior to Closing.
- 7.2 Without limiting any other condition to Seller's obligations to close set forth in this Agreement, the obligations of Seller under this Agreement are subject to the satisfaction at the time of Closing of each of the following conditions (any of which may be waived in whole or in part by Seller in writing at or prior to Closing):
 - 7.2.1 All of the representations and warranties by Buyer set forth in this Agreement shall be true and correct in all material respects.
 - 7.2.2 Buyer shall have delivered or caused to be delivered to Seller all documents required to be delivered or caused to be delivered by Buyer pursuant to the provisions of Article 7 hereof.
 - 7.2.3 Lender under the Assumed Loan shall have consented to the assumption by Buyer of the Assumed Loan and Buyer and such Lender shall have entered into satisfactory agreements providing for such assumption and the modification of the Assumed Loan Documents as provided in Section 2.3 above, all on terms reasonably acceptable to Seller.

17

- 7.3 If Buyer applies for VRAP liability protections, then Buyer must have received from the DEP, a VRAP No Further Action Assurance and/or Certificate of Completion, in form and substance reasonably acceptable to Buyer, which Seller must have recorded at its expense contemporaneously with the Closing (and if the VRAP No Further Action Assurance and/or Certificate of Completion stipulates as a condition that activity and use limitations must be recorded as an Environmental Covenant, then the Environmental Covenant must be executed and recorded by Seller, along with a subordination agreement from the holder of the Assumed Loan, contemporaneously with the Closing), all as contemplated in Section 6.9.
- 7.4 Buyer shall have received economic incentives from the City of Westbrook and the State of Maine in an amount and upon such terms and conditions as are acceptable to Buyer. This contingency shall be deemed to be satisfied unless Buyer provides written notice of dissatisfaction on or before March 15, 2006.
 - 7.5 The absence of any default by IDEXX Laboratories, Inc., as tenant under the IDEXX Lease shall not be a condition to Buyer's obligation to close hereunder.
 - CLOSING.

- 8.1 The closing (the "Closing") under this Agreement shall occur at 10:00 a.m. on the later of April 1, 2006 or fifteen days after the end of the Inspection Period, as the same may be extended as provided in Section 4.1 above (the "Closing Date"), at the offices of Pierce Atwood LLP, One Monument Square, Portland, Maine, or, at the election of either Seller or Buyer by written notice to the other at least five (5) business days prior to the Closing Date, by delivery of all documents, funds and other material in escrow to the Escrow Agent pursuant to closing instructions consistent with this Agreement and otherwise reasonably satisfactory to Seller and Buyer. The Closing may be extended as provided in Sections 2.3, 5.3 and 6.9 above. Provided all conditions to Closing have been satisfied, Buyer shall have the right to accelerate the Closing Date by giving Seller at least seven (7) days prior written notice.
 - 8.2 At Closing, Seller shall deliver to Buyer or the Escrow Agent, as applicable, the following:
 - 8.2.1 A Deed (the "<u>Deed</u>") substantially in the form attached hereto and made a part hereof as <u>Exhibit L</u>running to the Buyer or to such nominee of the Buyer as Buyer may designate in a written notice to Seller delivered to Seller at least three (3) business days prior to the Closing, conveying fee simple title to the Real Property, free from encumbrances except for Permitted Exceptions.
 - 8.2.2 A Bill of Sale and General Assignment and Assumption Agreement for the Personalty, substantially in the form attached hereto and made a part hereof as Exhibit J (the "Bill of Sale").
 - 8.2.3 An Assignment and Assumption Agreement substantially in the form attached hereto and made a part hereof as Exhibit K assigning to Buyer all right, title and interest of Seller in the Leases and in the Assumed Contracts (the "Assignment and Assumption"). Such assignment shall include an assignment, in common with Seller, of the rights of the landlord under the Sanmina Lease, as described in Section 1.3 above. In addition, Seller shall assign to Buyer, any rights, in common with Seller, that Seller may have against Data General, its immediate predecessor in title, with respect to any Hazardous Materials on the Real Property or violation of any Environmental Laws affecting the Real Property.

18

- 8.2.4 Written notices (the "Sale Notices"), signed by Seller and Buyer, addressed to each contractor to an Assumed Contract, indicating that the Property has been sold to Buyer and that all rights of Seller thereunder and in any deposits and any interest accrued thereon have been assigned to Buyer.
- 8.2.5 Standard affidavits regarding mechanics' liens and parties-in-possession addressed to the Buyer's title insurance company. Seller shall not be obligated under the terms of this Agreement to indemnify the title company for any matter affecting title to the Property (other than as set forth in the customary form of Seller's affidavit regarding parties in possession and mechanics' liens).
- 8.2.6 A certification and affidavit as required by the Foreign Investors Real Property Tax Act, substantially in the form attached hereto and made a part hereof as Exhibit L.
 - 8.2.7 A closing and proration statement reflecting all adjustments to the Purchase Price contemplated by this Agreement (the "Closing Statement"), signed by Seller.
 - 8.2.8 A Certificate as to Information Required Under Internal Revenue Code Section 6045, in the form required under Section 6045 of the Internal Revenue Code.
- 8.2.9 A State of Maine Residency Affidavit, or in the absence of such affidavit, Buyer shall be entitled to withhold a portion of the Purchase Price in accordance with 36 M.R.S.A. §5250-A.
 - 8.2.10 Documents and materials required in connection with the assignment and assumption of the Assumed Loan.
- 8.2.11 Evidence reasonably satisfactory to Buyer and the title company that the person executing the Closing documents on behalf of Seller has full right, power and authority to do so.
- 8.2.12 A certificate indicating that the representations and warranties of Seller set forth in this Agreement are true and correct in all material respects on the Closing Date, or, if there have been changes, describing such changes.
 - 8.2.13 All other documents reasonably required to effectuate this Agreement and the transaction contemplated hereby.

19

- 8.2.14 The following items, to the extent in the possession of Seller: (i) all keys for all entrance door and spaces which may be locked (whether occupied or not) in the Improvements; (ii) all original books, records, tenant files, operating reports, plans and specifications and other materials reasonably necessary to the continuity of operation of the Real Property; and (iii) the originals (or copies where originals are not available) of the Leases and tenant correspondence, and all other documentation that Buyer reasonably identifies that should be transferred to Buyer at Closing.
- 8.3 At Closing, Buyer shall deliver to Seller or the Escrow Agent, as applicable, the following:
 - 8.3.1 The balance of the Purchase Price in accordance with the provisions of Section 2.2 hereof.
 - 8.3.2 A duplicate original of the Bill of Sale in the form attached hereto as Exhibit J.
 - 8.3.3 A duplicate original of the Assignment and Assumption of Leases in the form attached hereto as <u>Exhibit K</u>.
 - 8.3.4 The Closing Statement, signed by Buyer.
 - 8.3.5 The Sale Notices, signed by Buyer.
 - 8.3.6 Documents and materials required in connection with the assignment and assumption of the Assumed Loan.

- 8.3.7 Evidence reasonably satisfactory to Seller and the title company that the person executing the Closing documents on behalf of Buyer has full right, power and authority to do so.
- 8.3.8 A certificate indicating that the representations and warranties of Buyer set forth in this Agreement are true and correct in all material respects on the Closing Date, or, if there have been changes, describing such changes.
 - 8.3.9 All other documents reasonably required to effectuate this Agreement and the transaction contemplated hereby.
- 8.4 If Buyer takes title to the Property, the IDEXX Lease shall terminate at the Closing with the same force and effect as if the Closing Date were the scheduled termination date of the IDEXX Lease. If Buyer elects to designate another entity to take title to the Property, or otherwise assigns this Agreement in accordance with Section 17.1, Seller shall assign the IDEXX Lease to such designee or assignee. Effective upon the Closing of the transaction contemplated hereby, and except as set forth in Section 6.8 above, the parties do hereby remise, release and forever discharge each other and their respective employees, affiliates and managers of and from any and all claims, liabilities and obligations arising under or in connection with the Lease between Seller, as lessor, and Buyer, as lessee, as described on Exhibit B (the "IDEXX Lease"), or the operation of Buyer's business at the premises demised thereunder.

20

APPORTIONMENTS; TAXES; EXPENSES.

9.1 Apportionments.

- 9.1.1 <u>Taxes</u>. To the extent not covered by Section 9.1.2, applicable real estate and personal property taxes for the then current tax period shall be apportioned as of the Closing and the net amounts shall be added to or deducted from, as the case may be, the Purchase Price. The term "<u>real estate taxes</u>" shall include any installments of betterment or similar assessments; provided, however, that Seller shall be obligated to pay any special assessments against the Property for public improvements installed prior to the Effective Date. If the amount of such taxes is not known at the Closing, such taxes shall be apportioned on the basis of the taxes assessed for the preceding tax period, with a reapportionment as soon as the new tax rate and valuation can be ascertained and if the taxes which are to be apportioned shall thereafter be reduced by abatement, the amount of such abatement, less the reasonable cost of obtaining the same, shall be apportioned between the parties. Neither party shall be obligated to commence or prosecute abatement proceedings.
- 9.1.2 <u>Rents, Security Deposits, etc.</u> The parties acknowledge that the Sanmina Lease will have expired prior to the Closing Date and no adjustments or prorations shall be required with respect to such lease. Seller shall be responsible for any post-Closing obligations under the Sanmina Lease. With respect to the IDEXX Lease, Seller and Buyer shall reconcile any rent and any obligations for operating expenses as of the Closing or otherwise paid or owing under the IDEXX Lease, including any estimated payments, with an appropriate adjustment to the Purchase Price. Buyer shall receive a credit at Closing for the full amount of the security deposit paid under the IDEXX Lease as set forth on <u>Exhibit D</u>, plus any additional interest on such Security Deposit accruing from December 31, 2005 until the Closing Date.
- 9.1.3 Operating Expenses. To the extent not already adjusted pursuant to Section 9.1.3 above, all electricity, water, gas, sewage and other utility expenses applicable to the Property shall be prorated between Seller and Buyer as of the Closing based on final readings therefor as of the Closing. All maintenance, management, and other operating expenses applicable to the Property and payments under any Assumed Contracts shall be prorated between Seller and Buyer as of the Closing based on estimates of the amounts that will be due and payable on the next payment date, unless final invoices therefor as of the Closing shall have been obtained, in which case such final invoices shall be utilized as the basis for adjustment. To the extent not already adjusted pursuant to Section 9.1.3 above, as soon as the amount of such expenses for the period shall be known, Seller and Buyer shall recalculate the foregoing adjustments with the result that Seller shall pay for those expenses attributable to the period of time prior to the Closing Date and Buyer shall pay for those expenses attributable to the period of time commencing with the Closing Date. Any and all deposits held by utility companies or with other providers of services to the Property shall remain the property of Seller and be returned to Seller by such companies and providers except to the extent that Buyer elects to credit to Seller the amount of any such deposits, in which event such deposits shall be assigned to Buyer.

21

9.2 Expenses.

- 9.2.1 <u>Seller's Expenses</u>. Seller shall pay one-half (½) of the deed transfer tax incurred in connection with the transaction contemplated by this Agreement as well as its own expenses incurred in connection with this Agreement, and except as provided in <u>Sections 11.2 and 17.11</u> below, its legal fees. Seller shall pay any assumption fee and other associated fees and expenses (including legal fees) charged by Lender or Servicer, as provided in Section 2.3 above.
- 9.2.2 <u>Buyer's Expenses</u>. Buyer shall pay one-half (½) of the deed transfer tax incurred in connection with the transaction contemplated by this Agreement as well as its own expenses incurred in connection with this Agreement, including, without limitation, (1) all recording charges incident to the recording of the deed for the Real Property, (2) all premiums for Buyer's title insurance policy, (3) the cost of any survey obtained by Buyer, and (4) except as provided in <u>Sections 11.2 and 17.11</u> below, its legal fees.
- 9.2.3 Other Expenses. Except as herein specifically provided, all income and expense attributable to the operation of the Real Property shall be apportioned between Seller and Buyer as of the Closing Date. Except as herein specifically provided, Seller and Buyer shall allocate all closing costs between them in accordance with standard practice in Westbrook, Maine.
- 9.3 Withholding. Unless Seller is a resident of the State of Maine and delivers to Buyer a Maine Bureau of Taxation Form REW-2, Affidavit of Residency, Buyer shall withhold a portion of the Purchase Price in accordance with 36 M.R.S.A. §5250-A and shall pay the amount so withheld to the appropriate governmental authorities at the time of Closing.
- 9.4 Errors; Survival. During the one (1) year period following the Closing, Buyer and Seller agree to re-prorate the closing adjustments under this Article 9 in order to correct any error in performing the prorations or to reflect information that becomes available during such one (1) year period indicating that the prorations performed at Closing were not accurate. The provisions of this Article 9 shall survive the Closing.
 - 9.5 Date of Settlement. For purposes of calculating adjustments and expenses under this Article 9, Buyer shall be deemed to be the owner of the Property on the Closing Date.

10.1 If all or substantially all of the Property is taken by eminent domain after the date hereof and prior to the Closing, this Agreement shall terminate and the Deposit shall be returned to Buyer. If a taking by eminent domain of a material part of the Property (as hereinafter defined) shall be made after the date hereof and prior to the Closing then Seller shall promptly notify Buyer of the same (the "Taking Notice"), and Buyer shall have the right to terminate this Agreement by notice from Buyer to Seller and Escrow Agent given before the date that is the earlier to occur of (a) ten (10) days after the date of the Taking Notice and (b) the Closing. In the event Buyer does not terminate this Agreement, Buyer shall accept such title to the Property as Seller can deliver, in which case Seller shall pay over or assign to Buyer all rights and proceeds arising by reason of such taking (less any collection costs incurred by Seller in connection therewith and any costs and expenses incurred by Seller to restore the Property) and Buyer shall pay the Purchase Price without reduction. If this Agreement so terminates, then the Deposit shall be paid to Buyer. For purposes of this Section, a taking by eminent domain of a material part of the Property shall be defined as any taking which is of a nature which would give the tenant under the IDEXX Lease the right to terminate the IDEXX Lease pursuant to the terms thereof.

22

- 10.2 If any time after the date hereof and prior to the Closing any portion of the Improvements is destroyed or damaged as a result of fire or any casualty, Seller shall promptly give notice thereof to Buyer. The rights and obligations of the parties by reason of such destruction or damage shall be as follows:
 - 10.2.1 If the "Cost of Repair and Restoration" (as hereinafter defined) of such destruction or damage shall be <u>Five Hundred Thousand Dollars</u> (\$500,000.00) or less, the obligations of the parties hereunder shall not be affected by such destruction or damage and Buyer shall accept title to the Property in its destroyed or damaged condition. Buyer shall pay the Purchase Price without reduction, and Seller shall pay over or assign to Buyer without recourse all rights to any proceeds of insurance paid or payable with respect to such destruction or damage (less any collection costs incurred by Seller in connection therewith and any costs and expenses incurred by Seller to restore the Property) and Buyer shall have a credit against the Purchase Price in the amount of any deductible.
 - 10.2.2 If the Cost of Repair and Restoration of such destruction or damage shall exceed Five Hundred Thousand Dollars (\$500,000.00), Buyer shall have the right to terminate this Agreement by notice from Buyer to Seller given before the date that is the earlier to occur of (a) ten (10) days after the date of delivery of the Cost Notice (as hereinafter defined) and (b) the Closing. In the event Buyer does not terminate this Agreement, Buyer shall accept title to the Property in its destroyed or damaged condition in accordance with and subject to the provisions of subparagraph 10.2.1 above. If Buyer so terminates this Agreement, then the Deposit shall be paid to Buyer.
 - 10.2.3 The term "Cost of Repair and Restoration" shall mean the amount of Buyer's good faith estimate of the actual cost of repair and restoration to restore the Property to the condition it was in immediately prior to the casualty. Buyer shall send Seller notice of the Cost of Repair and Restoration (the "Cost Notice") promptly after making the aforesaid estimate.

REMEDIES.

11.1 Seller's Remedies. If Seller shall have fulfilled all of its obligations under this Agreement in the time and manner specified herein (as such timeframes may be extended in accordance with the terms hereof) and Buyer breaches its obligation to close or fails to perform its material obligations under this Agreement, (it being agreed that a failure of a condition to Closing that is not caused by a default by Buyer shall not be construed as a breach by Buyer of its obligations to close or failure to perform its material obligations), then Seller shall, as its sole remedy therefor, be entitled to receive the entire Deposit as liquidated damages (and not as a penalty) in lieu of, and as full compensation for, all other rights or claims of Seller against Buyer by reason of such default. Thereupon this Agreement shall terminate and the parties shall be relieved of all further obligations and liabilities hereunder, except as expressly set forth herein. Buyer and Seller acknowledge that the damages to Seller resulting from Buyer's breach would be difficult, if not impossible, to ascertain with any accuracy, and that the liquidated damage amount set forth in this Section represents both parties' efforts to approximate such potential damages. This provision shall not limit Buyer's obligations under any indemnification provision contained in this Agreement.

23

- 11.2 Buyer's Remedies. If Buyer shall have fulfilled all of its obligations under this Agreement in the time and manner specified herein (as such timeframes may be extended in accordance with the terms hereof) and Seller breaches any of its obligations to close or fails to perform its material obligations under this Agreement, (it being agreed that a failure of a condition to Closing that is not caused by a default by Seller shall not be construed as a breach by Seller of its obligations to close or failure to perform its material obligations), then Buyer shall have, as its sole remedy therefor, the right either (a) to waive the breach or default and proceed to Closing in accordance with the provisions of this Agreement without adjustment of the Purchase Price; or (b) to terminate this Agreement by written notice to Seller, in which event the entire Deposit shall be returned to Buyer and Buyer shall be entitled to seek damages in an amount equal to all third party costs and expenses, including attorneys fees, incurred by Buyer in connection with the proposed transaction, including without limitation negotiating this Agreement and due diligence and other activities and expenses undertaken and incurred in connection with this Agreement, or if Seller's default is intentional or willful, Buyer shall be entitled to recover all damages incurred by Buyer as a result of Seller's default, provided that all events, Seller's liability for damages under this clause (b) shall not exceed \$600,000; or (c) to bring an action for specific performance of this Agreement and the obligations of Seller hereunder. If Buyer brings an action for specific performance, the prevailing party shall be entitled to recover reasonable attorneys' fees, as provided in Section 17.11.
- 11.3 <u>Remedies Exclusive</u>. By the express agreement of Buyer and Seller, the remedies set forth in this Article constitute the sole remedies at law or in equity available to Buyer and Seller, as the case may be, on account of the other party's breach of its obligations under this Agreement; provided, however, to the extent any terms or provisions hereof are specifically intended to survive the Closing and delivery of the Deed or termination of this Agreement, the other party shall have all remedies with respect thereto as may be available at law or in equity, subject to any limitations set forth in this Agreement or in any document or instrument delivered pursuant hereto. In no event, however, shall either party hereto be liable for any consequential, special, indirect or punitive damages.

24

12. <u>FURTHER ASSURANCES</u>. After the Closing, Seller and Buyer each agrees to perform such other acts, and to execute, acknowledge and deliver, prior to, at or subsequent to Closing, such other instruments, documents and other materials as the other may reasonably request (at no cost to such party) and as shall be necessary in order to effect the consummation of the transactions contemplated hereby, including terminations and transfers of existing governmental permits and approvals held by Seller with respect to the Property. The provisions of this Article 12 shall survive the Closing.

13. REPRESENTATIONS OF BUYER.

- 13.1 Buyer represents and warrants as of the date hereof and as of the Closing that:
 - 13.1.1 <u>Authority</u>. Buyer is a corporation, duly organized, validly existing and in good standing under the laws of the State of Delaware and has all requisite power and authority to enter into this Agreement and all documents now or hereafter to be executed and delivered by Buyer pursuant to this Agreement (collectively, the "<u>Buyer's Documents</u>") and to perform its obligations hereunder and under Buyer's Documents.

- 13.1.2 No Conflict. The execution and delivery of this Agreement and the consummation of the transactions contemplated hereunder on the part of Buyer do not and will not violate any applicable law, ordinance, statute, rule, regulation, order, decree or judgment, conflict with or result in the breach of any material terms or provisions of, or constitute a default under, or result in the creation or imposition of any lien, charge, or encumbrance upon any of the property or assets of the Buyer by reason of the terms of any contract, mortgage, lien, lease, agreement, indenture, instrument or judgment to which Buyer is a party so as to prevent Buyer from performing its obligations hereunder
- 13.2 <u>Survival of Representations</u>. The representations and warranties of Buyer contained in this Agreement shall survive Closing and delivery of the Deed for the duration of the Survival Period and, with respect to any written claim in connection with the representations and warranties made within such Survival Period, until final unappealable adjudication or settlement thereof. Any claim must be delivered to Buyer on or before that end of the Survival Period, time being of the essence. Any claim under this Article 13 is subject to the provisions of <u>Section 6.8</u> herein.
- 13.3 <u>Limitations</u>. Notwithstanding anything herein to the contrary, Buyer shall in no event have any liability for breach of any representation, warranty, indemnity or covenant set forth in this Agreement or in any document, instrument or agreement delivered pursuant hereto (i) unless and until the aggregate amount of all such claims for breaches exceeds Twenty Thousand Dollars (\$20,000.00) in which event Seller shall be entitled to seek recovery of the full amount of its claim under this Section, or (ii) for any amounts in excess of One Million Dollars (\$1,000,000.00) in the aggregate. To the extent that prior to Closing Seller discovers any inaccuracy in a representation and warranty of Buyer in this Agreement and Seller or Buyer expressly notifies the other party prior to Closing of any such inaccuracy and notwithstanding such inaccuracy and notification, Seller proceeds to close this transaction, such representation and warranty shall be deemed modified to reflect the inaccuracy discovered by Seller.

25

14. <u>NOTICES</u>. All notices and other communications provided for herein shall be in writing and shall be sent to the address set forth below (or such other address as a party may hereafter designate for itself by notice to the other parties as required hereby) of the party from whom such notice or communication is intended:

If to Seller:

CW Westbrook Limited Partnership c/o S.R. Weiner and Associates, Inc. 1330 Boylston Street Chestnut Hill, Massachusetts 02467 Fax No.: 617-739-5945 Attention: Thomas DeSimone

With a copy to:

Goulston & Storrs, P.C. 400 Atlantic Avenue Boston, Massachusetts 02110 Fax No.: 617-574-7608 Attention: Phillip G. Levy, Esq.

If to Buyer:

IDEXX Laboratories, Inc. 80 Eisenhower Drive Westbrook, Maine 04092 Fax No.: (207) 856-0347 Attention: Director of Facilities

With a copy to:

Office of General Counsel IDEXX Laboratories, Inc. Westbrook, Maine 04092 Fax No.: (207) 856-0347

With a copy to:

Pierce Atwood LLP One Monument Square Portland, Maine 04101 Fax No.: (207) 791-1350 Attention: Dennis C. Keeler

If to Escrow Agent: To such address as is indicated in the Escrow Agreement

15. <u>BROKERS</u>. Buyer and Seller each represents to the other that it has not dealt with any broker or agent in connection with this transaction (other than Daniel Cordeau and John Duffy of Spaulding & Slye (together, the "<u>Seller's Broker</u>"). Seller shall be solely responsible for the payment of any brokerage commission due and payable to Seller's Broker. Each party hereby indemnifies and holds harmless the other party from all loss, cost and expenses (including reasonable attorneys' fees) arising out of a breach of its representation or undertaking set forth in this Article. The provisions of this Article shall survive Closing or the termination of this Agreement.

16. ESCROW AGREEMENT.

- 16.1 The Deposit shall be held in escrow by Escrow Agent in accordance with the following: the Deposit shall be deposited in an insured, interest bearing account at a national bank or trust company having an office in the State of Maine approved by Seller and Buyer. All interest on the Deposit shall be held as part of the Deposit and disbursed with the Deposit to the party entitled to receive the Deposit under this Agreement.
 - 16.2 Seller and Buyer shall each bear fifty percent (50%) of the costs and expenses of the Escrow Agent and shall pay the same upon billing therefor.
- 16.3 In the performance of its obligations, Escrow Agent shall be under no liability to Seller or Buyer except for its willful misconduct or gross negligence and each of Seller and Buyer agrees to indemnify and hold harmless Escrow Agent from any and all such liability except that arising out of Escrow Agent's willful misconduct or gross negligence, this obligation to survive the Closing and delivery of the deed or termination of this Agreement. If at any time Escrow Agent shall determine that it requires written authorization from Seller or Buyer prior to making any disbursement from the Deposit, it shall request such authorization and may withhold any such disbursement until such written authority has been received by it. In the event of any legal action involving the disbursement of the Deposit, as regards their obligation to Escrow Agent, Seller and Buyer shall be jointly and severally liable for the legal fees and expenses incurred by Escrow Agent in such action, but as regards to the obligations to each other, Seller shall be responsible for one-half (½) and Buyer for one-half (½) of such fees and expenses. However, in the event of any such legal action, a court may determine that such legal fees and expenses of Escrow Agent shall be borne by Seller or Buyer in different percentages, or even that the same shall be borne by either Seller or Buyer in their entirety without contribution by the other party.

27

16.4 Seller and Buyer acknowledge that any federal insurance for the Deposit is limited to a cumulative maximum amount of One Hundred Thousand Dollars (\$100,000.00) for each individual deposit or for all the depositor's accounts at the same or related institution and that certain banking instruments such as, but not limited to, repurchase agreements and letters of credit are not covered at all by federal insurance. Seller and Buyer acknowledge and agree that Escrow Agent shall have no obligation or liability with respect to insuring the Deposit or with respect to the solvency of the depository institution or otherwise with respect to the appropriateness of the depository institution for the purposes of the Deposit.

17. MISCELLANEOUS.

- 17.1 <u>Assignability</u>. Prior to Closing, Buyer may not assign or transfer all or any portion of its rights or obligations under this Agreement to any other individual, entity or other person without the consent thereto by Seller which may be withheld in Seller's sole discretion and any such unauthorized attempted assignment shall be null and void. Notwithstanding the foregoing, prior to Closing, Buyer shall have the right to assign its rights and obligations hereunder to any entity which is a wholly-owned subsidiary of Buyer or which is under common ownership with Buyer, provided Buyer provides Seller with prior written notice of such assignment and evidence of such affiliation satisfactory to Seller. No assignment or transfer by Buyer will release Buyer of its obligations hereunder. In the event of any assignment by Buyer permitted under this Section the assignee must agree in writing to be bound by all of the terms and conditions of this Agreement and the obligations and liabilities of Buyer hereunder.
- 17.2 Governing Law; Parties in Interest. This Agreement shall be governed by the law of the State of Maine without regard for conflict of law principles, and shall bind and inure to the benefit of the parties hereto and, subject to the provisions of Section 17.1, their respective heirs, executors, administrators, successors, and permitted assigns. The parties agree that any action brought with respect to this Agreement shall be brought in the State of Maine and the parties hereto hereby consent to such forum and venue and waive any argument or defense to such forum and venue including without limitation any defense based upon inconvenient forum, lack of jurisdiction or otherwise.
- 17.3 <u>Recording; Confidentiality.</u> This Agreement or any notice or memorandum hereof shall not be recorded in the Cumberland County Registry of Deeds. Seller acknowledges that Buyer is a public company and Seller consents to the disclosure of the terms and conditions of this transaction to the extent such disclosure is required to fulfill Buyer's disclosure requirements under applicable SEC regulations. Seller acknowledges that Buyer will need to disclose this Agreement in a filing on form 8-K and will need to append a copy of this Agreement with the filing of its next 10-K. Buyer will also need to disclose the existence of this Agreement in connection with assuming the Assumed Loan and in dealings with the City of Westbrook and the State of Maine in obtaining economic incentives.
 - 17.4 <u>Time of the Essence</u>. Time is of the essence of this Agreement.
- 17.5 <u>Headings</u>. The headings preceding the text of the paragraphs and subparagraphs hereof are inserted solely for convenience of reference and shall not constitute a part of this Agreement, nor shall they affect its meaning, construction or effect.

28

- 17.6 Counterparts. This Agreement may be executed simultaneously in counterparts, each of which shall be deemed an original, but all of which together shall constitute one and the same instrument.
- 17.7 Exhibits. All Exhibits which are referred to herein and which are attached hereto or bound separately and initialed by the parties are expressly made and constitute a part of this Agreement.

17.8 Like-Kind Exchange.

- 17.8.1 At Seller's election, Buyer shall cooperate with Seller in effecting a so-called Section 1031 like-kind exchange provided that (i) all costs and liabilities associated with such exchange shall be borne by Seller; (ii) such cooperation shall not require Buyer to take title to any real estate which is not the subject of this Agreement; (iii) such exchange shall not affect Seller's obligation to deliver title in accordance with the terms hereof; and (iv) Seller shall indemnify Buyer from and against any and all liability arising out of such cooperation (including reasonable attorneys' fees). The provisions of this paragraph shall survive the Closing hereunder.
- 17.8.2 At Buyer's election, Seller shall cooperate with Buyer in effecting a so-called Section 1031 like-kind exchange provided that (i) all costs and liabilities associated with such exchange shall be borne by Buyer; (ii) such cooperation shall not require Seller to take title to any real estate which is not the subject of this Agreement; (iii) such exchange shall not affect Buyer's obligation to close in accordance with the terms hereof; and (iv) Buyer shall indemnify Seller from and against any and all liability arising out of such cooperation (including reasonable attorneys' fees). The provisions of this paragraph shall survive the Closing hereunder.

- 17.9 Entire Agreement; Amendments. This Agreement and the Exhibits hereto set forth all of the covenants, representations, warranties, agreements, conditions and undertakings between the parties hereto with respect to the subject matter hereof, and supersede all prior and contemporaneous agreements and understandings, inducements or conditions, express or implied, oral or written. This Agreement may not be changed orally but only by an agreement in writing, duly executed by or on behalf of the party or parties against whom enforcement of any waiver, change, modification, consent or discharge is sought.
- 17.10 Interpretation. This Agreement shall not be construed more strictly against one party than against the other merely by virtue of the fact that it may have been prepared by counsel for one of the parties, it being recognized that both Seller and Buyer have contributed substantially and materially to the preparation of this Agreement and have been represented by counsel experienced in commercial transactions of this nature.
- 17.11 Attorneys' Fees. In the event of a judicial or administrative proceeding or action by one party against the other party with respect to the interpretation or enforcement of this Agreement, the prevailing party in a final unappealable proceeding shall be entitled to recover reasonable costs and expenses including reasonable attorneys' fees and expenses, whether at the investigative, pretrial, trial or appellate level. The prevailing party shall be determined by the court based upon an assessment of which party's major arguments or position prevailed. The provisions of this Section 17.11 shall survive the Closing hereunder.

29

- 17.12 <u>Merger</u>. Except as otherwise specifically provided herein or in any closing document, the acceptance of the deed by the recordation thereof shall be deemed to be a full and complete performance and discharge of every agreement and obligation of the Seller herein contained. Without limiting the generality of the foregoing, the following provisions shall survive Closing hereunder to the extent provided in the applicable provision: Article 3, Section 4.3, Section 6.1, Section 6.8, Article 9, Article 12, Article 13, Article 14, Article 15 and Article 17.
- 17.13 Severability. If any one or more of the provisions hereof shall for any reason be held to be invalid, illegal or unenforceable in any respect, such invalidity, illegality or unenforceability shall not affect any other provision hereof, and this Agreement shall be construed as if such invalid, illegal or unenforceable provision had never been contained herein unless such severed provisions is material or affects the fundamental benefits of this Agreement to one or both of the parties.
- 17.14 <u>Business Day.</u> As used herein, "<u>Business Day.</u>" shall mean any day other than a Saturday or Sunday or a legal holiday, which is any holiday for which financial institutions and post offices are generally closed in the State of Maine and/or in the City of Boston. If any date herein set forth for the performance of any obligations of Seller or Buyer or for the delivery of any instrument or notice as herein provided should be on a day that is not a Business Day, the compliance with such obligations or delivery shall be deemed acceptable on the next Business Day.

[Signatures on Next Page]

30

[Signature Page to Purchase and Sale Agreement]

IN WITNESS WHEREOF, the parties have executed this Agreement under seal and delivered this Agreement as of the date first above written.

SELLER:

CW WESTBROOK LIMITED PARTNERSHIP,

a Delaware limited partnership

By:CW Westbrook Corp., a Delaware corporation, its general partner

By:/s/Stephen R. Weiner (SEAL)

Name: Stephen R. Weiner

Title: President

BUYER:

IDEXX LABORATORIES, INC.

a Delaware Corporation

By:/s/Jonathan W. Ayers (SEAL)

Name: Jonathan W. Ayers

Title: CEO

LIMITED JOINDER

The undersigned, Stephen R. Weiner, hereby joins in the execution of the foregoing Purchase and Sale Agreement solely for the purposes set forth in Section 3.5 thereof. Notices to the undersigned shall be provided at c/o S.R. Weiner & Associates, Inc., 1330 Boylston Street, Chestnut Hill, Massachusetts 02467, or such other address as the undersigned shall specify from time to time, by notice in compliance with the notice provisions of Article 14 above.

/s/Stephen R. Weiner	
Stephen R. Weiner	

LIMITED JOINDER OF ESCROW AGENT

The undersigned Escrow Agent joins in this Agreement for the limited purpose of acknowledging its agreement to act as Escrow Agent hereunder and to accept, hold and account for the Deposit in accordance with the terms of this Agreement. Escrow Agent acknowledges that it is not a party to this Agreement and need not be joined in any Amendment hereto, unless such amendment modifies the provisions of Section 16 pertaining to the Escrow Agent.

ESCROW AGENT:

LAWYERS TITLE INSURANCE COMPANY

By:/s/C. David Keith (SEAL)

Name: C. David Keith

Title: State Manager and Counsel

EXHIBITS:

Exhibit A - Legal Description

Exhibit B - List of Assumed Loan Documents

Exhibit C - List of Leases

Exhibit D - List of Security Deposits Under Leases

Exhibit E - List of Contracts

Exhibit F-1 - List of Pending Litigation
Exhibit F-2 - List of Asserted Violations of Law
Exhibit G - List of Environmental Reports

Exhibit G-1 - Notices of Any Environmental Violations

Exhibit H - Due Diligence Materials

Exhibit I - Form of Deed

Exhibit J - Form of Bill of Sale and General Assignment

Exhibit K - Form of Assignment and Assumption of Leases and Contracts
Exhibit L - Form of FIRPTA Affidavit (Non-Foreign Certification)

EXHIBITS

EXHIBIT A

LEGAL DESCRIPTION 80 EISENHOWER DRIVER, WESTBROOK, MAINE

The land and all improvements thereon located in the City of Westbrook, County of Cumberland and State of Maine bounded and described as follows:

Beginning at a monument on the southwesterly side of said Eisenhower Drive marking the most northerly corner of land conveyed by Greater Portland Building Fund, Inc. to Kidder Press Company, Inc. (having been merged into Moore Business Forms, Inc. on September 30, 1970) by deed dated October 1, 1969 and recorded in the Cumberland County Registry of Deeds in Book 3107, Page 726; thence by the southwesterly line of said Eisenhower Drive on a course of N 40° 47' W a distance of 1,980.65 feet to a point and other land now or formerly of Greater Portland Building Fund; thence by said Greater Portland Building Fund land and land now or formerly of Clifford Libby and Richard Libby on a course of S 55° 33'20" W a distance of 1,207.27 feet to an iron and land now or formerly of Central Maine Power Company; thence by said Central Maine Power Company land on a course of S 40° 18'45" E a distance of 2,118.73 feet to an iron and land now or formerly of Bennet; thence by said Bennet land on a course of N 49° 03' E a distance of 115.13 feet to an iron; thence continuing by said Bennet land on a course of N 48° 33' E a distance of 535.98 feet to an iron at the most westerly corner of land now or formerly of Moore Business Forms, Inc.; thence by land now or formerly of Moore Business Forms, Inc. land on a course of N 49° 24'26" E a distance of 566.23 feet to the point of BEGINNING.

Being the premises conveyed to CW Westbrook Limited Partnership by deed dated September 22, 1993 and recorded with the Cumberland County Registry of Deeds in Book 10971, Page 316.

A-1

EXHIBIT B

LIST OF ASSUMED LOAN DOCUMENTS

- 1. <u>Promissory Note</u> in the original principal amount of Nine Million Five Hundred Thousand Dollars (\$9,500,000.00), made payable to Sun Life Assurance Company of Canada, a Canadian corporation, dated April 24, 1995, under which CW Westbrook Limited Partnership is the maker;
- 2. Indemnity Agreement executed by Stephen R. Weiner in favor of Sun Life Assurance Company of Canada, dated April 24, 1995;

- 3. <u>Indemnity Agreement</u> executed by Julian Cohen in favor of Sun Life Assurance Company of Canada, dated April 24, 1995;
- Mortgage and Security Agreement from CW Westbrook Limited Partnership, as Mortgagor, in favor of Sun Life Assurance Company of Canada, dated April 24, 1995 and recorded with the Cumberland County Registry of Deeds in Book 11896, Page 004;
- 5. <u>UCC-1 Financing Statement</u> filed with the Cumberland County Registry of Deeds in Book 11896, Page 035;
- 6. <u>Collateral Assignment of Leases and Rents</u>, dated April 24, 1995, between CW Westbrook Limited Partnership, as Assignor, and Sun Life Assurance Company of Canada, as Assignee, recorded with the Cumberland County Registry of Deeds in Book 11896, Page 026;
- 7. Letter Agreement between Sun Life Assurance Company of Canada and CW Westbrook Limited Partnership, dated April 24, 1995;
- 8. <u>UCC-1 Financing Statement filed with the New Hampshire Secretary of State on April 25, 1995 as Document No. 1121819;</u>
- 9. <u>UCC-1 Financing Statement</u> filed with the City of Brookline, Massachusetts on April 24, 1995 as Document No. 30195;
- 10. <u>UCC-1 Financing Statement</u> filed with the Secretary of the Commonwealth of Massachusetts on April 24, 1995 as Document No. 308218.

Outstanding Principal Balance of the Assumed Loan as of November 30, 2005: \$6,670,123

There are no Escrow Accounts under the Assumed Loan.

B-1

EXHIBIT C

LIST OF LEASES

- 1. Lease between Data General Corporation, as Landlord, and IDEXX Corporation, as Tenant, dated December 7, 1990, as amended by Amendment No. 1 to Lease, dated as of March 15, 1993, and as amended by Amendment No. 2 to Lease, dated as of August 11, 1997, and as amended by Amendment No. 3 to Lease, dated as of August 6, 2003 (the "IDEXX Lease").
- 2. Net Building Lease dated May 18, 1990, by and between DATA GENERAL CORPORATION, as Landlord, and PRECISmetals, Inc., as Tenant, as affected by Amendment No. 1 to Lease dated September 15, 1991, Amendment No. 2 to Lease dated April 10, 1992, Amendment No. 3 to Lease dated December 3, 1992, Amendment No. 4 to Lease dated March 18, 1993, and Amendment No. 5 to Lease dated November 4, 1993.(the "Sanmina Lease") Note: The tenant under this lease (currently Sanmina SCI) has vacated the premises, and the term of this Lease expires by its terms on March 31, 2006. This Lease will no longer be in effect at the time of Closing under this Purchase and Sale Agreement, except for any post-termination or post-expiration obligations.

C-1

EXHIBIT D

LIST OF SECURITY DEPOSITS UNDER LEASES

1. \$59,573.41 Security Deposit under the IDEXX Lease (which includes interest accrued through December 31, 2005).

D-1

EXHIBIT E

LIST OF CONTRACTS

- 1. Contract for Snow Plowing between CW Westbrook Limited Partnership and Gorham Sand & Gravel, dated October 1, 2005.
- 2. Property Maintenance/Landscaping Service Contract between CW Westbrook Limited Partnership and SimplyUnique, dated January 6, 2005.
- 3. Proposal and Contract with Industrial Roofing Corp. dated April 7, 2005.
- 4. Peace of Mind 24/7 Roof Protection Program Agreement No. S-325-06, with IRC Roof Management Services, dated November 23, 2005.
- 5. Contract for Fire System with Norris, Inc., dated June 24, 2005.

F1-1

EXHIBIT F-2

LIST OF ASSERTED VIOLATIONS OF LAW

1. None

F2-1

EXHIBIT G

LIST OF ENVIRONMENTAL REPORTS

As of January 17, 2006

- 1. Updated Site Assessment dated April 13, 1995, prepared by TRC Environmental Corporation.
- 2. "Rescoring" Letter from the United States Environmental Protection Agency to Data General Corp., dated September 17, 1993.
- 3. Site Assessment prepared by TRC Environmental Corporation, dated July 16, 1993.
- 4. Phase 1 Environmental Site Assessment, Steego Auto Parts Depot, July 1997, prepared by Fluor Daniel GTI.
- 5. Summary of Third Party Review, Subsurface Environmental Assessment, dated June, 2005, prepared by Sanborn, Head & Associates, Inc.
- 6. Report of TRC Environmental dated May 23, 1995.
- 7. Report of IDEXX Corporation's Use of Radioactive Materials dated July 21, 1993, prepared by Physics Consultants, Inc.
- 8. Letter dated June 30, 1993 from Maine DEP to Data General.
- Letter dated July 6, 1993 from Jensen Baird Gardner & Henry to Goulston & Storrs;
- 10. Letter dated June 1, 1992 from Ernest Waterman of US EPA to James Jacobs of Data General.
- 11. Closure Plan dated March 29, 1991 prepared by Groundwater Technology, Inc.
- 12. Site Assessment of TRAC Environmental Corporation, dated July, 1993.
- 13. Closure Report, Data General Corporation, Submitted to Maine DEP, submitted by Groundwater Technology Inc., dated February 1993.
- 14. Compliance Inspection Results, Electronic manufacturing systems, Overhead Cranes & Hoists, dated February 1999.
- 15. Electronic Manufacturing Systems, Westbrook, Maine, Annual Inspection Report, Proposal 0009 TL, dated February 1998, prepared by HCSG.
- 16. Electronic Manufacturing Systems, Westbrook, Maine, Annual Inspection Report, Proposal 0281 TL, dated December 1995, prepared by HCSG.

G-1

- 17. Electronic Manufacturing Systems, Westbrook, Maine, Annual Inspection Report, Proposal 0425 TL, dated January 1997, prepared by HCSG.
- 18. Periodic Inspection Results, Electro-Mechanical Solutions, Cranes & Hoists, dated March 6, 2003, prepared by Somatex.
- 19. Periodic Inspection Results, Electro-Mechanical Solutions, Cranes & Hoists, dated February 29, 2000, prepared by Somatex.
- 20. Inspection Reports, E.M. Solutions, dated February 2001, prepared by Hooker Handley Systems, Inc.

G1-1

EXHIBIT H

DUE DILIGENCE MATERIALS

See attached.

H-1

EXHIBIT A

DUE DILIGENCE MATERIALS

- A current rent roll reflecting prepaid rent, security deposits, delinquencies, base rent, escalations, and percentage rent, if any, and historical occupancy reports for the periods covered by the financial statements and current year operating statements;
- b. Plans and specifications of all improvements on the Property;
- All existing soil and environmental reports;
- d. All existing engineering reports, including, but not limited to, structural, plumbing, electrical, mechanical, and civil;
- e. Copies of all maintenance, service, employment, and vendor contracts;
- f. Copies of real estate tax bills and real estate tax information for the current and three preceeding fiscal years;
- g. Copies of all leases (and applicable subleases) pertaining to the Property;
- h. Copies of all title policies issued with respect to the Property as of or following the Seller's acquisition of the Property with copies of all the underlying documents referenced in such policies (to the extent in Seller's possession or control);
- i. The most recent metes and bounds as-built and perimeter surveys and plot plans in the Seller's possession;
- j. Copies of any documents regarding existing rights or options to purchase or lease any part or all of the Property;
- k. Copies of any pending or threatened litigation affecting the Property, or the Seller;
- 1. Certificates of Occupancy for the Property (to the extent in Seller's possession or control);
- m. Such other data, information, plans, files, letters and materials pertaining to the operation of the Property or the ownership or debt with respect to the Property or the Seller as Purchaser shall reasonably request;
- n. Copies of any mortgages secured by the Property and the notes secured thereby, and all other loan documentation pertaining thereto.
- o. Copies of all documents evidencing or supporting environmental liens; Maine Voluntary Response Action Plan no action assurances or certificates of completion; institutional Controls including deed restrictions, activity and use limitations, and Environmental Covenants; and any environmental requirements issued by any governmental authority specifically for the Property.

S.R. WEINER

AND ASSOCIATES INCORPORATED

January 6, 2006

Brian Connelly IDEXX Laboratories, Inc. One IDEXX Drive Westbrook, ME 04092

Re: Due Diligence Materials

Dear Brian:

We are sending you all of the materials requested in Exhibit A as further described below.

Box 1 contains all information by letter designation on Exhibit A, except item c. Box 2 contains item c.

As follows are comments on some of the items as appropriate.

b. Several packages of information.

Package 1 represents old original drawings provided to us by Data General at the time of our purchase

Package 2 Information concerning structural roof repairs in years 2003 and 2004.

Package 3 Information concerning parking lot addition in years 2003 and 2004.

Package 4 Fire Alarm System Upgrade 2002

Package 1 Sanmina periodic inspection reports

Package 2 Our File # 3.2 containing all environmental reports from the time of our acquisition

We are also including a Phase report done in July 1997 on the Steego Auto Parts Depot and a report dated June 2005 for Subsurface Environment Assessment in connection with Sanmina performed by Sanborn, Head.

- g. We have not provided any of the leases since the Sanmina lease will expire prior to closing and we know you have a copy of your lease.
- h. We believe all title information has already been provided to you by our counsel, Phillip Levy.
- j. There are no rights to purchase the property, but we have included our files 1.4 a and 6.0 regarding easements and utility agreements.
- k. There are none. m. Nothing included.
- m. Nothing included.
- n. We are including the loan commitment, the promissory note, the mortgage security agreement, United States Environmental Protection Agency letter and Indemnity Agreement.
- o. All related information has been included with the information for Exhibit A c.

If you have any questions, please do not hesitate to call.

Sincerely,

/s/ Thomas J. DeSimone

Thomas J. DeSimone Executive Vice President

cc: Phillip Levy

EXHIBIT I

FORM OF DEED

(Statutory Short Form)

CW WESTBROOK LIMITED PARTNERSHIP ("Grantor") herein), a Delaware limited partnership having its principal place of business at c/o S.R. Weiner and Associates, Inc., 1330 Boylston Street, Chestnut Hill, Massachusetts 02467,

in consideration of One Dollar (\$1.00) and other valuable consideration paid by

("Grantee" herein), a Delaware corporation having a mailing address of 80 Eisenhower Drive, Westbrook, Maine 04092, the receipt and sufficiency of which are hereby acknowledged by Grantor,

does hereby grant to said Grantee, its successors and assigns, forever, with quitclaim covenant, a certain lot or parcel of land, together with improvements thereon, situated on the southwesterly side of Eisenhower Drive in the City of Westbrook, County of Cumberland and State of Maine, being bounded and described as follows:

Beginning at a monument on the southwesterly side of said Eisenhower Drive marking the most northerly corner of land conveyed by Greater Portland Building Fund, Inc. to Kidder Press Company, Inc. (having been merged into Moore Business Forms, Inc. on September 30, 1970) by Deed dated October 1, 1969, and recorded in the Cumberland County Registry of Deeds in Book 3107, Page 726; thence by the southwesterly line of said Eisenhower Drive on a course of N 40° 47' W a distance of 1,980.65 feet to a point and other land now or formerly of Greater Portland Building Fund; thence by said Greater Portland Building Fund land and land now or formerly of Clifford Libby and Richard Libby on a course of S 55° 33' 20" W a distance of 1,207.27 feet to an iron and land now or formerly of Central Main Power Company; thence by said Central Main Power Land on a course of S 40° 18' 45"E a distance of 2,118.73 feet to an iron and land now or formerly of Bennet; thence by said Bennet land on a course of N 49° 03' E a distance of 135.13 feet to an iron; thence continuing by said Bennet land on a course of N 48° 33' E a distance of 535.98 feet to an iron at

the most westerly corner of said Moore Business Forms, Inc. land; thence by said Moore Business Forms, Inc. land on a course of N 49° 24' 26" E a distance of 566.23 feet to the point of BEGINNING.

Being the premises conveyed to this Grantor by Data General Corporation by Deed dated September 22, 1993, and recorded in said Registry of Deeds in Book 10971, Page 316.

The premises conveyed hereby are being conveyed subject to the Permitted Exceptions set forth on Exhibit A attached hereto.

I-1

This conveyance is made further subject to those notes and conditions set forth in Orders of the Department of Environmental Protection dated April 28, 1976 and recorded in the Cumberland County Registry of Deeds in Book 3841, Page 166; dated July 27, 1977 and recorded in said Registry in Book 4079, Page 80; dated December 12, 1984 and recorded in said Registry in Book 7502, Page 340; and dated October 10, 1984 and recorded in said Registry in Book 7503, Page 5, insofar as applicable.

This conveyance is further subject to a certain Mortgage and Security Agreement granted by Grantor to Sun Life Assurance Company of Canada dated April 24, 1995 and recorded with said Registry in Book 11896, Page 004 (the "Mortgage"), securing a note in the original principal amount of Nine Million Five Hundred Thousand Dollars (\$9,500,000.00), which Mortgage is being assumed by Grantee. The Mortgage has been assigned by Sun Life Assurance Company of Canada to The Chase Manhattan Bank by instrument dated March 13, 1997 and recorded with said Registry in Book 13020, Page 292, and by The Chase Manhattan Bank to The Bank of New York, as successor Trustee under Trust Agreement dated March 31, 1997 by Assignment dated December 31, 1997 and recorded with the Registry in Book 13766, Page 233.

To have and to hold the same, together with all the privileges and appurtenances thereunto belong, to the said Grantee, its successors and assigns forever.

And the said Grantor does covenant with the said Grantee, its successors and assigns, that it will warrant and forever defend the premises to the said Grantee, its successors and assigns forever, against the lawful claims and demands of all persons claiming by, through or under Grantor, except for the Permitted Exceptions set forth on Exhibit A attached hereto.

GI-2

SIGNED, SEALED AND DELIVERD in presence of	CW WESTBROOK LIMITED PARTNERSHIP, a Delaware limited partnership
	By: CW Westbrook Corp., a Delaware corporation, its general partner
	Ву:
	NAME: TITLE:
	ned notary public, personally appeared, the of PARTNERSHIP, proved to me through satisfactory evidence of identification, which were preceding document, and acknowledged to me that (s)he signed the above document voluntarily for its
	Notary Public
	My commission expires: ————
	GI-3
	EXHIBIT A

PERMITTED EXCEPTIONS

[To be determined and listed in accordance with Article 5 of the Purchase and Sale Agreement]

GI-4

BILL OF SALE AND GENERAL ASSIGNMENT AND ASSUMPTION AGREEMENT

This BILL OF SALE AND GENERAL ASSIGNMENT AND ASSUMPTION AGREEMENT (the " <u>Agreement</u> ") is made as of the day of, 2006, by and between CW WESTBROOK LIMITED PARTNERSHIP, a Delaware limited partnership with an address of c/o S.R. Weiner and Associates, Inc., 1330 Boylston Street, Chestnut Hill, Massachusetts 02467 (" <u>Assignor</u> "), and [] corporation, having an address of [] ("Assignee").
For Ten Dollars (\$10.00) and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged by Assignor, Assignor and Assignee do acreby agree as follows:
Sale and Assignment of Personal Property.
(a) Assignor hereby sells, transfers, assigns, delivers, sets over and conveys to Assignee all of Assignor's right, title and interest in and to all furniture, fixtures, equipment including, without limiting the generality of the foregoing, any of the following: boilers, pumps, tanks, electric panel switchboards, lighting equipment and wiring, heating, plumbing, ventilating and air conditioning apparatus and equipment, elevators, escalators, and conveyors, and all other and other personal property owned by Assignor now existing or at any time hereafter placed in or attached to the property described on Exhibit A attached hereto (the "Property").
(b) The personal property hereby sold, transferred, assigned, delivered, set over and conveyed is being sold on an "as is, where is" basis.
2. General Assignment and Assumption.
(a) Assignor hereby sells, transfers, assigns, delivers, sets over and conveys to Assignee all right, title and interest of Assignor in and to the following:
(i) to the extent they may be transferred under applicable law, all licenses, permits and authorizations presently issued in connection with the operation of all or any part of the Property as it is presently being operated;
(ii) to the extent assignable, all contract rights, licenses, permits, approvals, utility contracts, telephone exchange numbers, advertisin materials, plan and specifications, drawings, surveys, governmental approvals and development rights and all warranties and guaranti (if any) issued to or held by Assignor by any manufacturer, supplier, contractor, subcontractor or materialman in connection with the construction or installation of any improvements on the Property or the equipment or fixtures included as part of the Property or the maintenance of the Property;
J-1
(iii) to the extent assignable, all trade names and general intangibles relating to the Property.
((i) through (iii) collectively referred to as the "Property Interests").
(b) Assignee hereby assumes all obligations of Assignor in connection with or arising out of the Property Interests identified on Exhibit B attached hereto and accruing after the date hereof.
3. <u>Representations and Warranties</u> . Subject to the provisions of Section 4 below, this Bill of Sale and General Assignment and Assumption Agreement is made without ecourse and without representation or warranty of any kind whatsoever, express or implied or by operation of law, except as set forth herein.
4. <u>Purchase Agreement</u> . This Assignment shall in no event enlarge, reduce or otherwise affect the rights or obligations of the parties as set forth in that certain Purchase and Gale Agreement between the parties dated January 17, 2006 pertaining to the Property.
5. Successors and Assigns. This Agreement shall be binding upon and shall inure to the benefit of the parties hereto and their respective successors and assigns.
6. Governing Law. This Agreement shall be governed by and construed in accordance with the laws of the State of Maine, without regard to conflicts of law principles.
7. <u>Counterparts</u> . This Agreement may be executed in two or more counterparts, each of which shall be deemed an original, but all of which together shall constitute one and he same instrument.
[Signatures On Next Page]
J-2
[Signature Page To Bill of Sale and General Assignment and Assumption Agreement]
IN WITNESS THE EXECUTION HEREOF, under seal, in any number of counterpart copies, each of which shall be deemed to be an original for all purposes, as of the lay of, 2006.
ASSIGNOR:
CW WESTBROOK LIMITED PARTNERSHIP,

By:CW Westbrook Corp., a Delaware corporation, its general partner

a Delaware limited partnership

By:	(SEAL)
Name: Title:	
ASSIGNEE:	
a	,
By:	(SEAL)
Name: Title:	
J-3	

EXHIBIT A

TO

BILL OF SALE AND GENERAL ASSIGNMENT AND ASSUMPTION AGREEMENT

LEGAL DESCRIPTION 80 EISENHOWER DRIVER, WESTBROOK, MAINE

The land and all improvements thereon located in the City of Westbrook, County of Cumberland and State of Maine bounded and described as follows:

Beginning at a monument on the southwesterly side of said Eisenhower Drive marking the most northerly corner of land conveyed by Greater Portland Building Fund, Inc. to Kidder Press Company, Inc. (having been merged into Moore Business Forms, Inc. on September 30, 1970) by deed dated October 1, 1969 and recorded in the Cumberland County Registry of Deeds in Book 3107, Page 726; thence by the southwesterly line of said Eisenhower Drive on a course of N 40° 47' W a distance of 1,980.65 feet to a point and other land now or formerly of Greater Portland Building Fund; thence by said Greater Portland Building Fund land and land now or formerly of Clifford Libby and Richard Libby on a course of S 55° 33'20" W a distance of 1,207.27 feet to an iron and land now or formerly of Central Maine Power Company; thence by said Central Maine Power Company land on a course of S 40° 18'45" E a distance of 2,118.73 feet to an iron and land now or formerly of Bennet; thence by said Bennet land on a course of N 49° 03' E a distance of 115.13 feet to an iron; thence continuing by said Bennet land on a course of N 48° 33' E a distance of 535.98 feet to an iron at the most westerly corner of land now or formerly of Moore Business Forms, Inc.; thence by land now or formerly of Moore Business Forms, Inc. land on a course of N 49° 24'26" E a distance of 566.23 feet to the point of BEGINNING.

Being the premises conveyed to CW Westbrook Limited Partnership by deed dated September 22, 1993 and recorded with the Cumberland County Registry of Deeds in Book 10971, Page 316.

JA-1

EXHIBIT K

ASSIGNMENT AND ASSUMPTION AGREEMENT RE: LEASES AND CONTRACTS

THIS	ASSIGNMENT AND ASSI	UMPTION AGREEMENT is made as of this day of _	, 2006, by and between CW WESTBROOK LIMITED
PARTNE	SHIP, a Delaware limited pa	artnership with an address of c/o S.R. Weiner and Associa	tes, Inc., 1330 Boylston Street, Chestnut Hill, Massachusetts 02467 ("Assignor"),
and [], a [], having an address of [] ("Assignee").

WITNESSETH:

WHEREAS, Assignee has this date purchased from Assignor certain real property (the "Premises") known and numbered as 80 Eisenhower Drive located in Westbrook, Cumberland County, Maine, all more particularly described on Exhibit A attached hereto made a part hereof, and

WHEREAS, under the terms and conditions of a certain Purchase and Sale Agreement dated January 17, 2006 pursuant to which the Premises were purchased (the "Purchase and Sale Agreement"), it was contemplated that Assignor and Assignee would enter into this Assignment;

NOW, THEREFORE, in consideration of the premises herein contained and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, and intending to be legally bound hereby, the parties hereto do hereby agree as follows:

- 1. Assignor hereby transfers and assigns to Assignee all right, title and interest of Assignor in and to the following described property:
 - (a) All leases, subleases and other occupancy agreements relating to or affecting the Premises, together with all guarantees of obligations of tenants and other parties under such leases and agreements, and copies of all files in connection with the foregoing, including, without limitation, the leases and other agreements more fully described on Exhibit B attached hereto and made a part hereof (collectively, the "Leases"); provide, however, with respect to the Sanmina Lease, Seller assign its rights to Buyer in common with Seller such that both parties shall have the right to enforce such rights independently of the other; and

(b) The current outstanding balance of all security deposits, escrows, prepaid rents, prepaid operating expenses and other prepaid items, together with all interest accrued thereon, as more fully described on Exhibit C attached hereto and made a part hereof (collectively, the "Security Deposits");

TO HAVE AND TO HOLD all of the foregoing unto the Assignee, its successors and assigns, from and after the date hereof, subject to the terms, covenants, conditions and provisions contained herein.

K-1

2. Assignor hereby transfers and assigns to Assignee all right, title and interest of Assignor in and to all those certain management, service, supply and maintenance agreements, equipment leases and other contracts with respect to or affecting the Premises specifically listed on Exhibit D attached hereto and made a part hereof (collectively, the "Assumed Contracts").

TO HAVE AND TO HOLD all of the foregoing unto the Assignee, its successors and assigns, from and after the date hereof, subject to the terms, covenants, conditions and provisions contained herein.

- 3. Assignee hereby accepts the foregoing assignment of the Leases, Security Deposits and Contracts and hereby assumes all the duties and obligations of Assignor accruing after the date hereof under or with respect to the Leases, Security Deposits and Contracts; provided, however, that Assignee does not assume, and Assignor shall retain all liability and responsibility for, the duties and obligations of Assignor under the "Samina Lease", as defined on Exhibit B. Assignee shall indemnify, defend and hold Assignor harmless from and against any and all claims, liabilities and costs (including reasonable attorneys' fees and costs) arising out of or relating to Assignee's failure to perform any duty or obligation assumed by Assignee under the Leases, exclusive of the Samina Lease, or Contracts or with respect to the Security Deposits. Assignor shall indemnify, defend and hold Assignee harmless from and against any and all claims, liabilities and costs (including reasonable attorneys' fees and costs) arising out of or relating to Assignor's failure to perform any duty or obligation of Assignor under (a) the Sammina Lease including the obligation to pay, any and all amounts owed to the tenant under the Sammina Lease, including without limitation the return of any security deposit, reconciliation of operating expenses, real estate taxes and prepaid rent;; and (b) the Assumed Contracts accruing on or prior to the date hereof. This Agreement shall bind and inure to the benefit of the parties hereto and their respective successors and assigns.
 - 5. This Assignment shall in no event enlarge, reduce or otherwise affect the rights or obligations of the parties as set forth in the Purchase and Sale Agreement.
- 6. This Agreement may be executed in two or more counterparts, each of which shall be deemed an original, but all of which together shall constitute one and the same instrument.
 - 7. Governing Law. This Agreement shall be governed by and construed in accordance with the laws of the State of Maine, without regard to conflicts of law principles.

[Signatures On Next Page]

K-2

[Signature Page To Assignment And Assumption Agreement]

IN WITNESS WHEREOF, the parties have executed this Agreement under seal on the day and year first above written.

TNESS:	ASSIGNOR:	
	CW WESTBROOK LIMITED PARTNERSHI a Delaware limited partnership	IP,
	By:CW Westbrook Corp., a Delaware corporation, its general partner	
	By: (S	EAL
	Name: Title:	
	ASSIGNEE:	
	a,	
	By: (S	EAL)
	Name: Title:	

ASSIGNMENT AND ASSUMPTION AGREEMENT RE: LEASES AND CONTRACTS

<u>LEGAL DESCRIPTION</u> 80 EISENHOWER DRIVER, WESTBROOK, MAINE

The land and all improvements thereon located in the City of Westbrook, County of Cumberland and State of Maine bounded and described as follows:

Beginning at a monument on the southwesterly side of said Eisenhower Drive marking the most northerly corner of land conveyed by Greater Portland Building Fund, Inc. to Kidder Press Company, Inc. (having been merged into Moore Business Forms, Inc. on September 30, 1970) by deed dated October 1, 1969 and recorded in the Cumberland County Registry of Deeds in Book 3107, Page 726; thence by the southwesterly line of said Eisenhower Drive on a course of N 40° 47' W a distance of 1,980.65 feet to a point and other land now or formerly of Greater Portland Building Fund; thence by said Greater Portland Building Fund land and land now or formerly of Clifford Libby and Richard Libby on a course of S 55° 33'20" W a distance of 1,207.27 feet to an iron and land now or formerly of Central Maine Power Company; thence by said Central Maine Power Company land on a course of S 40° 18'45" E a distance of 2,118.73 feet to an iron and land now or formerly of Bennet; thence by said Bennet land on a course of N 49° 03' E a distance of 115.13 feet to an iron; thence continuing by said Bennet land on a course of N 48° 33' E a distance of 535.98 feet to an iron at the most westerly corner of land now or formerly of Moore Business Forms, Inc.; thence by land now or formerly of Moore Business Forms, Inc. land on a course of N 49° 24'26" E a distance of 566.23 feet to the point of BEGINNING.

Being the premises conveyed to CW Westbrook Limited Partnership by deed dated September 22, 1993 and recorded with the Cumberland County Registry of Deeds in Book 10971, Page 316.

KA-1

EXHIBIT B

TO

ASSIGNMENT AND ASSUMPTION AGREEMENT RE: LEASES AND CONTRACTS

LEASES

- 1. Lease between Data General Corporation, as Landlord, and IDEXX Corporation, as Tenant, dated December 7, 1990, as amended by Amendment No. 1 to Lease, dated as of March 15, 1993, and as amended by Amendment No. 2 to Lease, dated as of August 11, 1997, and as amended by Amendment No. 3 to Lease, dated as of August 6, 2003 (the "IDEXX Lease").
- 2. Net Building Lease dated May 18, 1990, by and between DATA GENERAL CORPORATION, as Landlord, and PRECISmetals, Inc., as Tenant, as affected by Amendment No. 1 to Lease dated September 15, 1991, Amendment No. 2 to Lease dated April 10, 1992, Amendment No. 3 to Lease dated December 3, 1992, Amendment No. 4 to Lease dated March 18, 1993, and Amendment No. 5 to Lease dated November 4, 1993. (the "Sanmina Lease")

KB-1

EXHIBIT C

TO

ASSIGNMENT AND ASSUMPTION AGREEMENT RE: LEASES AND CONTRACTS

SECURITY DEPOSITS, ESCROWS AND OTHER PREPAID ITEMS

(to be attached)

KC-1

EXHIBIT D

TO

 ${\bf ASSIGNMENT\ AND\ ASSUMPTION\ AGREEMENTRE:} \\ {\bf LEASES\ AND\ CONTRACTS}$

CONTRACTS

(to be attached)

KD-1

EXHIBIT L

NON-FOREIGN CERTIFICATION(FIRPTA AFFIDAVIT)

Section 1445 of the Internal Revenue Code provides that a transferee (or buyer) of a U.S. real property interest must withhold tax if the transferor (or seller) is a foreign person. To inform the transferee that withholding of a tax is not required in connection with the transfer of a U.S. real property interest by CW WESTBROOK LIMITED PARTNERSHIP, a Delaware limited partnership ("Seller"), the undersigned being duly authorized hereby certifies as to the following:

- Neither Seller nor any partner therein is a foreign corporation, foreign partnership, foreign trust, foreign estate or non-resident alien individual (as those terms are defined in the Internal Revenue Code Section 1445 and regulations promulgated thereunder or under other provisions applicable thereto);
 - 2. The U.S. Taxpayer Identification Number of Seller is 04-3204237;

3. The address of Seller is c/o S.R.	Weiner and Associates, Inc., 1330 Boylston	Street, Chestnut Hill, Ma	assachusetts 02467.		
The undersigned understands that the prisonment or both.	is Certification may be disclosed to the Inter	nal Revenue Service and	that any false statement co	ontained herein could be punished by fin	Э,
Under penalties of perjury, the under	rsigned declares that he has examined this ce	ertification and to the best	of his knowledge and bel	lief it is true, correct and complete.	
Date:, 2006					
		CW WESTBROOK a Delaware limited pa	X LIMITED PARTNERS partnership	SHIP,	
		By:CW Westbrook C a Delaware corporation	Corp., ion, its general partner		
		By:		(SEAL)	
		Name: Title:			
		L-1			

IDEXX LABORATORIES, INC.

SUBSIDIARIES OF THE COMPANY

NAME

JURISDICTION OF INCORPORATION

Cardiopet Incorporated Diavet Labor AG Dr. Bommeli AG Genera Technologies Lin

Genera Technologies Limited
IDEXX Computer Systems, Inc.
IDEXX Distribution, Inc.
IDEXX Europe B.V
IDEXX GmbH
IDEXX Holding GmbH
IDEXX Laboratories B.V

IDEXX Laboratories Canada Corporation

IDEXX Laboratories Italia S.r.l
IDEXX Laboratories, KK
IDEXX Laboratories, KK
IDEXX Laboratories Limited
IDEXX Laboratories (NZ) Limited
IDEXX Laboratories Pty. Limited
IDEXX Laboratories, S. de R.L. de C.V
IDEXX Laboratorios, S.L.

IDEXX Laboratorios, S.L IDEXX Laboratories Inc. IDEXX Operations, Inc. IDEXX Pharmaceuticals, Inc. IDEXX Reference Laboratories, Inc. IDEXX S A R I

IDEXX S.A.R.L IDEXX Scandinavia AB IDEXX UK Acquisition Limited Mialot Lagadic SAS Syracuse Bioanalytical, Inc.

Veterinary Pathology Services Pty. Limited

Vet France SARL
Vet Med Lab S.r.l
Vet Med Labor GmbH
Vet Med Labor GmbH
Vet Med Lab ApS
Vet Med Lab BV
Vet Med Lab (UK) Ltd

All of the Company's subsidiaries are wholly owned.

Switzerland Switzerland England and Wales Delaware Massachusetts The Netherlands Germany Germany The Netherlands Canada

Delaware

Italy
Japan
England and Wales
New Zealand
Australia
Mexico
Spain
Taiwan R.O.C.
Delaware
Delaware
Delaware
France

Sweden England and Wales

France
New York
Australia
France
Italy
Austria
Germany
Denmark
The Netherlands
United Kingdom

CONSENT OF INDEPENDENT REGISTERED PUBLIC ACCOUNTING FIRM

We hereby consent to the incorporation by reference in the Registration Statement on Form S-3 (Nos. 333-36007 and 333-70846) and Form S-8 (Nos. 33-41806, 33-42845, 33-42846, 33-48404, 33-61494, 33-64202, 33-64204, 33-95616, 333-11201, 333-11199, 333-36009, 333-56685, 333-78769, 333-43172, 333-43170, 333-27733, 333-105426, 333-108906 and 333-108905) of IDEXX Laboratories, Inc. of our report dated March 10, 2006 relating to the financial statements, financial statement schedule, management's assessment of the effectiveness of internal control over financial reporting, which appears in this Form 10-K.

/s/PricewaterhouseCoopers LLP

Boston, Massachusetts March 10, 2006

CERTIFICATION

I, Jonathan W. Ayers, certify that:

- 1) I have reviewed this annual report on Form 10-K for the year ended December 31, 2005 of IDEXX Laboratories, Inc. (the "Annual Report");
- 2) Based on my knowledge, this Annual Report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this Annual Report;
- 3) Based on my knowledge, the financial statements, and other financial information included in this Annual Report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this Annual Report;
- 4) The registrant's other certifying officers and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
 - a) designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this Annual Report is being prepared;
 - b) designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
 - c) evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this Annual Report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this Annual Report, based on such evaluation; and
 - d) disclosed in this Annual Report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5) The registrant's other certifying officer(s) and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of registrant's board of directors (or persons performing the equivalent functions):
 - a) all significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - b) any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: March 10, 2006

/s/Jonathan W. Ayers Jonathan W. Ayers, Chairman, President and Chief Executive Officer

CERTIFICATION

I, Merilee Raines, certify that:

- 1) I have reviewed this annual report on Form 10-K for the year ended December 31, 2005 of IDEXX Laboratories, Inc. (the "Annual Report");
- 2) Based on my knowledge, this Annual Report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this Annual Report;
- 3) Based on my knowledge, the financial statements, and other financial information included in this Annual Report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this Annual Report;
- 4) The registrant's other certifying officers and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
 - designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this Annual Report is being prepared;
 - b) designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
 - c) evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this Annual Report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this Annual Report, based on such evaluation; and
 - d) disclosed in this Annual Report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5) The registrant's other certifying officer(s) and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of registrant's board of directors (or persons performing the equivalent functions):
 - a) all significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - b) any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: March 10, 2006

/s/Merilee Raines

Merilee Raines

Vice President and Chief F

Vice President and Chief Financial Officer

CERTIFICATION PURSUANT TO 18 U.S.C. SECTION 1350 AS ADOPTED BY SECTION 906 OF THE SARBANES-OXLEY ACT OF 2002

In connection with the Annual Report on Form 10-K of IDEXX Laboratories, Inc. (the "Company") for the year ended December 31, 2005 as filed with the Securities and Exchange Commission on the date hereof (the "Report"), the undersigned certifies, pursuant to 18 U.S.C. Section 1350, as adopted by Section 906 of the Sarbanes-Oxley Act of 2002, that:

- The Report fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934; and
- (2) The information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Company.

	/s/Jonathan W. Ayers
March 10, 2006	Jonathan W. Ayers President and Chief Executive Officer

A signed original of this written statement required by Section 906, has been provided to IDEXX Laboratories, Inc. and will be retained by IDEXX Laboratories, Inc. and furnished to the Securities and Exchange Commission or its staff upon request.

CERTIFICATION PURSUANT TO 18 U.S.C. SECTION 1350 AS ADOPTED BY SECTION 906 OF THE SARBANES-OXLEY ACT OF 2002

In connection with the Annual Report on Form 10-K of IDEXX Laboratories, Inc. (the "Company") for the year ended December 31, 2005 as filed with the Securities and Exchange Commission on the date hereof (the "Report"), the undersigned certifies, pursuant to 18 U.S.C. Section 1350, as adopted by Section 906 of the Sarbanes-Oxley Act of 2002, that:

- The Report fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934; and
- (2) The information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Company.

	/s/Merilee Raines	
March 10, 2006	Merilee Raines Vice President and Chief Financial Officer	

A signed original of this written statement required by Section 906, has been provided to IDEXX Laboratories, Inc. and will be retained by IDEXX Laboratories, Inc. and furnished to the Securities and Exchange Commission or its staff upon request.